

Helping our members work together to keep the lights on... today and in the future

September 11, 2014

VIA ELECTRONIC FILING

The Honorable Kimberly D. Bose Secretary Federal Energy Regulatory Commission 888 First Street, N.E. Washington, D.C. 20426

Re: Southwest Power Pool, Inc., Docket No. ER14-____-000

Revisions to Tariff, Bylaws, and Membership Agreement (Part 2 of 2)

Integration of the IS Parties into the Southwest Power Pool

Dear Secretary Bose:

Pursuant to Section 205 of the Federal Power Act, 16 U.S.C. 824d ("FPA"), and Section 35.13 of the Federal Energy Regulatory Commission's ("Commission") Regulations, 18 C.F.R. § 35.13, Southwest Power Pool, Inc. ("SPP"), as authorized by its Board of Directors, submits revisions to its Open Access Transmission Tariff ("Tariff")¹, Bylaws,² and Membership Agreement.³ SPP proposes the revisions to these governing documents in order to facilitate the decision of the Western Area Power Administration- Upper Great Plains Region ("Western-UGP"), a federal power marketing agency ("PMA"), Basin Electric Power Cooperative ("Basin Electric"), and Heartland Consumers Power District ("Heartland") (collectively, the "IS Parties," and individually an "IS Party") to join SPP as Transmission Owning Members, to place their respective transmission facilities under the functional control of SPP, and begin taking transmission service under the Tariff.

This filing letter and attachments are the second part of a two-part contemporaneous submission to the Commission. Due to technical limitations associated with SPP's eTariff system, SPP is not able to submit the revisions to the Bylaws and Membership Agreement in the same filing with changes to the Tariff, filing letter, and other materials, because the Bylaws and Membership Agreement are

Southwest Power Pool, Inc., FERC Electric Tariff, Sixth Revised Volume No. 1 ("Tariff").

Southwest Power Pool, Inc., Bylaws, First Revised Volume No. 4.

Southwest Power Pool, Inc., Membership Agreement, First Revised Volume No. 3.

The Honorable Kimberly D. Bose September 11, 2014 Page 2

maintained in a separate eTariff database from the Tariff. The explanation and supporting materials for the Tariff changes as well as the Bylaws and Membership Agreement changes are contained in Part 1 of the filing. Although the overall filing has been divided into two parts to accommodate the eTariff system, the Commission should treat the submissions as a single filing.

SPP requests the following effective dates for the proposed revisions: (1) October 1, 2015 for the proposed Tariff revisions; and (2) November 10, 2014 for the proposed revisions to the Bylaws and Membership Agreement. SPP respectfully requests waiver of the Commission's timing requirements to allow these tariff revisions to be effective on the dates prescribed, and for the reasons stated herein. This Part 2 letter and attachments are being served on the same parties as listed in the Part 1 filing letter.

Respectfully submitted,

/s/ Matthew Harward

Matthew Harward
Mike Riley
Southwest Power Pool, Inc.
201 Worthen Drive
Little Rock, AR 72223
Telephone: (501) 614-3560
mharward@spp.org

Attorneys for Southwest Power Pool, Inc.

1.0 Definitions

Affiliate Relationships

Affiliate Relationships are relationships between SPP Members that have one or more of the following attributes in common:

- (a) are subsidiaries of the same company;
- (b) one Member is a subsidiary of another Member;
- (c) have, through an agency agreement, turned over control of a majority of their generation facilities to another Member;
- (d) have, through an agency agreement, turned over control of a majority of their transmission system to another Member, except to the extent that the facilities are turned over to an independent transmission company recognized by FERC;
- (e) have an exclusive marketing alliance between Members; or
- (f) ownership by one Member of ten percent or greater of another Member.

Articles of Incorporation

SPP's articles of incorporation as filed with the state of Arkansas.

Board of Directors

The Board of Directors of SPP, which shall manage the general business of SPP pursuant to these Bylaws.

Bylaws

These bylaws.

Criteria

Planning and operating standards and procedures as approved by the Board of Directors.

Existing Obligations

Certain financial obligations as defined in Section 8.7.1 of these Bylaws.

ERO

The Electric Reliability Organization under FERC jurisdiction that regulates reliability of the electric power grid.

Federal Power Marketing Agency

This term shall include the term "Federal Power Marketing Administration" and have the same definition that is set forth in the Federal Power Act at 16 U.S.C. § 796(19), which defines "Federal power marketing agency" as "any agency or instrumentality of the United States (other than the Tennessee Valley Authority) which sells electric energy[.]"

Federal Power Marketing Agency Amendments

The amendments and revisions to the SPP Bylaws, the SPP Membership Agreement, and Section 39.3 of the OATT that are required by a Federal Power Marketing Agency for membership in SPP at the time of the Federal Power Marketing Agency's initial membership or as they may be revised in the future by mutual agreement between the Federal Power Marketing Agency and SPP.

Member

An entity that has met the requirements of Section 2.2 of these Bylaws.

Membership

The collective Members of SPP.

Membership Agreement

The contract, that specifies the rights and obligations of the parties, executed between SPP and an entity seeking to become an SPP member.

NERC

The North American Electric Reliability Corporation or successor organizations.

Net Energy for Load

The load served by transmission facilities under the SPP Open Access Transmission Tariff.

Officers

The officers of SPP as elected by the Board of Directors. The Officers consist of the President and the Corporate Secretary, at a minimum. Any Officer must be independent of any Member organization.

Organizational Group

A group, other than the Board of Directors, comprising a committee or working group that is charged with specific responsibilities toward accomplishing SPP's mission.

Regional Criteria

SPP planning and operating standards and procedures as approved by the Board of Directors.

Regional Entity Trustees

A governing body of SPP, independent of the Board of Directors, which specifically oversees SPP's function as an ERO Regional Entity pursuant to the Delegation Agreement between SPP and the ERO.

Regional Reliability Standards

Electric reliability requirements submitted to the ERO by the Regional Entity Trustees; and once approved, implemented and enforced by SPP under authority as the Regional Entity.

Registered Entity(ies)

A bulk electric system owner, operator or user that is required to comply with ERO reliability standards pursuant to the Energy Policy Act of 2005.

SPP

Southwest Power Pool, Inc.

SPP Regional Entity

That part of SPP responsible for the delegated functions pursuant to the Delegation Agreement between SPP and the ERO.

SPP Compliance Monitoring and Enforcement Program

The program used by the North American Electric Reliability Corporation ("NERC") and the Regional Entities to monitor, assess, and enforce compliance with Reliability Standards within the United States.

Staff

The technical and administrative staff of SPP as hired by the Officers to accomplish SPP's mission.

Standards Development Team

An SPP Organizational Group assigned or choosing to develop an SPP Regional Reliability Standard for submission to the ERO for approval for enforcement.

Terminated Member

An entity that was a Signatory to the Membership Agreement but whose membership in SPP has been terminated under Section 4 of the Membership Agreement.

Transmission Owning Member

A Member that has placed more than 500 miles of non-radial facilities operated at or above 60 kV under the independent administration of SPP for the provision of regional transmission service as set forth in the Membership Agreement.

Transmission Using Member

A Member that does not meet the definition of a Transmission Owning Member.

Western Area Power Administration-Upper Great Plains Region ("Western-UGP")

A division of the Western Area Power Administration that markets and transmits Federal power from reservoir projects under the control of the Department of the Army or the U.S. Bureau of Reclamation to Statutory Load Obligations, including preference power customers in Iowa, Minnesota, Montana, Nebraska, North Dakota, and South Dakota located in a defined marketing area. Western-UGP operates the WAUW Balancing Authority Area in the Western Interconnection, where certain of its transmission facilities are located.

2.1 Qualifications

Membership in SPP is voluntary and is open to any electric utility, Federal Power Marketing Agency, transmission service provider, any entity engaged in the business of producing, selling and/or purchasing electric energy for resale, and any entity willing to meet the membership requirements, including execution of the Membership Agreement. Membership also is open to any entity eligible to take service under the SPP Open Access Transmission Tariff (OATT). These entities desire the greater efficiency and service reliability gained through better coordination by voluntary association in SPP as constituted herein and in the SPP Articles of Incorporation. Members recognize that such association has a significant effect upon the availability and reliability of the bulk electric power supply of the region, and thereby affects the reliability of the nation's electric power supply.

3.15 Liability, Insurance and Indemnification

For purposes of this section "SPP" refers to SPP and its officers, directors, Regional Entity Trustees, employees or agents, and "Member" refers to the Members of SPP as defined in these Bylaws. None of the provisions of this section, including the waiver of liability in Section 3.15.1 below, absolving SPP or its Members, directors, Regional Entity Trustees, officer, agents, employees or other representatives of liability or any provisions for insurance or indemnification apply to actions which are unlawful, undertaken in bad faith, or are the result of gross negligence or willful misconduct.

3.15.1 Waiver of Liability

- (a) SPP shall not be liable to any Member for damages arising out of or related to any directive, order, procedure, action, or requirement of SPP, under the then effective Bylaws and Criteria.
- (b) No Member shall be liable to any other Member or to SPP for damages arising out of or related to any action by the Member pursuant to any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria.
- (c) Each Member waives any future claim it might have against SPP or other Members arising out of or resulting from any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria.
- (d) SPP waives any future claim it might have against any Member arising out of or resulting from any actions taken by a Member pursuant to any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria.

3.15.2 Insurance

The President is authorized to procure insurance to protect SPP, its directors, Regional Entity Trustees, officers, agents, employees, or other representatives against damages arising out of or related to any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria or pursuant to the OATT.

3.15.3 Indemnification of Directors, Officers, Agents and Employees

Except for actions which are unlawful, undertaken in bad faith, or are the result of gross negligence or willful misconduct, SPP shall indemnify its directors, officers, agents, employees, or other representatives to the maximum extent allowed by law consistent with these Bylaws. Each director, Regional Entity Trustee, officer, agent, employee, or other representative of SPP shall be indemnified by SPP against all judgments, penalties, fines, settlements, and reasonable expenses, including legal fees, incurred as a result of, or in connection with, any threatened, pending or completed civil, criminal, administrative, or investigative proceedings to which the incumbent may be made a party by reason of acting or having acted in official capacity as a director, Regional Entity Trustee, officer, agent, employee, or representative of SPP, or in any other capacity which the incumbent may hold at the request of SPP, as its representative in any other organization, subject to the following conditions:

- (a) Such director, Regional Entity Trustee, officer, agent, employee, or other representative must have acted in good faith and, in the case of criminal proceedings, must have had no reasonable cause to believe that conduct was unlawful; provided, that SPP shall not provide indemnification of any conduct judged unlawful in criminal proceedings. When acting in official capacity, the incumbent must have reasonably believed that conduct was in the best interests of SPP, and, when acting in any other capacity, must have reasonably believed that conduct was at least not opposed to the best interests of SPP.
- (b) If the proceeding was brought by or on behalf of SPP, however, indemnification shall be made only with respect to reasonable expenses referenced above. No indemnification of any kind shall be made in any such proceeding in which the director, Regional Entity Trustee, officer, agent, employee, or other representative shall have been adjudged liable to SPP.

- (c) In no event, however, will indemnification be made with respect to any described proceeding which charges or alleges improper personal benefit to a director, Regional Entity Trustee, officer, agent, employee, or other representative and where liability is imposed on the basis of the receipt of such improper personal benefit.
- (d) In order for any director, Regional Entity Trustee, agent, employee, or other representative to receive indemnification under this provision, the person shall vigorously assert and pursue any and all defenses to those claims, charges, or proceedings covered herein which are reasonable and legally available and shall fully cooperate with SPP or any attorneys involved in the defense of any such claim, charges, or proceedings on behalf of SPP.
- (e) No indemnification shall be made in any specific instance until it has been determined by SPP that indemnification is permissible in that specific case, under the standards set forth herein and that any expenses claimed or to be incurred are reasonable. These two (2) determinations shall be made by a majority vote of at least a quorum of the Board of Directors consisting solely of directors who were not parties to the proceeding for which indemnification or reimbursement of expenses is claimed. If such a quorum cannot be obtained, a majority of at least a quorum of the full Board of Directors, including directors who are parties to said proceeding, shall designate a special legal counsel who shall make said determinations on behalf of SPP.
- (f) Any reasonable expenses, as shall be determined above, that have been incurred by a director, Regional Entity Trustee, officer, agent, employee, or other representative who has been made a party to a proceeding as defined herein, may be paid or reimbursed in advance upon a majority vote of a quorum of the full Board of Directors, including those who may be a party to the same

proceeding. However, such director, Regional Entity Trustee, officer, agent, employee, or other representative shall have provided SPP with (i) a written affirmation under oath that the incumbent, in good faith, believes the conditions of indemnification herein have been met; and (ii) a written undertaking that the incumbent shall repay any amounts advanced, with interest accumulated at a reasonable rate, if it is ultimately determined that such conditions are not met.

3.15.4 Limitations

The provisions of this section 3.15 are subject to applicable state and Federal laws, if any, which limit the ability of a Member to waive liability or enter into agreements of indemnity. Any benefits under this Section 3.15 shall not extend to any Member so limited by state or Federal law in complying with the provisions thereof.

3.15.5 Modification of Rights by Agreement

Any provision of this Section 3.15.1 may be waived or modified by express written agreement between SPP and Member. Such express written agreement shall apply solely to the subject matter of the agreement and is not intended to be a general waiver or modification of the rights provided in Section 3.15.1.

3.15.6 Procedural Rights Not Affected

The limitations of liability provided in Section 3.15.1 shall not affect any procedural rights or obligation a Member may have at law or equity.

5.1 Members Committee

The Members Committee shall work with the Board of Directors to manage and direct the general business of SPP. Its duties shall include, but are not limited to the following:

- (a) Provide individual and collective input to the Board of Directors, including but not limited to a straw vote from the Members Committee representatives as an indication of the level of consensus among Members, on all actions pending before the Board of Directors;
- (b) Serve on committees reporting to the Board of Directors as appointed by the Board of Directors; and
- (c) Provide input with the Board of Directors to the Regional Entity Trustees on SPP Regional Reliability Standards presented by the MOPC to the Trustees or otherwise developed under the auspices of the Trustees for submission to the ERO for its approval.

5.1.1 Composition and Qualifications

5.1.1.1 Composition

Provided that Membership is sufficient to accommodate these provisions, the Members Committee shall consist of up to 24 persons. Six representatives shall be investor owned utilities Members; five representatives shall be cooperatives Members; two representatives shall be municipals Members (including municipal joint action agencies); three representatives shall be independent power producers/marketers Members; two representatives shall be state power agencies Members; one representative shall be from a Federal Power Marketing Agency; two representatives shall be alternative power/public interest Members; one representative shall be from an independent transmission company Member, defined as having assets under the OATT and no Affiliate Relationships in other categories of Membership; one representative shall be a large retail customer Member, defined as non-residential end-use customers with individual or aggregated loads of 1-MW or more; and one representative shall be a small retail customer Member, defined as residential customers and other customers

with individual or aggregated loads of less than 1-MW. Representatives will be elected in accordance with Section 5.1.2 of these Bylaws.

5.1.1.2 Qualifications

A representative shall be an officer or employee of a Member with decision-making responsibility over SPP related activities, and must be the Member's representative to the Membership.

5.1.2 Term and Election

Representatives shall be nominated by the Corporate Governance Committee and elected each year at the meeting of Members to staggered three-year terms commencing upon election and continuing until their duly elected successors take office. The election process shall be as follows:

- (a) At least 90 calendar days prior to the meeting of Members at which election of new representatives is required, the Corporate Governance Committee shall nominate persons equal in number to the representatives to be elected;
- (b) At least 30 calendar days prior to the meeting of Members, the Corporate Governance Committee shall determine the persons it nominates for election as representatives, specifying the nominee for any vacancy to be filled. The Corporate Secretary shall prepare the ballot accordingly, leaving space for additional names, and shall deliver same to Members at least two weeks prior to the meeting of Members;
- (c) For purposes of electing and removing representatives only, each group of Members with Affiliate Relationships shall be considered a single vote;
- (d) At the meeting of Members, any additional nominee or nominees may be added to the ballot if a motion is made and seconded to add such nominee or nominees; and
- (e) The required number of representatives shall be elected by written ballot.

 A Member shall be entitled to cast a number of votes equal to the number of representatives to be elected. A Member may not cumulate votes. The candidates in each sector receiving the greatest number of votes will fill vacancies.

5.1.3 Resignation and Removal of Members Committee Representatives

Any representative may resign by written notice to the President noting the effective date of the resignation. A representative may be removed, with cause, by the affirmative vote of a majority of the Members at a meeting of Members. Removal proceedings may only be initiated by a petition signed by not less than twenty percent of the Members. The petition shall state the specific grounds for removal and shall specify whether the removal vote is to be taken at a special meeting of Members or at the next regular meeting of Members. A representative who is the subject of removal proceedings shall be given fifteen days to respond to the Member petition in writing to the President.

5.1.4 Vacancies

If a vacancy occurs the Corporate Governance Committee may elect an interim representative from the same sector to serve until a replacement representative from the same sector is elected and takes office. A special election shall be held at the next meeting of Members to fill the vacancy for the unexpired term. The replacement representative shall take office immediately following the election.

5.1.5 Meetings

The Members Committee shall meet only with the Board of Directors.

6.6 Corporate Governance Committee

The Corporate Governance Committee is responsible for the overall governance structure, including nominations, for the company in accordance with its scope as approved by the Board of Directors.

To the extent that the membership allows, the CGC shall be comprised of ten members. One representative shall be the President of SPP who will serve as the Chair; one representative shall be the Chairman of the Board, unless his/her position is under consideration, in which case the Vice Chairman of the Board; one representative shall be representative of and selected by investor owned utilities Members; one representative shall be representative of and selected by co-operatives Members; one representative shall be representative of and selected by municipals Members; one representative shall be representative of and selected by state power agencies Members; one representative shall be representative of and selected by alternative power/public interest Members; one representative shall be representative of and selected by large/small retail Members; and one representative shall be representative of and selected by a Federal Power Marketing Agency Member(s).

Where a vacancy occurs with respect to a representative of a sector, the representatives from the appropriate sector will fill the vacancy. For purposes of selecting or removing representatives only, each group of Members with Affiliate Relationships shall be considered a single Member.

The CGC shall meet at least once per calendar year, and additionally as needed, provided that a quorum, as defined in these Bylaws, is present. The CGC shall report to the Board of Directors following each CGC meeting with respect to its activities and with such recommendations, as the CGC deems necessary.

8.7 Financial Obligation of Withdrawing Members

8.7.1 Existing Obligations

"Existing Obligations" are the following:

- a. Member's unpaid annual membership fee.
- b. Member's unpaid dues, assessments, and other amounts charged under Section 3.8 of the Membership Agreement, section 8.4 of the Bylaws, or otherwise under the Bylaws, plus the Member's share of costs SPP customarily includes in such dues, assessments or other charges, but which as of the Termination Date SPP had not included in such dues, assessments or other charges.
- c. Member's share (computed in accordance with the Bylaws) of the entire principal amounts of all SPP Financial Obligations outstanding as of the Termination Date. "Financial Obligations" are all long-term (in excess of six (6) months) financial obligations of SPP, including but not limited to the following:
 - debts under all mortgages, loans, loan agreements, borrowings, promissory notes, bonds, and credit lines, under which SPP is obligated, including principal and interest;
 - ii. all payment obligations under equipment leases, financing leases, capital leases, real estate and office space leases, consulting contracts, and contracts for outsourced services;
 - iii. any unfunded liabilities of any SPP employee pension funds, whether or not liquidated or demanded; and
 - iv. the general and administrative overhead of SPP for a period of three (3) months.
- d. Any costs, expenses or liabilities incurred by SPP directly due to the Termination, regardless of when incurred or payable, and including without limitation prepayment premiums or penalties arising under SPP Financial Obligations.

e. Member's share (computed in accordance with the Bylaws) of all interest that will become due for payment with respect to all interest bearing Financial Obligations after the Termination Date and until the maturity of all Financial Obligations in accordance with their respective terms ("Future Interest"). In the event that a Financial Obligation carries a variable interest rate, the interest rate in effect at the Termination Date shall be used to calculate the applicable Future Interest. In determining the Member's share of Future Interest, SPP shall take into account any reduction of Financial Obligations due to mitigation under this Section.

8.7.2 Computation of a Member's Existing Obligations

For purposes of computing the Existing Obligations of any withdrawing or terminated Member in accordance with the Membership Agreement, such "Member's share" is a percentage calculated as follows:

A = 100 [0.25(1/N) + 0.75(B/C)]

Where: A = Member's share (expressed as a percentage)

N = Total number of Members

B = The Member's previous year Net Energy for Load

C = Total of factor B for all Members

The Finance Committee shall have the discretion to reduce the Existing Obligations of any withdrawing or Terminated Member, to reflect any SPP costs or expenses that may be mitigated in connection with such Member's withdrawal or termination. In the event of consolidation of affiliate memberships or the transfer of membership from one corporate entity to another, whereby one entity remains a member of SPP, the withdrawal obligation for the departing company(ies) may be waived at SPP's sole discretion.

8.7.3 Financial Obligations for Transmission Facilities

To the extent that Section 4.3.3A of the Membership Agreement is applicable, a Terminated Member shall remain financially responsible for all financial obligations incurred and costs allocated to its load for transmission facilities approved prior to the Termination Date.

8.7.4 Penalty Costs

A Terminated Member shall remain liable for its share of costs associated with penalties assessed against SPP by FERC, the FERC-approved Electric Reliability Organization, any Electric Reliability Organization-approved Regional Entity, or any other governmental or regulatory authority with jurisdiction over SPP that SPP incurs as a result of events that occurred prior to Member's Termination Date but that SPP is unable to recover under the SPP OATT.

8.7.5 Limitation on Financial and Penalty Obligations

- (a) Notwithstanding the delineation of Members' financial obligations in Section 8.7, a Federal Power Marketing Agency shall not be subject to the financial obligations listed in this Section 8.7 in the event FERC finds that SPP has not adhered to all of the Federal Power Marketing Agency Amendments as that term is defined in Section 1.0 of these Bylaws or if SPP files and FERC approves material changes to the Federal Power Marketing Agency Amendments.
- (b) Provided further, notwithstanding any language to the contrary in these Bylaws, a Federal Power Marketing Agency has not waived or conceded any defense it may have, including sovereign immunity, intergovernmental immunity, or lack of subject matter jurisdiction in any action against it by an Enforcement Authority, nor has it accepted any liability, responsibility, or obligation to pay any civil monetary penalties or fines imposed by an Enforcement Authority to which it would not have been subject in the absence of these Bylaws. SPP, in accepting Western-UGP as a member, does not thereby concede or accept responsibility for any portion of a penalty or fine attributable to the actions or omissions of Western-UGP. SPP will identify the amount of any penalty or fine that SPP allocates to Western-UGP or that SPP determines is attributable to Western-UGP and will identify that amount to FERC as uncollectable and not otherwise owed by SPP. Enforcement Authority in these Bylaws means the Federal Energy Regulatory Commission (FERC), Electric Reliability Organization (ERO), or Regional Entities with enforcement

authority pursuant to a delegation from an ERO or FERC for the purpose of proposing and enforcing reliability standards.

10.0 Amendments To These Bylaws, The Articles Of Incorporation, And Membership Agreement

Except for modifications to Section 4.0 BOARD OF DIRECTORS, Section 5.0 COMMITTEES ADVISING THE BOARD OF DIRECTORS, Section 8.7.5 LIMITATION ON FINANCIAL AND PENALTY OBLIGATIONS, Section 9.0 REGIONAL ENTITY FUNCTION, and Section 10.0 AMENDMENTS, these Bylaws may be amended, repealed, or added to by the Board of Directors only upon 30 days written notice to the Membership of the proposed modification(s). Approval of amendments to the Bylaws by the Board of Directors must be by an affirmative vote of at least five directors. Sections 4.0, 5.0, 8.7.5, 9.0, and 10.0 of these Bylaws and the Articles of Incorporation may be amended, repealed, or added to only by approval of the Membership. Provided, that all changes to Federal Power Marketing agency/administration representation in Section 5.1.1 and any change to Section 8.7.5 must be mutually agreed to by the Federal Power Marketing Agency Member and SPP. All amendments are subject to the requisite regulatory approval(s).

MEMBERSHIP AGREEMENT TABLE OF CONTENTS

4	\sim	7	_	C*	•	. •		
	.0.		De	†11	11	tı.	α n	C
	. (/) ()				. ,,,	

7	.0	Dighte	DOWNER	and Ol	bligations	of CDD
4.		INTERIOR	LUWEIS	$and \mathbf{V}$	DUSAUUNS	() () I

- 2.1 Operation and Planning
 - 2.1.1 General
 - 2.1.2 Reliability
 - 2.1.3 Transmission Maintenance
 - 2.1.4 Generation Maintenance
 - 2.1.5 Planning Activities
- 2.2 Non-Discriminatory Transmission Service
 - 2.2.1 Pricing
 - 2.2.2 Standards of Conduct
 - 2.2.3 OASIS
 - 2.2.4 Ancillary Services
 - 2.2.5 Transmission Service Scheduling
- 2.3 Fiduciary Responsibilities and Duties of SPP to Members
- 2.4 Additional Obligations of SPP
 - 2.4.1 Inspection and Auditing Procedures
 - 2.4.2 Stranded Cost Recovery Charges

3.0 Commitments, Rights, Powers, and Obligations of Member

- 3.1 Redispatch and Curtailment
- 3.2 Transmission and Generation Maintenance Practices
- 3.3 Construction
- 3.4 Use of Distribution Facilities
- 3.5 Providing Information
- 3.6 Facilities Access
- 3.7 Inspection and Auditing Procedures
- 3.8 Compliance and Bylaws and Other Policies and Procedures
- 3.9 Planning and Participation
- 3.10 Pricing
- 3.11 No Waiver of Jurisdictional Immunity
- 3.12 Compliance with Federal or State Law

4.0 Termination of Membership

- 4.1 Events of Termination and Partial Termination
- 4.2 Termination Procedures and Effective Dates
 - 4.2.1 Notice of Voluntary Withdrawal
 - 4.2.2 Effective Date of Termination
- 4.3 Obligations Upon Termination
 - 4.3.1 Obligation to Hold Users Harmless
 - 4.3.2 Obligation to Pay Current and Existing Obligations

- 4.3.3 Construction of Transmission Facilities
- 4.3.3A Financial Obligations for Transmission Facilities
- 4.3.4 Regulatory and Other Approvals or Procedure
- 5.0 Regulatory, Tax, And Other Authorities
 - 5.1 Regulatory and Other Authorities
 - 5.2 Tax Authorities
 - 5.3 Effectiveness as to Certain Members
- 6.0 Removal of Members
- 7.0 Effective Date, Duration, and Transition
- 8.0 Miscellaneous Provisions
 - 8.1 Governing Law
 - 8.2 Successors and Assigns
 - 8.3 No Implied Waivers
 - 8.4 Severability
 - 8.5 Renegotiation
 - 8.6 Representations and Warranties
 - 8.7 Further Assurances
 - 8.8 Delivery of Notices
 - 8.9 Entire Agreement
 - 8.10 Good Faith Efforts
 - 8.11 Third Party Joint Agreements
 - 8.12 Amendment
 - 8.13 Counterparts

This Agreement is made between the Member and SPP, as defined herein.

1.0 Definitions

Agreement

This Membership Agreement.

Basin Electric Amendments

The amendments and revisions to the SPP OATT, to Basin Electric Power Cooperative's ("Basin Electric's") Network Integration Transmission Service Agreement, or to Basin Electric's Network Operating Agreement necessary for Basin Electric's initial SPP membership or as they may be revised in the future by agreement between Basin Electric and SPP.

Board of Directors

The Board of Directors elected pursuant to the Bylaws.

Bylaws

SPP's Bylaws or any successor document.

Distribution Facilities

Facilities that are the subject of a separate distribution charge pursuant to the Open Access Transmission Tariff.

Eastern Interconnection

One of the three major alternating-current electrical grids in North America. The Eastern Interconnection reaches from Central Canada eastward to the Atlantic coast (excluding Quebec), south to Florida, and back west to the foot of the Rockies (excluding most of Texas).

Effective Date

This Agreement is effective on January 1, 2000 or upon the date of execution by Member if after January 1, 2000.

Electric Transmission System

The transmission facilities subject to SPP's tariff administration, except for any Distribution Facilities.

Existing Obligations

Shall have the meaning given in Section 4.3.2(b).

Federal Power Marketing Agency

This term shall include the term "Federal Power Marketing Administration" and have the same definition that is set forth in the Federal Power Act at 16 U.S.C. § 796(19), which defines "Federal power marketing agency" as "any agency or instrumentality of the United States (other than the Tennessee Valley Authority) which sells electric energy[.]"

Federal Power Marketing Agency Amendments

The amendments and revisions to the SPP Bylaws, the SPP Membership Agreement, and Section 39.3 of the OATT that are required by a Federal Power Marketing Agency for initial SPP membership or as they may be revised in the future by mutual agreement between a Federal Power Marketing Agency and SPP.

Federal Power-Western-UGP

All capacity and energy generated at reservoir projects under the control of the Department of the Army or the Bureau of Reclamation in the marketing area of the Western-UGP for the purpose of fulfilling Western-UGP's Statutory Load Obligations for the sale of capacity and energy. This shall also include any capacity and energy delivered to or from Western-UGP under the pre-OATT bi-directional agreement with Southwestern Power Administration through Associated Electric Cooperative, Inc. for delivery and receipt at the Maryville Substation. Western-UGP's deliveries to Southwestern shall be considered part of Western's Statutory Load Obligations, and receipts from Southwestern to Western-UGP will be considered as coming from Federal resources. Federal Power-Western-UGP resources shall be eligible to be considered as Designated Resources.

FERC

The Federal Energy Regulatory Commission or successor organization.

Financial Obligations

Shall have the meaning given in Section 4.3.2(b).

Future Interest

Shall have the meaning given in Section 4.3.2(b).

Good Utility Practice

Any of the practices, methods, and acts engaged in or approved by a significant portion of the electric utility industry during the relevant time period, or any of the practices, methods, and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety, and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act, to the exclusion of all others, but rather to be a range of acceptable practices, methods, or acts generally accepted in the region. SPP Criteria and NERC Policies and Standards are considered Good Utility Practice.

Heartland Amendments

The amendments and revisions to the SPP OATT, to Heartland Consumers Power District's ("Heartland's") Network Integration Transmission Service Agreement, or to Heartland's Network Operating Agreement necessary for Heartland's initial SPP membership or as they may be revised in the future by agreement between Heartland and SPP.

Member

Signatory to this Agreement that has completed the application requirements pursuant to the Bylaws.

NERC

North American Electric Reliability Corporation or successor organizations.

Non-Transmission Owner

A Member that is not a Transmission Owner. A Non-Transmission Owner that owns or controls Tariff Facilities may have its status changed to a Transmission Owner under this Agreement upon notice to SPP and execution of this Agreement as a Transmission Owner.

Open Access Transmission Tariff (OATT)

The SPP nondiscriminatory, Open-Access Transmission Tariff (OATT) on file with FERC pursuant to Section 205 of the Federal Power Act under which SPP will offer transmission service, or any such successor tariff.

Partial Termination

Shall have the meaning given in Section 4.1.

Regional Entity

An entity having enforcement authority delegated to it by NERC pursuant to a delegation agreement accepted by FERC.

Reliability Coordinator

SPP, in performing its reliability coordinator function as recognized by NERC pursuant to its policies, and pursuant to SPP Criteria and this Agreement.

SPP

Southwest Power Pool, Inc., or successor organization.

SPP Criteria

SPP's approved operating and planning criteria.

SPP Region

The geographic area encompassing the transmission systems of Members that are Transmission Owners.

Standards of Conduct

SPP's Standards of Conduct that apply to the conduct of its directors, officers, employees, Regional Entity trustees, contractors, and agents.

Statutory Load Obligations

Western-UGP's power marketing function obligations under Federal law to deliver capacity and energy from the output of the Federal hydroelectric projects operated by the Department of the Army and the Bureau of Reclamation to loads which include project use loads, preference power customer loads defined pursuant to a power marketing plan, and other loads required to be served under Federal law.

Tariff Facilities

The Electric Transmission System and the Distribution Facilities subject to SPP's tariff administration.

Termination

Shall have the meaning given in Section 4.1.

Termination Date

Shall mean the date of Termination is effective in accordance with Section 4.2.2(b).

Transmission Customer

A customer under the Open Access Transmission Tariff

Terminated Member

An entity that was a Signatory to this Agreement but whose membership in SPP has been terminated under Section 4 of this Agreement.

Transmission Owner

A signatory to this Agreement which: (1) transfers functional control of Tariff Facilities related to the rates, terms and conditions of the OATT to SPP by executing this Agreement; or (2) appoints SPP under another agreement to provide service under the Transmission Tariff over Tariff Facilities which it owns or controls; or (3) is assigned by SPP to construct and accepts the obligation to construct new Tariff Facilities; or (4) undertakes another Transmission Owner's obligation to construct Tariff Facilities in accordance with Section 3.3(c) of this Agreement and Attachment O of the SPP OATT.

Upper Missouri Zone ("UMZ")

The_Upper Missouri Zone ("UMZ" or "Zone 19") is the rate pricing zone initially consisting of the following facilities that meet the requirements of Attachment AI, upon the transfer of those facilities to the functional control of the Transmission Provider: (i) the facilities of Western-UGP within the Eastern and Western Interconnections; (ii) the facilities owned or leased by Basin Electric Power Cooperative or Heartland Consumers Power District within the Eastern Interconnection; (iii) a portion of the facilities owned or leased by Basin Electric Power Cooperative within the Western Interconnection; and (iv) other facilities of the Western Area Power Administration transferred to the functional control of the Transmission Provider.

Western Area Power Administration-Upper Great Plains Region ("Western-UGP")

A division of the Western Area Power Administration that markets and transmits Federal power from reservoir projects under the control of the Department of the Army or the U.S. Bureau of Reclamation to Statutory Load Obligations, including preference power customers in Iowa, Minnesota, Montana, Nebraska, North Dakota, and South Dakota located in a defined marketing area. Western-UGP operates the WAUW Balancing Authority Area in the Western Interconnection, where certain of its transmission facilities are located.

Western Interconnection

One of the three major alternating-current electrical grids in North America. The Western Interconnection stretches from Western Canada South to Baja California in Mexico, reaching eastward over the Rockies to the Great Plains.

2.1 Operation and Planning

2.1.1 General

- (a) SPP shall schedule transactions and administer transmission service over Tariff Facilities as necessary to provide service in accordance with the SPP OATT.
- (b) SPP shall function in accordance with Good Utility Practice and shall conform to applicable reliability criteria, policies, standards, rules, regulations, guidelines and other requirements of SPP and NERC, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements specified in this paragraph), and all applicable requirements of Federal and state regulatory authorities.
- (c) SPP shall maintain a publicly available registry of all facilities that are not classified as critical energy infrastructure information that constitute the Electric Transmission System.
- (d) SPP shall review and approve, as appropriate, requests for service, schedule transmission transactions, and determine available transfer capability under the OATT, provided that SPP shall coordinate with the Transmission Owner when processing requests for service involving its Tariff Facilities.
- (e) SPP shall be responsible for coordinating with neighboring regional organizations and/or non-member transmission owners or providers as appropriate.
- (f) SPP shall not exercise its administration of transmission service over the Tariff Facilities in such a way as to interfere with contracts between Transmission Owner and any Transmission Customer that are in effect as of the Effective Date of this Agreement except as permitted by the OATT.
- (g) SPP shall be responsible for documenting all transmission service requests, the disposition of such requests, and any supporting data required to support the decision with respect to such requests. SPP

- shall negotiate as appropriate to develop reciprocal service, equitable tariff application, compensation principles, and any related arrangements.
- (h) SPP shall propose and file with FERC pursuant to Section 205 of the Federal Power Act modifications to the OATT and make any other necessary filings subject to approval by the Board of Directors.
- (i) SPP shall develop penalties and incentives, subject to FERC filings where appropriate.
- (j) SPP shall direct Transmission Owner pursuant to the provisions of Section 3.3 to construct transmission facilities in accordance with coordinated planning criteria, or if necessary under the OATT.
- (k) SPP shall have the authority to direct the day-to-day operations of the Tariff Facilities in order to carry out its responsibilities as a Transmission Provider and Reliability Coordinator; provided, however, nothing in this Agreement or the OATT shall be construed to require a change in the physical control of any Tariff Facilities using a Party's existing facilities or equipment.
- (l) SPP shall take any actions necessary for it to carry out its duties and responsibilities, subject to receiving any necessary regulatory approvals and any necessary approvals from the Board of Directors.

2.1.2 Reliability

SPP shall have responsibility for reliability of the Electric Transmission System in connection with its rights, powers, and obligations under this Agreement. SPP shall act as the Reliability Coordinator of the Electric Transmission System, and as such, shall have reliability monitoring and emergency response responsibilities pursuant to related SPP Criteria and the following requirements:

(a) SPP shall monitor real-time data to determine whether any control areas are experiencing generation capacity deficiencies. If a

generation capacity deficiency event threatens the security of the Electric Transmission System, SPP is authorized to and shall direct the acquisition of generation capacity and, if that direction is not satisfied, is authorized to and shall direct the shedding of firm load in the deficient control area.

- (b) SPP shall work with other reliability coordinators and non-member transmission owners or providers to develop regional reliability plans and emergency operating procedures.
- (c) SPP shall maintain emergency response procedures for responding to specified critical contingencies and shall continuously analyze issues that may require the initiation of such actions.
- (d) SPP is authorized to and shall direct the response to any emergency and Members shall carry out the required emergency actions as directed by SPP (except in cases involving endangerment to the safety of employees or the public), including the shedding of firm load if required for regional reliability.
- (e) After the conclusion of an emergency condition, any affected entity that disagrees with SPP's handling of the emergency may resolve that disagreement pursuant to the dispute resolution procedures in the Bylaws.
- (f) SPP shall monitor and coordinate the maintenance of adequate Electric Transmission System voltage levels with control areas and Transmission Owner, where appropriate.
- (g) SPP shall direct redispatch of generation in accordance with the OATT and in its role as Reliability Coordinator, subject to the generator receiving appropriate compensation pursuant to an applicable rate schedule.

2.1.3 Transmission Maintenance

SPP is required to approve all planned maintenance of the Electric Transmission System consistent with the following requirements:

- (a) SPP shall review planned transmission maintenance schedules submitted by Transmission Owner for a minimum of a rolling one-year period. These planned maintenance schedules shall be updated daily. Planned transmission maintenance requests shall be submitted to SPP at least one week in advance of an outage.
- (b) SPP shall analyze a planned transmission maintenance request to determine its effect on available transfer capability, ancillary services, the reliability of the Electric Transmission System, and any other relevant effects. Within two business days of receiving a planned maintenance request, SPP shall provide a response. If SPP's response indicates that such planned transmission maintenance will have an adverse impact, Transmission Owner shall work with SPP to minimize the impact of such planned maintenance, up to and including re-scheduling the maintenance.
- (c) SPP shall notify Transmission Owner of the need to change previously reviewed planned transmission maintenance outages if forced transmission outages or other circumstances compromise the integrity or reliability of the Electric Transmission System. If Transmission Owner is fully compensated for any additional costs resulting from any changes in maintenance schedules as provided in an applicable rate schedule, Transmission Owner shall revise maintenance outages to address such emergency circumstances.
- (d) As part of its review process, SPP shall identify planned transmission maintenance schedules that limit available transfer capability. If requested by a Transmission Customer, SPP shall identify opportunities and associated costs for rescheduling planned maintenance to enhance available transfer capability. Transmission Owner shall be compensated for the additional costs of rescheduled maintenance as provided in the SPP OATT.

- (e) SPP shall be responsible for documenting all planned transmission maintenance requests, the disposition of those requests, and all data supporting the disposition of each request.
- (f) SPP shall coordinate with Transmission Owner to the extent practicable to implement schedules for unplanned transmission maintenance when conditions endanger the safety of employees or the public, may result in damage to facilities, or may result in the unsatisfactory operation of its transmission system or any other transmission system.

2.1.4 Generation Maintenance

SPP shall coordinate the maintenance of generating units as appropriate to the extent such generation maintenance directly affects the capacity or reliability of the Electric Transmission System and the generation is located in the SPP Region as follows:

- (a) SPP shall review planned generating unit maintenance schedules submitted by generation owners for a minimum of a rolling one-year period. The planned maintenance schedules shall be updated daily. SPP shall keep such information confidential.
- (b) SPP shall analyze a planned generating unit maintenance schedule to determine its effect on available transfer capability, ancillary services, the reliability of the Electric Transmission System, and any other relevant effects. SPP shall inform a generation owner if its maintenance schedule is expected to have an impact on the reliability of the Electric Transmission System.
- (c) As part of its review process, SPP shall identify generating unit maintenance schedules that limit available transfer capability and shall identify opportunities and associated costs for rescheduling planned maintenance to enhance available transfer capability.
- (d) A generation owner that changes planned maintenance at the request of SPP pursuant to this Section 2.1.4 shall be compensated in accordance with the SPP OATT.

(e) SPP shall be responsible for documenting all planned generating unit maintenance schedules, all schedule changes, and all SPP studies and services performed with respect to planned generation maintenance.

2.1.5 Planning Activities

- (a) SPP shall engage in such planning activities, in coordination with Member, as are necessary to fulfill its obligations under this Agreement, SPP Criteria and the OATT. Such planning shall conform to applicable reliability requirements of SPP, NERC, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements), and all applicable requirements of Federal or state regulatory authorities. Such planning shall seek to minimize costs, consistent with the reliability and other requirements set forth in this Agreement. The division of responsibility for planning between Member and SPP is set forth in the SPP Criteria.
- (b) As part of its planning activities, SPP shall be responsible for planning, and for directing or arranging, necessary transmission expansions, additions, and upgrades that will enable it to provide efficient, reliable and non-discriminatory transmission service and to coordinate such efforts with the appropriate state authorities.

2.4 Additional Obligations of SPP

2.4.1 Inspection and Auditing Procedures

SPP shall grant Member, its employees, agents, or external auditors, and Federal and state regulatory authorities having jurisdiction over SPP or Member, such access to SPP's books, records, business practices, control procedures and required audit test results, and related financial transactions and settlement activities as is necessary to verify compliance by SPP with this Agreement, to audit and verify transactions under this Agreement, and to assist Member in complying with its statutory and regulatory requirements. Such access shall be at reasonable times and under reasonable conditions. SPP shall also comply with the reporting requirements of Federal and state regulatory authorities having jurisdiction over SPP with respect to the business aspects of its operations. Contact between officers, employees, and agents of Member and those of SPP shall comply with the Standards of Conduct.

2.4.2 Stranded Cost Recovery Charges

SPP shall collect and distribute, as appropriate, any stranded cost recovery charges pursuant to applicable schedules accepted by appropriate regulatory entities.

3.0 Commitments, Rights, Powers, and Obligations of Member

Member has made the following commitments, and shall have the following rights and shall be responsible for the following functions, some of which apply only to a Transmission Owner, some only to a Non-Transmission Owner.

- (a) Transmission Owner shall transfer functional control related to the rates, terms and conditions of the OATT of its Transmission Facilities, subject to receiving all necessary regulatory authorizations, thereby allowing SPP to (i) direct the operation of the Transmission Facilities in accordance with the terms of this Agreement; (ii) administer transmission service under the Transmission Tariff over that Transmission Owner's Tariff Facilities; and (iii) receive funds from Transmission Customers relating to transmission service over Tariff Facilities and distribute funds to the Transmission Owner. Where Member, owns generators within the SPP Region which directly affect the capacity or reliability of the Electric Transmission System, it shall offer to provide the ancillary services required under the OATT at rates approved by regulatory authorities, where appropriate, to the extent such generators are able to provide such ancillary services.
- (b) Transmission Owner shall operate and maintain its Tariff Facilities subject to the requirements of this Agreement.
- (c) Where Transmission Owner is a balancing area operator, it shall continue to operate its balancing areas for local generation control and economic dispatch, and shall be responsible for identifying and addressing local problems in a reliable manner.
- (d) Transmission Owner shall provide transmission service over its Tariff Facilities at the direction of SPP pursuant to the terms of the OATT.
- (e) Member agrees to comply with the instructions of SPP in its role as Reliability Coordinator.
- (f) Transmission Owner shall retain all rights of ownership, including legal and equitable title in its Tariff Facilities, subject to the provisions of this Agreement.

 Transmission Owner, or one acting under its authority, shall retain all rights to

- access to its Tariff Facilities so long as such access is consistent with the provisions of this Agreement.
- (g) Notwithstanding any other provision in this Agreement, Transmission Owner shall not be obligated or be considered to be allowing transmission over its facilities if such transmission would cause the loss of the tax-exempt status of Transmission Owner or any bonds or other debt of Transmission Owner.
- (h) Notwithstanding any other provisions of this Agreement, Member reserves the right to exercise operational authority of Member's Tariff Facilities (1) to protect public safety and the safety of its workers, to prevent damage to equipment, and to preserve reliability in compliance with NERC standards, and (2) as necessary to preserve Member's rights, duties and obligations regarding electric service to its retail and wholesale native load customers pursuant to any applicable Federal or state law and consistent with NERC standards, if SPP's exercise of operational authority over the Tariff Facilities would endanger said electric service or is contrary to or would curtail, surrender or delegate such Federal or state law rights, duties and obligations. Member will, as soon as reasonably practicable thereafter, notify SPP of such actions taken by Member. Member and SPP will meet and confer regarding the matter and, as necessary, negotiate in good faith to modify the Agreement to address the matter.

3.3 Construction

- (a) As part of its planning activities, SPP shall be responsible for planning, and for directing or arranging, necessary transmission expansions, additions, and upgrades that will enable it to provide efficient, reliable and non-discriminatory transmission service and to coordinate such efforts with the appropriate Federal or state authorities, including the Member's governing board where it serves as that authority, or in the case of a Federal Power Marketing Agency, the Administrator. Transmission Owner shall use due diligence to construct transmission facilities as directed by SPP in accordance with the OATT and this Agreement, subject to such siting, permitting, and environmental constraints as may be imposed by Federal, state, and local laws and regulations, and subject to the receipt of any necessary Federal or state regulatory approvals, including, as necessary, the Member's governing board where it serves as that authority. Such construction shall be performed in accordance with Good Utility Practice, applicable SPP Criteria, industry standards, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements), and in accordance with all applicable requirements of Federal or state regulatory authorities. Transmission Owner shall be fully compensated to the greatest extent permitted by FERC, or other legislative or regulatory authority for the costs of construction undertaken in accordance with the OATT.
- (b) After a new transmission project has received the required approvals and been approved by SPP, SPP will direct the appropriate Transmission Owner(s) to begin implementation of the project in accordance with Attachment O of the OATT.

3.10 Pricing

Transmission Owner shall possess the unilateral right to file with FERC, and, if the Transmission Owner is non-jurisdictional, the Transmission Owner shall have the option to file with FERC, or submit to SPP for filing with FERC, pursuant to Section 205 of the Federal Power Act and Federal statutory and regulatory requirements, including Delegation Order No. 00-37.00A, 10 CFR Part 903 and 18 CFR Part 300, as amended or superseded, modifications to change the rates or rate structure for transmission service over its Tariff Facilities, including filing a fixed revenue requirement and supporting data or a rate formula template for its cost of service revenue requirements, and to submit proposals or filings governing new construction with FERC; provided, however, Transmission Owner may not submit a proposal which results in a Transmission Customer paying two or more transmission charges for transmission for one transaction under the OATT (excluding Distribution Facilities for which an additional charge may be imposed, and Grandfathered Agreements as defined in the OATT). Transmission Owner shall notify SPP in advance of its intention to submit a filing to FERC and provide SPP with a copy of the filing. No approval from SPP is required for such filings.

3.11 No Waiver of Jurisdictional Immunity

If Member is not subject to the jurisdiction of FERC as a public utility under the Federal Power Act, Member shall not be required to take any action or participate in any filing or appeal that would confer FERC jurisdiction over Member that does not otherwise exist. Any order, decision, rule or regulation issued by FERC to SPP or any other Members or Member of SPP relating to matters exempt from FERC jurisdiction under Section 201(f) of the Federal Power Act shall not apply directly or separately to a non-jurisdictional Member. Without limiting the generality of the foregoing, except as otherwise provided in the Federal Power Act, a non-jurisdictional Member shall not be bound or obligated by any FERC order, decision, rule or regulation requiring a change in the rates, terms or conditions for transmission service or compensation for utilizing the transmission facilities of a non-jurisdictional Member, which conflicts with applicable Federal or state law, including any order requiring the suspension of the use of such rates, terms or conditions or the payment of refunds of rates or compensation previously collected or received. A non-jurisdictional Member and SPP acknowledge that FERC, in the context of its jurisdiction over SPP's rates, may review a non-jurisdictional Member's revenue requirement and rates to the extent they comprise or affect the rates charged by SPP or other Members. In the case of a Federal Power Marketing Agency, this review shall be consistent with the Delegation Order No. 00-037.00A, as superseded or amended, from the Secretary of Energy to the Power Marketing Administrations and the FERC, including the regulations implementing this review authority. If FERC does not accept a non-jurisdictional Member's revenue requirement or rates, the non-jurisdictional Member may terminate this Agreement pursuant to the withdrawal provisions of the Agreement. In such event, the non-jurisdictional Member and SPP agree to meet and confer prior to any termination of this Agreement. Nothing in this Agreement, or the participation of a non-jurisdictional Member in SPP and its operations waives any objection to or otherwise constitutes a consent to, the jurisdiction by FERC that does not otherwise exist over the non-jurisdictional Member or its transmission service, facilities and rates.

3.12 Compliance with Federal or State Law

Notwithstanding any other provision of this Agreement, a non-jurisdictional Member shall not be required to take any action or do any other thing with respect to rates, charges, terms or conditions of service, the resolution of disputes under this Agreement or any other matter regarding its obligations and performance under this Agreement, that (i) the non-jurisdictional Member is not permitted by Federal or state law to undertake or that is prohibited in whole or in part by any Federal or state law or regulation applicable to the non-jurisdictional Member; or (ii) would require the nonjurisdictional Member to violate a provision of such state or Federal law or regulation in order to comply with this Agreement. Determination of compliance with and permissible action, conduct or obligations under this Section 3.12 by a non-jurisdictional Member shall be within the sole jurisdiction of the non-jurisdictional Member's governing board, or in the case of a Federal Power Marketing Agency, its Administrator, subject to applicable Federal or state court review. A non-jurisdictional Member shall not object to SPP's participation in any Federal or state proceedings that impact the non-jurisdictional Member's ability to perform under this Agreement or determinations regarding such impact. To the extent possible without violating Federal or state law, a non-jurisdictional Member shall notify SPP in advance of any action that the non-jurisdictional Member is required to take that the non-jurisdictional Member believes would constitute a violation of Federal or state law, and the non-jurisdictional Member and SPP promptly shall meet and confer regarding the matter. As necessary, the non-jurisdictional Member and SPP agree to negotiate in good faith to modify the Agreement as consistent as possible with the original intent to allow SPP to exercise operational authority over the nonjurisdictional Member's Tariff Facilities as otherwise provided in the Agreement. If the non-jurisdictional Member and SPP are unable to resolve the matter, the nonjurisdictional Member may terminate this Agreement pursuant to the withdrawal provisions of the Agreement.

4.1 Events of Termination and Partial Termination.

A "Termination" shall mean any cessation of Membership, voluntary or involuntary, or a termination of this Agreement for any reason including the following:

- a. Member voluntarily withdraws from membership under Sections 4.0 or
 5.0 of this Agreement;
- b. An involuntary termination of membership occurs pursuant to Section 6.0 of this Agreement;
- c. Member withdraws from membership or terminates this Agreement to comply with the terms of any applicable law or regulation;
- d. A withdrawal from membership or termination of this Agreement is ordered by any court or administrative agency of competent jurisdiction; SPP reserves the right, but is not obligated, to maintain before such court or administrative agency, or on any appeal, that FERC has preemptive jurisdiction;
- e. A material breach or repudiation of this Agreement, in the discretion of the non-breaching or non-repudiating party;
- f. The liquidation or dissolution of SPP, unless a third party has assumed the rights and obligations of SPP under this Agreement consistent with Section 8.2, and has reasonably demonstrated capability to perform SPP's obligations under this Agreement;
- g. An agreement between SPP and the Member to terminate this Agreement.

A "Partial Termination" occurs upon a Member's voluntary removal of a portion of its transmission facilities or customers from the SPP Region, including, by way of example and not limitation, sale of a part of the Member's distribution or transmission network or transfer to another service provider of a portion of its retail load.

4.2 Termination Procedures and Effective Dates

4.2.1 Notice of Voluntary Withdrawal

- (a) **Notice.** Subject to Section 4.3, a Member may withdraw voluntarily from this Agreement, provided that it has given written notice to the President of its intent to withdraw. Notice of intent to withdraw must state a proposed date for the withdrawal and be delivered to the President no less than twenty-four (24) months prior to such date. The President will advise the Members and the Board of Directors of any withdrawal notices received. In order to assure that there is no more than one proposed termination date with respect to a Member, a withdrawal notice shall be deemed to supersede any prior withdrawal notice given by the Member, except that a Member may not submit a withdrawal notice less than twenty-four (24) months prior to the termination date proposed in the Member's previous notice of intent to withdraw. Voluntary withdrawal is a Termination and creates the same obligations as a Termination for any other reason. Upon receiving a notice of intent to withdraw, SPP shall account for such notice of intent to withdraw in the SPP planning process, unless the Member plans to continue to take transmission service from SPP after the termination date.
- (b) Withdrawal Deposit. A Member submitting a written notice of its intent to withdraw from this Agreement must simultaneously submit a cash withdrawal deposit to SPP, as set forth in the table below. SPP will not accept a notice of intent to withdraw without a withdrawal deposit. SPP will treat the withdrawal deposit as a pre-payment of a portion of the costs SPP incurs to process the Member's withdrawal from SPP, as set forth in Section 4.3.2(d) of this Agreement, or the costs associated with reintegrating the Member into SPP if the Member subsequently rescinds its notice of intent to withdraw and SPP incurs costs to reintegrate the Member. Withdrawal deposits are as follows:

Member Category	Withdrawal Deposit
Load Serving Entity	\$ 150,000
Non-Load Serving Entity	\$ 50,000

If the cost of processing Member's withdrawal as calculated by SPP pursuant to Section 4.3.2(d) of this Agreement exceeds the withdrawal deposit, the additional amount shall be included in the invoice SPP provides to the Member under 4.3.2(e) of this Agreement. If the Member rescinds

its notice of intent to withdraw and the cost of processing the Member's withdrawal and subsequent reintegration into SPP exceeds the withdrawal deposit, SPP shall invoice the Member for the amount of the cost that exceeds the deposit, and the Member shall provide payment to SPP within thirty (30) days of receipt of the invoice. If the withdrawal deposit exceeds the costs of processing the Member's withdrawal and/or reintegration, SPP shall refund the difference to the Member. Notwithstanding the foregoing, a Federal Power Marketing Agency shall not be required to make a withdrawal deposit and shall only be responsible for paying SPP costs after they are incurred and appropriately invoiced pursuant to 4.3.2(e).

4.2.2 Effective Date of Termination

- Owner subject to FERC jurisdiction, the Termination Date shall be the date proposed in the withdrawal notice under Section 4.2.1 or otherwise agreed by SPP. If the withdrawing Member is a Transmission Owner subject to FERC jurisdiction, the Termination Date shall be the later of (i) the proposed date specified in the withdrawal notice or otherwise agreed by SPP, (ii) the effective date, if any, set by the FERC order approving the withdrawal; or (iii) the date that such FERC order is no longer subject to review by a court of competent jurisdiction.
- (b) **Termination other than Voluntary Withdrawal**. If the Termination occurs for any reason other than the Member's voluntary withdrawal under Section 4.2.1 or by agreement with SPP, the Termination Date shall be as follows:
 - (i) If the Member is not a Transmission Owner subject to FERC jurisdiction, the Termination Date shall be the date of the event by which the Termination occurs, for example, the date a party gives notice that it will treat a breach or repudiation as a Termination or the date a Member withdraws in order to comply with the terms of a law or regulation. The foregoing notwithstanding, if the Termination occurs due to the order of a court or administrative agency, the Termination Date shall be the date the order is no longer subject to review by a court of competent jurisdiction.

- (ii) If the Member is a Transmission Owner requiring regulatory agency approval prior to effectively withdrawing from SPP, then the Termination Date shall be the later of (i) the effective date, if any, set by the regulatory agency order approving the Termination; or (ii) the date that such regulatory order is no longer subject to review by a court of competent jurisdiction.
- (iii) In conjunction with the requirements and limitations imposed in Section 4.2.2(b)(ii) above; Transmission Owning Members of SPP wishing to withdraw from SPP and requiring regulatory agency approval prior to such withdrawal being effective are required to initiate the required regulatory filings seeking approval to withdraw prior to the end of the 24-month notice period. Additionally, these members must provide SPP copies of the regulatory filings in a timely manner after filing with the appropriate regulatory agency. Failure to comply with the terms of this paragraph will effectively rescind the notice of the withdrawing member.
- (iv) Member may terminate this Agreement with less than the required twenty-four (24) months' notice, in the event that the Federal or state law governing Member changes, or any provisions of this Agreement, the provisions of SPP's OATT, or SPP's Bylaws are changed or modified in a manner that causes a conflict with the Member's Federal or state law, regulations, or rate schedules, and the internal dispute resolution process described in Section 12 of the OATT is unable to resolve such conflict. In such event, Member and SPP shall meet and confer to facilitate the withdrawal as soon as practicable as necessary to ensure compliance with Federal or state law.
- (v) Any Member with Transmission Facilities located in the Upper Missouri Zone may terminate this Agreement with less than the required twenty-four (24) month notice in the event that Western-

UGP or Basin Electric Power Cooperative withdraws from SPP in accordance with its respective withdrawal rights or if FERC finds that SPP has not adhered to all of the Federal Power Marketing Agency Amendments, the Basin Electric Amendments, or the Heartland Amendments. In such event, Member and SPP shall meet and confer to facilitate the withdrawal as soon as practicable or as necessary to ensure compliance with state or Federal law. In the event of a withdrawal by Western-UGP or Basin Electric Power Cooperative, Member's withdrawal will become effective on the same date as that of Western-UGP or Basin Electric Power Cooperative. Such Member also may terminate this Agreement in the event that SPP files and FERC approves changes to the Federal Power Marketing Agency Amendments, the Basin Electric Amendments, or the Heartland Amendments that have a material adverse effect on such Member. If such Member exercises its withdrawal rights under this provision, the financial obligations will be calculated under § 4.3 of this Agreement.

4.3 Obligations Upon Termination

4.3.1 Obligation to Hold Users Harmless

Transmission Customers taking service which involves facilities being withdrawn by a Transmission Owner from the functional control of SPP and where such service is under transmission contracts executed before the Termination Date shall continue to receive the same service for the remaining term of each such contract at the same rates, terms, and conditions that would have been applicable if the Termination or Partial Termination had not occurred. Transmission Owner agrees to continue providing service to such Transmission Customers in accordance with the preceding sentence, and shall receive revenues calculated in accordance with the OATT but no more in revenues for that service that if there had been no Termination or Partial Termination.

4.3.2 Obligation to Pay Current and Existing Obligations

- (a) In the event of a Termination or Partial Termination, Member shall pay all obligations incurred under this Agreement at any time prior to the Termination Date. In addition, in order for SPP to recover a portion of certain debts and cost payable by SPP after the Termination Date as further specified in this Agreement, the Member shall pay all Existing Obligations (as defined herein) calculated as of the Termination Date. SPP shall make reasonable efforts to mitigate the Member's Existing Obligations by commercially reasonable actions (such as prepayment of allocable debt, or investment of part or all of the Member's payment in an interest-bearing instrument) and, in its discretion, may further discount the Member's Existing Obligations to reflect any additional mitigation SPP determines it will achieve.
- (b) "Existing Obligations" are all of the following and other obligations as may be set forth in the Bylaws from time to time;
 - i. Member's unpaid annual membership fee,
 - ii. Member's unpaid dues, assessments, and other amounts charged under Section 3.8 of this Agreement, Section 8.4 of the Bylaws, or otherwise under the Bylaws, plus the Member's share of costs SPP

- customarily includes in such dues, assessments or other charges, but which as of the Termination Date SPP had not included in such dues assessments or other charges.
- iii. Member's share (computed in accordance with the Bylaws) of the entire principal amounts of all SPP Financial Obligations outstanding as of the Termination Date. "Financial Obligations" are all long-term (in excess of six (6) months) financial obligations of SPP, including but not limited to the following:
 - debts under all mortgages, loans, loan agreements, borrowings, promissory notes, bonds, and credit lines under which SPP is obligated, including principal and interest;
 - all payment obligations under equipment leases, financing leases, capital leases, real estate and office space leases, consulting contracts, and contracts for outsourced services;
 - c. any unfunded liabilities of any SPP employee pension funds, whether or not liquidated or demanded; and
 - d. the general and administrative overhead of SPP for a period of three (3) months.
- iv. Any costs, expenses or liabilities incurred by SPP directly due to the Termination, regardless of when incurred or payable, and including without limitation prepayment premiums or penalties arising under SPP Financial Obligations.
- v. Member's share (computed in accordance with the Bylaws) of all interest that will become due for payment with respect to all interest bearing Financial Obligations after the Termination Date and until the maturity of all Financial Oblations in accordance with their respective terms ("Future Interest"). In the event that a Financial Obligation carries a variable interest rate, the interest rate in effect at the Termination Date shall be used to calculate the applicable Future Interest. In determining the Member's share of

Future Interest, SPP shall take into account any reduction of Financial Obligations due to mitigation under this Section.

- (c) In the event of a Partial Termination, Existing Obligations shall first be calculated as though a Termination occurred, and the Member shall pay a percentage thereof as Existing Obligations due to the Partial Termination. Such percentage shall be the percentage reduction of the Net Energy for Load Ratio applicable to the Member resulting from the Partial Termination.
- In the event of a Termination or Partial Termination by a Member, the Member shall pay to SPP all costs SPP incurs to remove the Member's facilities and/or load from SPP markets and operations. Such costs will be determined by SPP and shall include but not be limited to costs associated with modifying systems and databases, staff time, legal costs, and all costs of completing other tasks necessary to process the Member's Termination. SPP will apply the Member's withdrawal deposit, as specified in Section 4.2.1(b), to such costs, and any costs exceeding the withdrawal deposit shall be included in the invoice to the Member as discussed in Section 4.3.2(e) of this Agreement.
- (e) SPP shall invoice Member for Existing Obligations within one month after the Termination Date, except that delay by SPP in issuing the invoice shall not diminish Member's obligation to make timely payment. The invoice shall be due and payable no later than five (5) business days after issuance. Any amounts owed by SPP to the Member shall, solely at SPP's election and in its discretion, be offset against the Member's Existing Obligations or paid to the Member concurrently with issuance of the invoice.
- (f) The Member acknowledges and agrees that Existing Obligations include amounts that SPP expects to accrue and that will become payable by SPP between the date of Member's Notice of Termination and the Member's Termination Date, and that no part of a payment of Existing Obligations shall be refundable to the Member under any circumstances, including

(except as provided in this Section with respect to mitigation or the execution of a new Agreement by the Member after the Member's Termination) any reduction of the Financial Obligations. Any disagreement as to the calculation of Existing Obligations shall be resolved in accordance with the dispute resolution procedures in the Bylaws. If, after Termination, the Member elects to re-join SPP and execute the Agreement then in effect, SPP, in its sole discretion, may elect to credit a portion or all of the Member's Existing Obligations paid to SPP upon the Member's earlier Termination against any future payments owed by the Member to SPP.

4.3.3 Construction of Transmission Facilities

Any obligations relating to the construction of new facilities pursuant to an approved plan of SPP shall be negotiated between SPP and the Transmission Owner prior to the Termination Date so as to continue the Transmission Owner's construction obligation for facilities for which SPP has issued a notification to construct to the Transmission Owner prior to the Termination Date. If such obligations cannot be resolved through negotiations, they shall be resolved in accordance with the dispute resolution procedures in the Bylaws and Agreement.

4.3.3A Financial Obligations for Transmission Facilities

This Section 4.3.3A applies to any Terminated Member that was a Transmission Owner at the time it submitted its notice of intent to withdraw. Such Terminated Member shall remain financially responsible for all financial obligations incurred and costs allocated to its load for transmission facilities approved prior to the Termination Date. Payments in fulfillment of any such obligations and allocated costs shall commence on the date that the costs of such transmission facilities are reflected in SPP's generally applicable rates, unless SPP and the Terminated Member agree to an alternate date. Rights, obligations, and payments applicable to time periods prior to the Termination Date shall be honored by SPP and the Terminated Member. Fulfillment and performance of such rights and obligations, and rights and obligations regarding the use of such transmission facilities, shall be negotiated between SPP and the Terminated

Member, and any disputes involving such rights and obligations shall be resolved in accordance with the dispute resolution procedures in the Bylaws and Agreement.

4.3.4 Regulatory and Other Approvals or Procedures

Any Termination with respect to a Transmission Owner shall be subject to applicable Federal and state law and regulatory approvals or procedures.

5.1 Regulatory and Other Authorities

This Agreement and the participation of Member is subject to acceptance or approval by FERC, and may be subject to actions of respective Federal or state regulatory authorities to which Member may be subject, and to the actions of any other governmental body which may affect the ability of Member to participate in this Agreement. The following items describe Member's rights and obligations in the event regulatory and other approvals or acceptances are not obtained or changes are required:

- (a) In the event FERC disapproves or refuses to accept this Agreement or the changes to the OATT, transmission service agreements, and Bylaws developed together with this Agreement, then this Agreement shall cease to be effective except that the signatories shall be obligated to attempt expeditiously and in good faith to negotiate a substitute agreement and OATT, transmission service agreements, and Bylaws which address the reasons for such FERC action. If, despite such good faith negotiation, the signatories are unable to produce such a substitute agreement and OATT, transmission service agreements, and Bylaws, then the signatories shall have no further obligations under this Agreement or any filing associated herewith.
- (b) In the event of any order or decision by FERC or by a court modifying this Agreement or the OATT, transmission service agreements, and Bylaws submitted as part of the initial filing seeking FERC acceptance or approval, that in the judgment of Member adversely affects it, then Member, at its sole discretion, may withdraw from this Agreement by providing written notice to the President of SPP no later than thirty days after such order or decision without receiving any FERC authorization. In such event, Member will in good faith negotiate to determine whether changes should be made to the Agreement or the OATT, transmission service agreements, and Bylaws to address the reasons for Member's withdrawal.

5.2 Tax Authorities

If the Internal Revenue Service or any other Federal, state, or local taxing authority issues, or fails to issue, any ruling, or imposes any requirement or obligation, in connection with this Agreement on Member adverse to Member (in its sole judgment), or if adherence to this Agreement jeopardizes the tax-exempt status of Member or its bonds, then Member may, within 30 days of the date of such final order, or a good faith belief of such adverse consequences, withdraw from this Agreement subject to receiving any necessary regulatory approvals. In such event, Member and SPP will, in good faith, negotiate to determine whether changes should be made to the Agreement to address the reasons for Member's withdrawal.

Nothing in this Agreement, nor Member's obligations and performance thereunder, shall affect, or require Member to take or refrain from taking any action that would affect the rights and obligations or enforceability of Member's present or future bond resolutions, tax-exempt debt covenants and financing agreements. Member shall determine in its sole discretion and judgment, in accordance with advice and opinions from its legal counsel, what actions, conduct and performance it is permitted to or must take under its bond resolutions, tax-exempt debt covenants and financing agreements. Member and SPP will meet and confer regarding the matter and, as necessary, negotiate in good faith to modify the Agreement to address the matter.

5.3 Effectiveness as to Certain Members

The effectiveness of this Agreement as to Member where it is a governmental entity or non-public utility and has outstanding tax-exempt bonds issued to finance, in whole or in part, generation, transmission, or Distribution Facilities is dependent upon satisfaction or Member's written waiver of the following conditions precedent:

- (a) Receipt of an unqualified opinion of a nationally recognized bond counsel to the effect that the provisions of this Agreement do not adversely affect the exclusion from gross income of interest on any such outstanding bonds issued to finance generation, transmission, and Distribution Facilities under the Internal Revenue code of 1986, as amended;
- (b) Receipt of an unqualified opinion of a nationally recognized bond counsel or general counsel to Member to the effect that the provisions of this Agreement do not constitute a breach or impairment of, or a default under, any agreement to which it is a party, including, but not limited to, its master bond resolution, as amended, and any power sales contracts with its municipal transmission users (if any), as amended, or other agreements;
- c) Receipt of a certificate of the trustee for any such outstanding bonds issued for generation, transmission and Distribution Facilities to the effect that Member's entry into this Agreement is permitted under the master bond resolution, as amended; and
- (d) Receipt of an opinion of nationally recognized bond counsel or general counsel to Member that it has full constitutional and statutory authority to enter into this Agreement.

In the event that any of the foregoing conditions are not satisfied or waived by Member, then it shall promptly give notice of its objections or conditions which have not been satisfied to SPP, and SPP shall expeditiously attempt in good faith to negotiate a substitute agreement.

8.1 Governing Law

This Agreement shall be interpreted, construed, and governed by the laws of the State of Arkansas, except to the extent preempted by the law and/or unless a court with jurisdiction rules otherwise, and in a case involving a Federal Power Marketing Agency, Federal law shall apply, provided, however, that (i) all matters relating to real property or any interest in realty shall be governed by the laws of the State wherein such real property or interest in realty is physically located, and (ii) any court or regulatory body applying Arkansas law shall give full effect to Section 3.12 of this Agreement regarding Member's obligations under state law.

8.2 Successors and Assigns

This Agreement shall inure to the benefit of, and be binding upon SPP and Member, their respective successors and assigns permitted hereunder, but shall not be assignable by SPP without prior written agreement from Member, with such written agreement not to be unreasonably withheld, or by Member, by operation of law or otherwise, without the approval of the Board of Directors which approval shall not be unreasonably withheld, except that no Board of Directors approval is required as to a successor in the operation of Transmission Owner's Tariff Facilities committed to administration by SPP by reason of a merger, consolidation, reorganization, sale, spin-off, or foreclosure, as a result of which substantially all such transmission facilities are acquired by such successor, and such successor becomes a Transmission Owner under this Agreement.

8.4 Severability

Except as may be stated otherwise in any Amendments to this Agreement, each provision of this Agreement shall be considered severable, and if for any reason any provision of this Agreement, or the application thereof to any person, entity, or circumstance, is determined by a court or regulatory authority of competent jurisdiction to be invalid, void, or unenforceable, then the remaining provisions of this Agreement shall continue in full force and effect and shall in no way be affected, impaired, or invalidated, and such invalid, void, or unenforceable provision shall be replaced with a suitable and equitable provision in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid, void, or unenforceable provision. This Section 8.4 does not modify or change in any way the right of Member to withdraw as provided elsewhere in this Agreement.

8.6 Representations and Warranties

Member and SPP each represent and warrant to the other that as of the later of the date it executes this Agreement or the Effective Date of this Agreement:

- (a) It is duly organized, validly existing, and in good standing under the laws of the jurisdiction where organized;
- (b) Subject to any necessary approvals by Federal or state regulatory authorities of SPP, the execution and delivery by Member and SPP of this Agreement, and the performance of its respective obligations hereunder have been duly and validly authorized by all requisite action on the part of the signatories and does not conflict with any applicable law or with any other agreement binding upon the signatories, other than third party joint agreements covered in this Agreement. This Agreement has been duly executed and delivered by Member and SPP, and, subject to the conditions set forth in this Agreement, constitutes the legal, valid, and binding obligation on the part of Member and SPP, enforceable against it in accordance with its terms except insofar as the enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium, or other similar laws affecting the enforcement of creditor's rights generally, and by general principles of equity regardless of whether such principles are considered in a proceeding at law or in equity; and
- (c) There are no actions at law, suits in equity, proceedings, or claims pending or, to the knowledge of Member or SPP, threatened against Member or SPP before or by any Federal, state, foreign or local court, tribunal, or governmental agency or authority that might materially delay, prevent, or hinder the performance by such entity of its obligations hereunder.

8.10 Good Faith Efforts

Member and SPP agree that each shall in good faith take all reasonable actions necessary to fulfill its respective obligations under this Agreement. Where the consent, agreement, or approval of Member or SPP must be obtained hereunder, such consent, agreement, or approval shall not be unreasonably withheld, conditioned, or delayed. Where Member or SPP is required or permitted to act, or omit to act, based on its opinion or judgment, such opinion or judgment shall not be unreasonably exercised. To the extent that the jurisdiction of any Federal or state regulatory authority applies to any part of this Agreement and/or the transactions or actions covered by this Agreement, Member and SPP secure any necessary or desirable approval or acceptance of such regulatory authorities of such part of this Agreement and/or such transactions or actions.

1.0 Definitions

Affiliate Relationships

Affiliate Relationships are relationships between SPP Members that have one or more of the following attributes in common:

- (a) are subsidiaries of the same company;
- (b) one Member is a subsidiary of another Member;
- (c) have, through an agency agreement, turned over control of a majority of their generation facilities to another Member;
- (d) have, through an agency agreement, turned over control of a majority of their transmission system to another Member, except to the extent that the facilities are turned over to an independent transmission company recognized by FERC;
- (e) have an exclusive marketing alliance between Members; or
- (f) ownership by one Member of ten percent or greater of another Member.

Articles of Incorporation

SPP's articles of incorporation as filed with the state of Arkansas.

Board of Directors

The Board of Directors of SPP, which shall manage the general business of SPP pursuant to these Bylaws.

Bylaws

These bylaws.

Criteria

Planning and operating standards and procedures as approved by the Board of Directors.

Existing Obligations

Certain financial obligations as defined in Section 8.7.1 of these Bylaws.

ERO

The Electric Reliability Organization under FERC jurisdiction that regulates reliability of the electric power grid.

Federal Power Marketing Agency

This term shall include the term "Federal Power Marketing Administration" and have the same definition that is set forth in the Federal Power Act at 16 U.S.C. § 796(19), which defines "Federal power marketing agency" as "any agency or instrumentality of the United States (other than the Tennessee Valley Authority) which sells electric energy[.]"

Federal Power Marketing Agency Amendments

The amendments and revisions to the SPP Bylaws, the SPP Membership Agreement, and Section 39.3 of the OATT that are required by a Federal Power Marketing Agency for membership in SPP at the time of the Federal Power Marketing Agency's initial membership or as they may be revised in the future by mutual agreement between the Federal Power Marketing Agency and SPP.

Member

An entity that has met the requirements of Section 2.2 of these Bylaws.

Membership

The collective Members of SPP.

Membership Agreement

The contract, that specifies the rights and obligations of the parties, executed between SPP and an entity seeking to become an SPP member.

NERC

The North American Electric Reliability Corporation or successor organizations.

Net Energy for Load

The load served by transmission facilities under the SPP Open Access Transmission Tariff.

Officers

The officers of SPP as elected by the Board of Directors. The Officers consist of the President and the Corporate Secretary, at a minimum. Any Officer must be independent of any Member organization.

Organizational Group

A group, other than the Board of Directors, comprising a committee or working group that is charged with specific responsibilities toward accomplishing SPP's mission.

Regional Criteria

SPP planning and operating standards and procedures as approved by the Board of Directors.

Regional Entity Trustees

A governing body of SPP, independent of the Board of Directors, which specifically oversees SPP's function as an ERO Regional Entity pursuant to the Delegation Agreement between SPP and the ERO.

Regional Reliability Standards

Electric reliability requirements submitted to the ERO by the Regional Entity Trustees; and once approved, implemented and enforced by SPP under authority as the Regional Entity.

Registered Entity(ies)

A bulk electric system owner, operator or user that is required to comply with ERO reliability standards pursuant to the Energy Policy Act of 2005.

SPP

Southwest Power Pool, Inc.

SPP Regional Entity

That part of SPP responsible for the delegated functions pursuant to the Delegation Agreement between SPP and the ERO.

SPP Compliance Monitoring and Enforcement Program

The program used by the North American Electric Reliability Corporation ("NERC") and the Regional Entities to monitor, assess, and enforce compliance with Reliability Standards within the United States.

Staff

The technical and administrative staff of SPP as hired by the Officers to accomplish SPP's mission.

Standards Development Team

An SPP Organizational Group assigned or choosing to develop an SPP Regional Reliability Standard for submission to the ERO for approval for enforcement.

Terminated Member

An entity that was a Signatory to the Membership Agreement but whose membership in SPP has been terminated under Section 4 of the Membership Agreement.

Transmission Owning Member

A Member that has placed more than 500 miles of non-radial facilities operated at or above 60 kV under the independent administration of SPP for the provision of regional transmission service as set forth in the Membership Agreement.

Transmission Using Member

A Member that does not meet the definition of a Transmission Owning Member.

Western Area Power Administration-Upper Great Plains Region ("Western-UGP")

A division of the Western Area Power Administration that markets and transmits Federal power from reservoir projects under the control of the Department of the Army or the U.S. Bureau of Reclamation to Statutory Load Obligations, including preference power customers in Iowa, Minnesota, Montana, Nebraska, North Dakota, and South Dakota located in a defined marketing area. Western-UGP operates the WAUW Balancing Authority Area in the Western Interconnection, where certain of its transmission facilities are located.

2.1 Qualifications

Membership in SPP is voluntary and is open to any electric utility, Federal power marketing agencyPower Marketing Agency, transmission service provider, any entity engaged in the business of producing, selling and/or purchasing electric energy for resale, and any entity willing to meet the membership requirements, including execution of the Membership Agreement. Membership also is open to any entity eligible to take service under the SPP Open Access Transmission Tariff (OATT). These entities desire the greater efficiency and service reliability gained through better coordination by voluntary association in SPP as constituted herein and in the SPP Articles of Incorporation. Members recognize that such association has a significant effect upon the availability and reliability of the bulk electric power supply of the region, and thereby affects the reliability of the nation's electric power supply.

3.15 Liability, Insurance and Indemnification

For purposes of this section "SPP" refers to SPP and its officers, directors, Regional Entity Trustees, employees or agents, and "Member" refers to the Members of SPP as defined in these Bylaws. None of the provisions of this section, including the waiver of liability in Section 3.15.1 below, absolving SPP or its Members, directors, Regional Entity Trustees, officer, agents, employees or other representatives of liability or any provisions for insurance or indemnification apply to actions which are unlawful, undertaken in bad faith, or are the result of gross negligence or willful misconduct.

3.15.1 Waiver of Liability

- (a) SPP shall not be liable to any Member for damages arising out of or related to any directive, order, procedure, action, or requirement of SPP, under the then effective Bylaws and Criteria.
- (b) No Member shall be liable to any other Member or to SPP for damages arising out of or related to any action by the Member pursuant to any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria.
- (c) Each Member waives any future claim it might have against SPP or other Members arising out of or resulting from any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria.
- (d) SPP waives any future claim it might have against any Member arising out of or resulting from any actions taken by a Member pursuant to any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria.

3.15.2 Insurance

The President is authorized to procure insurance to protect SPP, its directors, Regional Entity Trustees, officers, agents, employees, or other representatives against damages arising out of or related to any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria or pursuant to the OATT.

3.15.3 Indemnification of Directors, Officers, Agents and Employees

Except for actions which are unlawful, undertaken in bad faith, or are the result of gross negligence or willful misconduct, SPP shall indemnify its directors, officers, agents, employees, or other representatives to the maximum extent allowed by law consistent with these Bylaws. Each director, Regional Entity Trustee, officer, agent, employee, or other representative of SPP shall be indemnified by SPP against all judgments, penalties, fines, settlements, and reasonable expenses, including legal fees, incurred as a result of, or in connection with, any threatened, pending or completed civil, criminal, administrative, or investigative proceedings to which the incumbent may be made a party by reason of acting or having acted in official capacity as a director, Regional Entity Trustee, officer, agent, employee, or representative of SPP, or in any other capacity which the incumbent may hold at the request of SPP, as its representative in any other organization, subject to the following conditions:

- (a) Such director, Regional Entity Trustee, officer, agent, employee, or other representative must have acted in good faith and, in the case of criminal proceedings, must have had no reasonable cause to believe that conduct was unlawful; provided, that SPP shall not provide indemnification of any conduct judged unlawful in criminal proceedings. When acting in official capacity, the incumbent must have reasonably believed that conduct was in the best interests of SPP, and, when acting in any other capacity, must have reasonably believed that conduct was at least not opposed to the best interests of SPP.
- (b) If the proceeding was brought by or on behalf of SPP, however, indemnification shall be made only with respect to reasonable expenses referenced above. No indemnification of any kind shall be made in any such proceeding in which the director, Regional Entity Trustee, officer, agent, employee, or other representative shall have been adjudged liable to SPP.

- (c) In no event, however, will indemnification be made with respect to any described proceeding which charges or alleges improper personal benefit to a director, Regional Entity Trustee, officer, agent, employee, or other representative and where liability is imposed on the basis of the receipt of such improper personal benefit.
- (d) In order for any director, Regional Entity Trustee, agent, employee, or other representative to receive indemnification under this provision, the person shall vigorously assert and pursue any and all defenses to those claims, charges, or proceedings covered herein which are reasonable and legally available and shall fully cooperate with SPP or any attorneys involved in the defense of any such claim, charges, or proceedings on behalf of SPP.
- (e) No indemnification shall be made in any specific instance until it has been determined by SPP that indemnification is permissible in that specific case, under the standards set forth herein and that any expenses claimed or to be incurred are reasonable. These two (2) determinations shall be made by a majority vote of at least a quorum of the Board of Directors consisting solely of directors who were not parties to the proceeding for which indemnification or reimbursement of expenses is claimed. If such a quorum cannot be obtained, a majority of at least a quorum of the full Board of Directors, including directors who are parties to said proceeding, shall designate a special legal counsel who shall make said determinations on behalf of SPP.
- (f) Any reasonable expenses, as shall be determined above, that have been incurred by a director, Regional Entity Trustee, officer, agent, employee, or other representative who has been made a party to a proceeding as defined herein, may be paid or reimbursed in advance upon a majority vote of a quorum of the full Board of Directors, including those who may be a party to the same

proceeding. However, such director, Regional Entity Trustee, officer, agent, employee, or other representative shall have provided SPP with (i) a written affirmation under oath that the incumbent, in good faith, believes the conditions of indemnification herein have been met; and (ii) a written undertaking that the incumbent shall repay any amounts advanced, with interest accumulated at a reasonable rate, if it is ultimately determined that such conditions are not met.

3.15.4 Limitations

The provisions of this section 3.15 are subject to applicable state and federal Federal laws, if any, which limit the ability of a Member to waive liability or enter into agreements of indemnity. Any benefits under this Section 3.15 shall not extend to any Member so limited by state or federal Federal law in complying with the provisions thereof.

3.15.5 Modification of Rights by Agreement

Any provision of this Section 3.15.1 may be waived or modified by express written agreement between SPP and Member. Such express written agreement shall apply solely to the subject matter of the agreement and is not intended to be a general waiver or modification of the rights provided in Section 3.15.1.

3.15.6 Procedural Rights Not Affected

The limitations of liability provided in Section 3.15.1 shall not affect any procedural rights or obligation a Member may have at law or equity.

5.1 Members Committee

The Members Committee shall work with the Board of Directors to manage and direct the general business of SPP. Its duties shall include, but are not limited to the following:

- (a) Provide individual and collective input to the Board of Directors, including but not limited to a straw vote from the Members Committee representatives as an indication of the level of consensus among Members, on all actions pending before the Board of Directors; and
- (b) Serve on committees reporting to the Board of Directors as appointed by the Board of Directors; and
- (c) Provide input with the Board of Directors to the Regional Entity Trustees on SPP Regional Reliability Standards presented by the MOPC to the Trustees or otherwise developed under the auspices of the Trustees for submission to the ERO for its approval.

5.1.1 Composition and Qualifications

5.1.1.1 Composition

Provided that Membership is sufficient to accommodate these provisions, the Members Committee shall consist of up to 1924 persons. FourSix representatives shall be investor owned utilities Members; fourfive representatives shall be cooperatives Members; two representatives shall be municipals Members (including municipal joint action agencies); three representatives shall be independent power producers/marketers Members; two representatives shall be state/federal power agencies Members; one representative shall be from a Federal Power Marketing Agency; two representatives shall be alternative power/public interest Members; one representative shall be from an independent transmission company Member, defined as having assets under the OATT and no Affiliate Relationships in other categories of Membership; one representative shall be a large retail customer Member; defined as non-residential end-use customers with individual or aggregated loads of 1-MW or more; and one representative shall be a small retail customer Member, defined as residential customers and other

customers with individual or aggregated loads of less than 1-MW. Representatives will be elected in accordance with Section 5.1.2 of these Bylaws.

5.1.1.2 Qualifications

A representative shall be an officer or employee of a Member with decision-making responsibility over SPP related activities, and must be the Member's representative to the Membership.

5.1.2 Term and Election

Representatives shall be nominated by the Corporate Governance Committee and elected each year at the meeting of Members to staggered three-year terms commencing upon election and continuing until their duly elected successors take office. The election process shall be as follows:

- (a) At least 90 calendar days prior to the meeting of Members at which election of new representatives is required, the Corporate Governance Committee shall nominate persons equal in number to the representatives to be elected;
- (b) At least 30 calendar days prior to the meeting of Members, the Corporate Governance Committee shall determine the persons it nominates for election as representatives, specifying the nominee for any vacancy to be filled. The Corporate Secretary shall prepare the ballot accordingly, leaving space for additional names, and shall deliver same to Members at least two weeks prior to the meeting of Members;
- (c) For purposes of electing and removing representatives only, each group of Members with Affiliate Relationships shall be considered a single vote;
- (d) At the meeting of Members, any additional nominee or nominees may be added to the ballot if a motion is made and seconded to add such nominee or nominees; and
- (e) The required number of representatives shall be elected by written ballot.

 A Member shall be entitled to cast a number of votes equal to the number of representatives to be elected. A Member may not cumulate votes. The candidates in each sector receiving the greatest number of votes will fill vacancies.

5.1.3 Resignation and Removal of Members Committee Representatives

Any representative may resign by written notice to the President noting the effective date of the resignation. A representative may be removed, with cause, by the affirmative vote of a majority of the Members at a meeting of Members. Removal proceedings may only be initiated by a petition signed by not less than twenty percent of the Members. The petition shall state the specific grounds for removal and shall specify whether the removal vote is to be taken at a special meeting of Members or at the next regular meeting of Members. A representative who is the subject of removal proceedings shall be given fifteen days to respond to the Member petition in writing to the President.

5.1.4 Vacancies

If a vacancy occurs the Corporate Governance Committee may elect an interim representative from the same sector to serve until a replacement representative from the same sector is elected and takes office. A special election shall be held at the next meeting of Members to fill the vacancy for the unexpired term. The replacement representative shall take office immediately following the election.

5.1.5 Meetings

The Members Committee shall meet only with the Board of Directors.

6.6 Corporate Governance Committee

The Corporate Governance Committee is responsible for the overall governance structure, including nominations, for the company in accordance with its scope as approved by the Board of Directors.

To the extent that the membership allows, the CGC shall be comprised of nineten members. One representative shall be the President of SPP who will serve as the Chair; one representative shall be the Chairman of the Board, unless his/her position is under consideration, in which case the Vice Chairman of the Board; one representative shall be representative of and selected by investor owned utilities Members; one representative shall be representative of and selected by municipals Members; one representative shall be representative of and selected by independent power producers/marketers Members; one representative shall be representative of and selected by state/federal power agencies Members; one representative shall be representative of and selected by alternative power/public interest Members; and one representative shall be representative of and selected by large/small retail Members; and one representative shall be representative of and selected by a Federal Power Marketing Agency Member(s).

Where a vacancy occurs with respect to a representative of a sector, the representatives from the appropriate sector will fill the vacancy. For purposes of selecting or removing representatives only, each group of Members with Affiliate Relationships shall be considered a single Member.

The CGC shall meet at least once per calendar year, and additionally as needed, provided that a quorum, as defined in these Bylaws, is present. The CGC shall report to the Board of Directors following each CGC meeting with respect to its activities and with such recommendations, as the CGC deems necessary.

8.7 Financial Obligation of Withdrawing Members

8.7.1 Existing Obligations

"Existing Obligations" are the following:

- a. Member's unpaid annual membership fee.
- b. Member's unpaid dues, assessments, and other amounts charged under Section 3.8 of the Membership Agreement, section 8.4 of the Bylaws, or otherwise under the Bylaws, plus the Member's share of costs SPP customarily includes in such dues, assessments or other charges, but which as of the Termination Date SPP had not included in such dues, assessments or other charges.
- c. Member's share (computed in accordance with the Bylaws) of the entire principal amounts of all SPP Financial Obligations outstanding as of the Termination Date. "Financial Obligations" are all long-term (in excess of six (6) months) financial obligations of SPP, including but not limited to the following:
 - debts under all mortgages, loans, loan agreements, borrowings, promissory notes, bonds, and credit lines, under which SPP is obligated, including principal and interest;
 - ii. all payment obligations under equipment leases, financing leases, capital leases, real estate and office space leases, consulting contracts, and contracts for outsourced services;
 - iii. any unfunded liabilities of any SPP employee pension funds, whether or not liquidated or demanded; and
 - iv. the general and administrative overhead of SPP for a period of three (3) months.
- d. Any costs, expenses or liabilities incurred by SPP directly due to the Termination, regardless of when incurred or payable, and including without limitation prepayment premiums or penalties arising under SPP Financial Obligations.

e. Member's share (computed in accordance with the Bylaws) of all interest that will become due for payment with respect to all interest bearing Financial Obligations after the Termination Date and until the maturity of all Financial Obligations in accordance with their respective terms ("Future Interest"). In the event that a Financial Obligation carries a variable interest rate, the interest rate in effect at the Termination Date shall be used to calculate the applicable Future Interest. In determining the Member's share of Future Interest, SPP shall take into account any reduction of Financial Obligations due to mitigation under this Section.

8.7.2 Computation of a Member's Existing Obligations

For purposes of computing the Existing Obligations of any withdrawing or terminated Member in accordance with the Membership Agreement, such "Member's share" is a percentage calculated as follows:

A = 100 [0.25(1/N) + 0.75(B/C)]

Where: A = Member's share (expressed as a percentage)

N = Total number of Members

B = The Member's previous year Net Energy for Load

C = Total of factor B for all Members

The Finance Committee shall have the discretion to reduce the Existing Obligations of any withdrawing or Terminated Member, to reflect any SPP costs or expenses that may be mitigated in connection with such Member's withdrawal or termination. In the event of consolidation of affiliate memberships or the transfer of membership from one corporate entity to another, whereby one entity remains a member of SPP, the withdrawal obligation for the departing company(ies) may be waived at SPP's sole discretion.

8.7.3 Financial Obligations for Transmission Facilities

To the extent that Section 4.3.3A of the Membership Agreement is applicable, a Terminated Member shall remain financially responsible for all financial obligations incurred and costs allocated to its load for transmission facilities approved prior to the Termination Date.

8.7.4 Penalty Costs

A Terminated Member shall remain liable for its share of costs associated with penalties assessed against SPP by FERC, the FERC-approved Electric Reliability Organization, any Electric Reliability Organization-approved Regional Entity, or any other governmental or regulatory authority with jurisdiction over SPP that SPP incurs as a result of events that occurred prior to Member's Termination Date but that SPP is unable to recover under the SPP OATT.

8.7.5 Limitation on Financial and Penalty Obligations

- (a) Notwithstanding the delineation of Members' financial obligations in Section 8.7, a Federal Power Marketing Agency shall not be subject to the financial obligations listed in this Section 8.7 in the event FERC finds that SPP has not adhered to all of the Federal Power Marketing Agency Amendments as that term is defined in Section 1.0 of these Bylaws or if SPP files and FERC approves material changes to the Federal Power Marketing Agency Amendments.
- (b) Provided further, notwithstanding any language to the contrary in these Bylaws, a Federal Power Marketing Agency has not waived or conceded any defense it may have, including sovereign immunity, intergovernmental immunity, or lack of subject matter jurisdiction in any action against it by an Enforcement Authority, nor has it accepted any liability, responsibility, or obligation to pay any civil monetary penalties or fines imposed by an Enforcement Authority to which it would not have been subject in the absence of these Bylaws. SPP, in accepting Western-UGP as a member, does not thereby concede or accept responsibility for any portion of a penalty or fine attributable to the actions or omissions of Western-UGP. SPP will identify the amount of any penalty or fine that SPP allocates to Western-UGP or that SPP determines is attributable to Western-UGP and will identify that amount to FERC as uncollectable and not otherwise owed by SPP. Enforcement Authority in these Bylaws means the Federal Energy Regulatory Commission (FERC), Electric Reliability Organization (ERO), or Regional Entities with enforcement

authority pursuant to a delegation from an ERO or FERC for the purpose of proposing and enforcing reliability standards.

10.0 Amendments To These Bylaws, The Articles Of Incorporation, And Membership Agreement

Except for modifications to Section 4.0 BOARD OF DIRECTORS, Section 5.0 COMMITTEES ADVISING THE BOARD OF DIRECTORS, Section 8.7.5 LIMITATION ON FINANCIAL AND PENALTY OBLIGATIONS, Section 9.0 REGIONAL ENTITY FUNCTION, and Section 10.0 AMENDMENTS, these Bylaws may be amended, repealed, or added to by the Board of Directors only upon 30 days written notice to the Membership of the proposed modification(s). Approval of amendments to the Bylaws by the Board of Directors must be by an affirmative vote of at least five directors. Sections 4.0, 5.0, 8.7.5, 9.0, and 10.0 of these Bylaws and the Articles of Incorporation may be amended, repealed, or added to only by approval of the Membership. Provided, that all changes to Federal Power Marketing agency/administration representation in Section 5.1.1 and any change to Section 8.7.5 must be mutually agreed to by the Federal Power Marketing Agency Member and SPP. All amendments are subject to the requisite regulatory approval(s).

MEMBERSHIP AGREEMENT

	TABLE OF CONTENTS	
1.0	Definitions	
2.0	Rights, Powers and Obligations of SPP 2.1 Operation and Planning	

- 2.1.1 General
- 2.1.2 Reliability
- 2.1.3 Transmission Maintenance
- 2.1.4 Generation Maintenance
- 2.1.5 Planning Activities
- 2.2 Non-Discriminatory Transmission Service
 - 2.2.1 Pricing
 - 2.2.2 Standards of Conduct
 - 2.2.3 OASIS
 - 2.2.4 Ancillary Services
 - 2.2.5 Transmission Service Scheduling
- 2.3 Fiduciary Responsibilities and Duties of SPP to Members
- 2.4 Additional Obligations of SPP
 - 2.4.1 Inspection and Auditing Procedures
 - 2.4.2 Stranded Cost Recovery Charges

3.0 Commitments, Rights, Powers, and Obligations of Member

- 3.1 Redispatch and Curtailment
- 3.2 Transmission and Generation Maintenance Practices
- 3.3 Construction
- 3.4 Use of Distribution Facilities
- 3.5 Providing Information
- 3.6 Facilities Access
- 3.7 Inspection and Auditing Procedures
- 3.8 Compliance and Bylaws and Other Policies and Procedures
- 3.9 Planning and Participation
- 3.10 Pricing
- 3.11 No Waiver of Jurisdictional Immunity
- 3.12 Compliance with Federal or State Law

4.0 Termination of Membership

- 4.1 Events of Termination and Partial Termination
- 4.2 Termination Procedures and Effective Dates
 - 4.2.1 Notice of Voluntary Withdrawal
 - 4.2.2 Effective Date of Termination
- 4.3 Obligations Upon Termination
 - 4.3.1 Obligation to Hold Users Harmless
 - 4.3.2 Obligation to Pay Current and Existing Obligations

- 4.3.3 Construction of Transmission Facilities
- 4.3.3A Financial Obligations for Transmission Facilities
- 4.3.4 Regulatory and Other Approvals or Procedure
- 5.0 Regulatory, Tax, And Other Authorities
 - 5.1 Regulatory and Other Authorities
 - 5.2 Tax Authorities
 - 5.3 Effectiveness as to Certain Members
- 6.0 Removal of Members
- 7.0 Effective Date, Duration, and Transition
- 8.0 Miscellaneous Provisions
 - 8.1 Governing Law
 - 8.2 Successors and Assigns
 - 8.3 No Implied Waivers
 - 8.4 Severability
 - 8.5 Renegotiation
 - 8.6 Representations and Warranties
 - 8.7 Further Assurances
 - 8.8 Delivery of Notices
 - 8.9 Entire Agreement
 - 8.10 Good Faith Efforts
 - 8.11 Third Party Joint Agreements
 - 8.12 Amendment
 - 8.13 Counterparts

This Agreement is made between the Member and SPP, as defined herein.

1.0 Definitions

Agreement

This Membership Agreement.

Basin Electric Amendments

The amendments and revisions to the SPP OATT, to Basin Electric Power Cooperative's ("Basin Electric's") Network Integration Transmission Service Agreement, or to Basin Electric's Network Operating Agreement necessary for Basin Electric's initial SPP membership or as they may be revised in the future by agreement between Basin Electric and SPP.

Board of Directors

The Board of Directors elected pursuant to the Bylaws.

Bylaws

SPP's Bylaws or any successor document.

Distribution Facilities

Facilities that are the subject of a separate distribution charge pursuant to the Open Access Transmission Tariff.

Eastern Interconnection

One of the three major alternating-current electrical grids in North America. The Eastern Interconnection reaches from Central Canada eastward to the Atlantic coast (excluding Quebec), south to Florida, and back west to the foot of the Rockies (excluding most of Texas).

Effective Date

This Agreement is effective on January 1, 2000 or upon the date of execution by Member if after January 1, 2000.

Electric Transmission System

The transmission facilities subject to SPP's tariff administration, except for any Distribution Facilities.

Existing Obligations

Shall have the meaning given in Section 4.3.2(b).

Federal Power Marketing Agency

This term shall include the term "Federal Power Marketing Administration" and have the same definition that is set forth in the Federal Power Act at 16 U.S.C. § 796(19), which defines "Federal power marketing agency" as "any agency or instrumentality of the United States (other than the Tennessee Valley Authority) which sells electric energy[.]"

Federal Power Marketing Agency Amendments

The amendments and revisions to the SPP Bylaws, the SPP Membership Agreement, and Section 39.3 of the OATT that are required by a Federal Power Marketing Agency for initial SPP membership or as they may be revised in the future by mutual agreement between a Federal Power Marketing Agency and SPP.

Federal Power-Western-UGP

All capacity and energy generated at reservoir projects under the control of the Department of the Army or the Bureau of Reclamation in the marketing area of the Western-UGP for the purpose of fulfilling Western-UGP's Statutory Load Obligations for the sale of capacity and energy. This shall also include any capacity and energy delivered to or from Western-UGP under the pre-OATT bi-directional agreement with Southwestern Power Administration through Associated Electric Cooperative, Inc. for delivery and receipt at the Maryville Substation. Western-UGP's deliveries to Southwestern shall be considered part of Western's Statutory Load Obligations, and receipts from Southwestern to Western-UGP will be considered as coming from Federal resources. Federal Power-Western-UGP resources shall be eligible to be considered as Designated Resources.

FERC

The Federal Energy Regulatory Commission or successor organization.

Financial Obligations

Shall have the meaning given in Section 4.3.2(b).

Future Interest

Shall have the meaning given in Section 4.3.2(b).

Good Utility Practice

Any of the practices, methods, and acts engaged in or approved by a significant portion of the electric utility industry during the relevant time period, or any of the practices, methods, and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety, and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act, to the exclusion of all others, but rather to be a range of acceptable practices, methods, or acts generally accepted in the region. SPP Criteria and NERC Policies and Standards are considered Good Utility Practice.

Heartland Amendments

The amendments and revisions to the SPP OATT, to Heartland Consumers Power District's ("Heartland's") Network Integration Transmission Service Agreement, or to Heartland's Network Operating Agreement necessary for Heartland's initial SPP membership or as they may be revised in the future by agreement between Heartland and SPP.

Member

Signatory to this Agreement that has completed the application requirements pursuant to the Bylaws.

NERC

North American Electric Reliability Corporation or successor organizations.

Non-Transmission Owner

A Member that is not a Transmission Owner. A Non-Transmission Owner that owns or controls Tariff Facilities may have its status changed to a Transmission Owner under this Agreement upon notice to SPP and execution of this Agreement as a Transmission Owner.

Open Access Transmission Tariff (OATT)

The SPP nondiscriminatory, Open-Access Transmission Tariff (OATT) on file with FERC pursuant to Section 205 of the Federal Power Act under which SPP will offer transmission service, or any such successor tariff.

Partial Termination

Shall have the meaning given in Section 4.1.

Regional Entity

An entity having enforcement authority delegated to it by NERC pursuant to a delegation agreement accepted by FERC.

Reliability Coordinator

SPP, in performing its reliability coordinator function as recognized by NERC pursuant to its policies, and pursuant to SPP Criteria and this Agreement.

SPP

Southwest Power Pool, Inc., or successor organization.

SPP Criteria

SPP's approved operating and planning criteria.

SPP Region

The geographic area encompassing the transmission systems of Members that are Transmission Owners.

Standards of Conduct

SPP's Standards of Conduct that apply to the conduct of its directors, officers, employees, Regional Entity trustees, contractors, and agents.

Statutory Load Obligations

Western-UGP's power marketing function obligations under Federal law to deliver capacity and energy from the output of the Federal hydroelectric projects operated by the Department of the Army and the Bureau of Reclamation to loads which include project use loads, preference power customer loads defined pursuant to a power marketing plan, and other loads required to be served under Federal law.

Tariff Facilities

The Electric Transmission System and the Distribution Facilities subject to SPP's tariff administration.

Termination

Shall have the meaning given in Section 4.1.

Termination Date

Shall mean the date of Termination is effective in accordance with Section 4.2.2(b).

Transmission Customer

A customer under the Open Access Transmission Tariff

Terminated Member

An entity that was a Signatory to this Agreement but whose membership in SPP has been terminated under Section 4 of this Agreement.

Transmission Owner

A signatory to this Agreement which: (1) transfers functional control of Tariff Facilities related to the rates, terms and conditions of the OATT to SPP by executing this Agreement; or (2) appoints SPP under another agreement to provide service under the Transmission Tariff over Tariff Facilities which it owns or controls; or (3) is assigned by SPP to construct and accepts the obligation to construct new Tariff Facilities; or (4) undertakes another Transmission Owner's obligation to construct Tariff Facilities in accordance with Section 3.3(c) of this Agreement and Attachment O of the SPP OATT.

Upper Missouri Zone ("UMZ")

The Upper Missouri Zone ("UMZ" or "Zone 19") is the rate pricing zone initially consisting of the following facilities that meet the requirements of Attachment AI, upon the transfer of those facilities to the functional control of the Transmission Provider: (i) the facilities of Western-UGP within the Eastern and Western Interconnections; (ii) the facilities owned or leased by Basin Electric Power Cooperative or Heartland Consumers Power District within the Eastern Interconnection; (iii) a portion of the facilities owned or leased by Basin Electric Power Cooperative within the Western Interconnection; and (iv) other facilities of the Western Area Power Administration transferred to the functional control of the Transmission Provider.

Western Area Power Administration-Upper Great Plains Region ("Western-UGP")

A division of the Western Area Power Administration that markets and transmits Federal power from reservoir projects under the control of the Department of the Army or the U.S. Bureau of Reclamation to Statutory Load Obligations, including preference power customers in Iowa, Minnesota, Montana, Nebraska, North Dakota, and South Dakota located in a defined marketing area. Western-UGP operates the WAUW Balancing Authority Area in the Western Interconnection, where certain of its transmission facilities are located.

Western Interconnection

One of the three major alternating-current electrical grids in North America. The Western Interconnection stretches from Western Canada South to Baja California in Mexico, reaching eastward over the Rockies to the Great Plains.

2.1 Operation and Planning

2.1.1 General

- (a) SPP shall schedule transactions and administer transmission service over Tariff Facilities as necessary to provide service in accordance with the SPP OATT.
- (b) SPP shall function in accordance with Good Utility Practice and shall conform to applicable reliability criteria, policies, standards, rules, regulations, guidelines and other requirements of SPP and NERC, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements specified in this paragraph), and all applicable requirements of **F**ederal and state regulatory authorities.
- (c) SPP shall maintain a publicly available registry of all facilities that are not classified as critical energy infrastructure information that constitute the Electric Transmission System.
- (d) SPP shall review and approve, as appropriate, requests for service, schedule transmission transactions, and determine available transfer capability under the OATT, provided that SPP shall coordinate with the Transmission Owner when processing requests for service involving its Tariff Facilities.
- (e) SPP shall be responsible for coordinating with neighboring regional organizations and/or non-member transmission owners or providers as appropriate.
- (f) SPP shall not exercise its administration of transmission service over the Tariff Facilities in such a way as to interfere with contracts between Transmission Owner and any Transmission Customer that are in effect as of the Effective Date of this Agreement except as permitted by the OATT.
- (g) SPP shall be responsible for documenting all transmission service requests, the disposition of such requests, and any supporting data required to support the decision with respect to such requests. SPP

- shall negotiate as appropriate to develop reciprocal service, equitable tariff application, compensation principles, and any related arrangements.
- (h) SPP shall propose and file with FERC pursuant to Section 205 of the Federal Power Act modifications to the OATT and make any other necessary filings subject to approval by the Board of Directors.
- (i) SPP shall develop penalties and incentives, subject to FERC filings where appropriate.
- (j) SPP shall direct Transmission Owner pursuant to the provisions of Section 3.3 to construct transmission facilities in accordance with coordinated planning criteria, or if necessary under the OATT.
- (k) SPP shall have the authority to direct the day-to-day operations of the Tariff Facilities in order to carry out its responsibilities as a Transmission Provider and Reliability Coordinator; provided, however, nothing in this Agreement or the OATT shall be construed to require a change in the physical control of any Tariff Facilities using a Party's existing facilities or equipment.
- (1) SPP shall take any actions necessary for it to carry out its duties and responsibilities, subject to receiving any necessary regulatory approvals and any necessary approvals from the Board of Directors.

2.1.2 Reliability

SPP shall have responsibility for reliability of the Electric Transmission System in connection with its rights, powers, and obligations under this Agreement. SPP shall act as the Reliability Coordinator of the Electric Transmission System, and as such, shall have reliability monitoring and emergency response responsibilities pursuant to related SPP Criteria and the following requirements:

(a) SPP shall monitor real-time data to determine whether any control areas are experiencing generation capacity deficiencies. If a

generation capacity deficiency event threatens the security of the Electric Transmission System, SPP is authorized to and shall direct the acquisition of generation capacity and, if that direction is not satisfied, is authorized to and shall direct the shedding of firm load in the deficient control area.

- (b) SPP shall work with other reliability coordinators and non-member transmission owners or providers to develop regional reliability plans and emergency operating procedures.
- (c) SPP shall maintain emergency response procedures for responding to specified critical contingencies and shall continuously analyze issues that may require the initiation of such actions.
- (d) SPP is authorized to and shall direct the response to any emergency and Members shall carry out the required emergency actions as directed by SPP (except in cases involving endangerment to the safety of employees or the public), including the shedding of firm load if required for regional reliability.
- (e) After the conclusion of an emergency condition, any affected entity that disagrees with SPP's handling of the emergency may resolve that disagreement pursuant to the dispute resolution procedures in the Bylaws.
- (f) SPP shall monitor and coordinate the maintenance of adequate Electric Transmission System voltage levels with control areas and Transmission Owner, where appropriate.
- (g) SPP shall direct redispatch of generation in accordance with the OATT and in its role as Reliability Coordinator, subject to the generator receiving appropriate compensation pursuant to an applicable rate schedule.

2.1.3 Transmission Maintenance

SPP is required to approve all planned maintenance of the Electric Transmission System consistent with the following requirements:

- (a) SPP shall review planned transmission maintenance schedules submitted by Transmission Owner for a minimum of a rolling one-year period. These planned maintenance schedules shall be updated daily. Planned transmission maintenance requests shall be submitted to SPP at least one week in advance of an outage.
- (b) SPP shall analyze a planned transmission maintenance request to determine its effect on available transfer capability, ancillary services, the reliability of the Electric Transmission System, and any other relevant effects. Within two business days of receiving a planned maintenance request, SPP shall provide a response. If SPP's response indicates that such planned transmission maintenance will have an adverse impact, Transmission Owner shall work with SPP to minimize the impact of such planned maintenance, up to and including re-scheduling the maintenance.
- (c) SPP shall notify Transmission Owner of the need to change previously reviewed planned transmission maintenance outages if forced transmission outages or other circumstances compromise the integrity or reliability of the Electric Transmission System. If Transmission Owner is fully compensated for any additional costs resulting from any changes in maintenance schedules as provided in an applicable rate schedule, Transmission Owner shall revise maintenance outages to address such emergency circumstances.
- (d) As part of its review process, SPP shall identify planned transmission maintenance schedules that limit available transfer capability. If requested by a Transmission Customer, SPP shall identify opportunities and associated costs for rescheduling planned maintenance to enhance available transfer capability. Transmission Owner shall be compensated for the additional costs of rescheduled maintenance as provided in the SPP OATT.

- (e) SPP shall be responsible for documenting all planned transmission maintenance requests, the disposition of those requests, and all data supporting the disposition of each request.
- (f) SPP shall coordinate with Transmission Owner to the extent practicable to implement schedules for unplanned transmission maintenance when conditions endanger the safety of employees or the public, may result in damage to facilities, or may result in the unsatisfactory operation of its transmission system or any other transmission system.

2.1.4 Generation Maintenance

SPP shall coordinate the maintenance of generating units as appropriate to the extent such generation maintenance directly affects the capacity or reliability of the Electric Transmission System and the generation is located in the SPP Region as follows:

- (a) SPP shall review planned generating unit maintenance schedules submitted by generation owners for a minimum of a rolling one-year period. The planned maintenance schedules shall be updated daily. SPP shall keep such information confidential.
- (b) SPP shall analyze a planned generating unit maintenance schedule to determine its effect on available transfer capability, ancillary services, the reliability of the Electric Transmission System, and any other relevant effects. SPP shall inform a generation owner if its maintenance schedule is expected to have an impact on the reliability of the Electric Transmission System.
- (c) As part of its review process, SPP shall identify generating unit maintenance schedules that limit available transfer capability and shall identify opportunities and associated costs for rescheduling planned maintenance to enhance available transfer capability.
- (d) A generation owner that changes planned maintenance at the request of SPP pursuant to this Section 2.1.4 shall be compensated in accordance with the SPP OATT.

(e) SPP shall be responsible for documenting all planned generating unit maintenance schedules, all schedule changes, and all SPP studies and services performed with respect to planned generation maintenance.

2.1.5 Planning Activities

- (a) SPP shall engage in such planning activities, in coordination with Member, as are necessary to fulfill its obligations under this Agreement, SPP Criteria and the OATT. Such planning shall conform to applicable reliability requirements of SPP, NERC, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements), and all applicable requirements of Federal or state regulatory authorities. Such planning shall seek to minimize costs, consistent with the reliability and other requirements set forth in this Agreement. The division of responsibility for planning between Member and SPP is set forth in the SPP Criteria.
- (b) As part of its planning activities, SPP shall be responsible for planning, and for directing or arranging, necessary transmission expansions, additions, and upgrades that will enable it to provide efficient, reliable and non-discriminatory transmission service and to coordinate such efforts with the appropriate state authorities.

2.4 Additional Obligations of SPP

2.4.1 Inspection and Auditing Procedures

SPP shall grant Member, its employees, agents, or external auditors, and federal and state regulatory authorities having jurisdiction over SPP or Member, such access to SPP's books, records, business practices, control procedures and required audit test results, and related financial transactions and settlement activities as is necessary to verify compliance by SPP with this Agreement, to audit and verify transactions under this Agreement, and to assist Member in complying with its statutory and regulatory requirements. Such access shall be at reasonable times and under reasonable conditions. SPP shall also comply with the reporting requirements of federal and state regulatory authorities having jurisdiction over SPP with respect to the business aspects of its operations. Contact between officers, employees, and agents of Member and those of SPP shall comply with the Standards of Conduct.

2.4.2 Stranded Cost Recovery Charges

SPP shall collect and distribute, as appropriate, any stranded cost recovery charges pursuant to applicable schedules accepted by appropriate regulatory entities.

3.0 Commitments, Rights, Powers, and Obligations of Member

Member has made the following commitments, and shall have the following rights and shall be responsible for the following functions, some of which apply only to a Transmission Owner, some only to a Non-Transmission Owner.

- (a) Transmission Owner shall transfer functional control related to the rates, terms and conditions of the OATT of its Transmission Facilities, subject to receiving all necessary regulatory authorizations, thereby allowing SPP to (i) direct the operation of the Transmission Facilities in accordance with the terms of this Agreement; (ii) administer transmission service under the Transmission Tariff over that Transmission Owner's Tariff Facilities; and (iii) receive funds from Transmission Customers relating to transmission service over Tariff Facilities and distribute funds to the Transmission Owner. Where Member, owns generators within the SPP Region which directly affect the capacity or reliability of the Electric Transmission System, it shall offer to provide the ancillary services required under the OATT at rates approved by regulatory authorities, where appropriate, to the extent such generators are able to provide such ancillary services.
- (b) Transmission Owner shall operate and maintain its Tariff Facilities subject to the requirements of this Agreement.
- (c) Where Transmission Owner is a balancing area operator, it shall continue to operate its balancing areas for local generation control and economic dispatch, and shall be responsible for identifying and addressing local problems in a reliable manner.
- (d) Transmission Owner shall provide transmission service over its Tariff Facilities at the direction of SPP pursuant to the terms of the OATT.
- (e) Member agrees to comply with the instructions of SPP in its role as Reliability Coordinator.
- (f) Transmission Owner shall retain all rights of ownership, including legal and equitable title in its Tariff Facilities, subject to the provisions of this Agreement.

 Transmission Owner, or one acting under its authority, shall retain all rights to

- access to its Tariff Facilities so long as such access is consistent with the provisions of this Agreement.
- (g) Notwithstanding any other provision in this Agreement, Transmission Owner shall not be obligated or be considered to be allowing transmission over its facilities if such transmission would cause the loss of the tax-exempt status of Transmission Owner or any bonds or other debt of Transmission Owner.
- (h) Notwithstanding any other provisions of this Agreement, Member reserves the right to exercise operational authority of Member's Tariff Facilities (1) to protect public safety and the safety of its workers, to prevent damage to equipment, and to preserve reliability in compliance with NERC standards, and (2) as necessary to preserve Member's rights, duties and obligations regarding electric service to its retail and wholesale native load customers pursuant to itsany applicable Federal or state law and consistent with NERC standards, if SPP's exercise of operational authority over the Tariff Facilities would endanger said electric service or is contrary to or would curtail, surrender or delegate such Federal or state law rights, duties and obligations. Member will, as soon as reasonably practicable thereafter, notify SPP of such actions taken by Member. Member and SPP will meet and confer regarding the matter and, as necessary, negotiate in good faith to modify the Agreement to address the matter.

3.3 Construction

- (a) As part of its planning activities, SPP shall be responsible for planning, and for directing or arranging, necessary transmission expansions, additions, and upgrades that will enable it to provide efficient, reliable and non-discriminatory transmission service and to coordinate such efforts with the appropriate Federal or state authorities, including the Member's governing board where it serves as that authority, or in the case of a Federal Power Marketing Agency, the Administrator. Transmission Owner shall use due diligence to construct transmission facilities as directed by SPP in accordance with the OATT and this Agreement, subject to such siting, permitting, and environmental constraints as may be imposed by Federal, state, local and federallocal laws and regulations, and subject to the receipt of any necessary #Federal or state regulatory approvals, including, as necessary, the Member's governing board where it serves as that authority. Such construction shall be performed in accordance with Good Utility Practice, applicable SPP Criteria, industry standards, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements), and in accordance with all applicable requirements of fFederal or state regulatory authorities. Transmission Owner shall be fully compensated to the greatest extent permitted by FERC, or other legislative or regulatory authority for the costs of construction undertaken in accordance with the OATT.
- (b) After a new transmission project has received the required approvals and been approved by SPP, SPP will direct the appropriate Transmission Owner(s) to begin implementation of the project in accordance with Attachment O of the OATT.

3.10 Pricing

Transmission Owner shall possess the unilateral right to file with FERC, and, if the Transmission Owner is non-jurisdictional, the Transmission Owner shall have the option to file with FERC, or submit to SPP for filing with FERC, pursuant to Section 205 of the Federal Power Act and Federal statutory and regulatory requirements, including Delegation Order No. 00-37.00A, 10 CFR Part 903 and 18 CFR Part 300, as amended or superseded, modifications to change the rates or rate structure for transmission service over its Tariff Facilities, including filing a fixed revenue requirement and supporting data or a rate formula template for its cost of service revenue requirements, and to submit proposals or filings governing new construction with FERC; provided, however, Transmission Owner may not submit a proposal which results in a Transmission Customer paying two or more transmission charges for transmission for one transaction under the OATT (excluding Distribution Facilities for which an additional charge may be imposed, and Grandfathered Agreements as defined in the OATT). Transmission Owner shall notify SPP in advance of its intention to submit a filing to FERC and provide SPP with a copy of the filing. No approval from SPP is required for such filings.

3.11 No Waiver of Jurisdictional Immunity

If Member is not subject to the jurisdiction of FERC as a public utility under the Federal Power Act, Member shall not be required to take any action or participate in any filing or appeal that would confer FERC jurisdiction over Member that does not otherwise exist. Any order, decision, rule or regulation issued by FERC to SPP or any other Members or Member of SPP relating to matters exempt from FERC jurisdiction under Section 201(f) of the Federal Power Act shall not apply directly or separately to a non-jurisdictional Member. Without limiting the generality of the foregoing, except as otherwise provided in the Federal Power Act, a non-jurisdictional Member shall not be bound or obligated by any FERC order, decision, rule or regulation requiring a change in the rates, terms or conditions for transmission service or compensation for utilizing the transmission facilities of a non-jurisdictional Member, which conflicts with applicable Federal or state law, including any order requiring the suspension of the use of such rates, terms or conditions or the payment of refunds of rates or compensation previously collected or received. A non-jurisdictional Member and SPP acknowledge that FERC, in the context of its jurisdiction over SPP's rates, may review a non-jurisdictional Member's revenue requirement and rates to the extent they comprise or affect the rates charged by SPP or other Members. In the case of a Federal Power Marketing Agency, this review shall be consistent with the Delegation Order No. 00-037.00A, as superseded or amended, from the Secretary of Energy to the Power Marketing Administrations and the FERC, including the regulations implementing this review authority. If FERC does not accept a non-jurisdictional Member's revenue requirement or rates, the non-jurisdictional Member may terminate this Agreement pursuant to the withdrawal provisions of the Agreement. In such event, the non-jurisdictional Member and SPP agree to meet and confer prior to any termination of this Agreement. Nothing in this Agreement, or the participation of a non-jurisdictional Member in SPP and its operations waives any objection to or otherwise constitutes a consent to, the jurisdiction by FERC that does not otherwise exist over the non-jurisdictional Member or its transmission service, facilities and rates.

3.12 Compliance with Federal or State Law

Notwithstanding any other provision of this Agreement, a non-jurisdictional Member shall not be required to take any action or do any other thing with respect to rates, charges, terms or conditions of service, the resolution of disputes under this Agreement or any other matter regarding its obligations and performance under this Agreement, that (i) the non-jurisdictional Member is not permitted by Federal or state law to undertake or that is prohibited in whole or in part by any Federal or state law or regulation applicable to the non-jurisdictional Member; or (ii) would require the nonjurisdictional Member to violate a provision of such state or Federal law or regulation in order to comply with this Agreement. Determination of compliance with and permissible action, conduct or obligations under this Section 3.12 by a non-jurisdictional Member shall be within the sole jurisdiction of the non-jurisdictional Member's governing board, or in the case of a Federal Power Marketing Agency, its Administrator, subject to applicable Federal or state court review. A non-jurisdictional Member shall not object to SPP's participation in any Federal or state proceedings that impact the non-jurisdictional Member's ability to perform under this Agreement or determinations regarding such impact. To the extent possible without violating Federal or state law, a non-jurisdictional Member shall notify SPP in advance of any action that the non-jurisdictional Member is required to take that the non-jurisdictional Member believes would constitute a violation of Federal or state law, and the non-jurisdictional Member and SPP promptly shall meet and confer regarding the matter. As necessary, the non-jurisdictional Member and SPP agree to negotiate in good faith to modify the Agreement as consistent as possible with the original intent to allow SPP to exercise operational authority over the nonjurisdictional Member's Tariff Facilities as otherwise provided in the Agreement. If the non-jurisdictional Member and SPP are unable to resolve the matter, the nonjurisdictional Member may terminate this Agreement pursuant to the withdrawal provisions of the Agreement.

4.1 Events of Termination and Partial Termination.

A "Termination" shall mean any cessation of Membership, voluntary or involuntary, or a termination of this Agreement for any reason including the following:

- a. Member voluntarily withdraws from membership under Sections 4.0 or
 5.0 of this Agreement;
- b. An involuntary termination of membership occurs pursuant to Section 6.0 of this Agreement;
- c. Member withdraws from membership or terminates this Agreement to comply with the terms of any applicable law or regulation;
- d. A withdrawal from membership or termination of this Agreement is ordered by any court or administrative agency of competent jurisdiction; SPP reserves the right, but is not obligated, to maintain before such court or administrative agency, or on any appeal, that FERC has preemptive jurisdiction;
- e. A material breach or repudiation of this Agreement, in the discretion of the non-breaching or non-repudiating party;
- f. The liquidation or dissolution of SPP, unless a third party has assumed the rights and obligations of SPP under this Agreement consistent with Section 8.2, and has reasonably demonstrated capability to perform SPP's obligations under this Agreement;
- g. An agreement between SPP and the Member to terminate this Agreement.

A "Partial Termination" occurs upon a Member's voluntary removal of a portion of its transmission facilities or customers from the SPP Region, including, by way of example and not limitation, sale of a part of the Member's distribution or transmission network or transfer to another service provider of a portion of its retail load.

4.2 Termination Procedures and Effective Dates

4.2.1 Notice of Voluntary Withdrawal

- (a) Notice. Subject to Section 4.3, a Member may withdraw voluntarily from this Agreement, provided that it has given written notice to the President of its intent to withdraw. Notice of intent to withdraw must state a proposed date for the withdrawal and be delivered to the President no less than twenty-four (24) months prior to such date. The President will advise the Members and the Board of Directors of any withdrawal notices received. In order to assure that there is no more than one proposed termination date with respect to a Member, a withdrawal notice shall be deemed to supersede any prior withdrawal notice given by the Member, except that a Member may not submit a withdrawal notice less than twenty-four (24) months prior to the termination date proposed in the Member's previous notice of intent to withdraw. Voluntary withdrawal is a Termination and creates the same obligations as a Termination for any other reason. Upon receiving a notice of intent to withdraw, SPP shall account for such notice of intent to withdraw in the SPP planning process, unless the Member plans to continue to take transmission service from SPP after the termination date.
- (b) Withdrawal Deposit. A Member submitting a written notice of its intent to withdraw from this Agreement must simultaneously submit a cash withdrawal deposit to SPP, as set forth in the table below. SPP will not accept a notice of intent to withdraw without a withdrawal deposit. SPP will treat the withdrawal deposit as a pre-payment of a portion of the costs SPP incurs to process the Member's withdrawal from SPP, as set forth in Section 4.3.2(d) of this Agreement, or the costs associated with reintegrating the Member into SPP if the Member subsequently rescinds its notice of intent to withdraw and SPP incurs costs to reintegrate the Member. Withdrawal deposits are as follows:

Member Category	Withdrawal Deposit
Load Serving Entity	\$ 150,000
Non-Load Serving Entity	\$ 50,000

If the cost of processing Member's withdrawal as calculated by SPP pursuant to Section 4.3.2(d) of this Agreement exceeds the withdrawal deposit, the additional amount shall be included in the invoice SPP provides to the Member under 4.3.2(e) of this Agreement. If the Member rescinds

its notice of intent to withdraw and the cost of processing the Member's withdrawal and subsequent reintegration into SPP exceeds the withdrawal deposit, SPP shall invoice the Member for the amount of the cost that exceeds the deposit, and the Member shall provide payment to SPP within thirty (30) days of receipt of the invoice. If the withdrawal deposit exceeds the costs of processing the Member's withdrawal and/or reintegration, SPP shall refund the difference to the Member. Notwithstanding the foregoing, a Federal Power Marketing Agency shall not be required to make a withdrawal deposit and shall only be responsible for paying SPP costs after they are incurred and appropriately invoiced pursuant to 4.3.2(e).

4.2.2 Effective Date of Termination

- Owner subject to FERC jurisdiction, the Termination Date shall be the date proposed in the withdrawal notice under Section 4.2.1 or otherwise agreed by SPP. If the withdrawing Member is a Transmission Owner subject to FERC jurisdiction, the Termination Date shall be the later of (i) the proposed date specified in the withdrawal notice or otherwise agreed by SPP, (ii) the effective date, if any, set by the FERC order approving the withdrawal; or (iii) the date that such FERC order is no longer subject to review by a court of competent jurisdiction.
- (b) **Termination other than Voluntary Withdrawal**. If the Termination occurs for any reason other than the Member's voluntary withdrawal under Section 4.2.1 or by agreement with SPP, the Termination Date shall be as follows:
 - (i) If the Member is not a Transmission Owner subject to FERC jurisdiction, the Termination Date shall be the date of the event by which the Termination occurs, for example, the date a party gives notice that it will treat a breach or repudiation as a Termination or the date a Member withdraws in order to comply with the terms of a law or regulation. The foregoing notwithstanding, if the Termination occurs due to the order of a court or administrative agency, the Termination Date shall be the date the order is no longer subject to review by a court of competent jurisdiction.

- (ii) If the Member is a Transmission Owner requiring regulatory agency approval prior to effectively withdrawing from SPP, then the Termination Date shall be the later of (i) the effective date, if any, set by the regulatory agency order approving the Termination; or (ii) the date that such regulatory order is no longer subject to review by a court of competent jurisdiction.
- (iii) In conjunction with the requirements and limitations imposed in Section 4.2.2(b)(ii) above; Transmission Owning Members of SPP wishing to withdraw from SPP and requiring regulatory agency approval prior to such withdrawal being effective are required to initiate the required regulatory filings seeking approval to withdraw prior to the end of the 24-month notice period. Additionally, these members must provide SPP copies of the regulatory filings in a timely manner after filing with the appropriate regulatory agency. Failure to comply with the terms of this paragraph will effectively rescind the notice of the withdrawing member.
- (iv) Member may terminate this Agreement with less than the required twenty-four (24) months' notice, in the event that the Federal or state law governing Member changes, or any provisions of this Agreement or, the provisions of SPP's OATT, or SPP's Bylaws are changed or modified in a manner that causes a conflict with the Member's Federal or state law, regulations, or rate schedules, and the internal dispute resolution process described in Section 12 of the OATT is unable to resolve such conflict. In such event, Member and SPP shall meet and confer to facilitate the withdrawal as soon as practicable as necessary to ensure compliance with Federal or state law.
- (v) Any Member with Transmission Facilities located in the Upper

 Missouri Zone may terminate this Agreement with less than the
 required twenty-four (24) month notice in the event that Western-

UGP or Basin Electric Power Cooperative withdraws from SPP in accordance with its respective withdrawal rights or if FERC finds that SPP has not adhered to all of the Federal Power Marketing Agency Amendments, the Basin Electric Amendments, or the Heartland Amendments. In such event, Member and SPP shall meet and confer to facilitate the withdrawal as soon as practicable or as necessary to ensure compliance with state or Federal law. In the event of a withdrawal by Western-UGP or Basin Electric Power Cooperative, Member's withdrawal will become effective on the same date as that of Western-UGP or Basin Electric Power Cooperative. Such Member also may terminate this Agreement in the event that SPP files and FERC approves changes to the Federal Power Marketing Agency Amendments, the Basin Electric Amendments, or the Heartland Amendments that have a material adverse effect on such Member. If such Member exercises its withdrawal rights under this provision, the financial obligations will be calculated under § 4.3 of this Agreement.

4.3 Obligations Upon Termination

4.3.1 Obligation to Hold Users Harmless

Transmission Customers taking service which involves facilities being withdrawn by a Transmission Owner from the functional control of SPP and where such service is under transmission contracts executed before the Termination Date shall continue to receive the same service for the remaining term of each such contract at the same rates, terms, and conditions that would have been applicable if the Termination or Partial Termination had not occurred. Transmission Owner agrees to continue providing service to such Transmission Customers in accordance with the preceding sentence, and shall receive revenues calculated in accordance with the OATT but no more in revenues for that service that if there had been no Termination or Partial Termination.

4.3.2 Obligation to Pay Current and Existing Obligations

- (a) In the event of a Termination or Partial Termination, Member shall pay all obligations incurred under this Agreement at any time prior to the Termination Date. In addition, in order for SPP to recover a portion of certain debts and cost payable by SPP after the Termination Date as further specified in this Agreement, the Member shall pay all Existing Obligations (as defined herein) calculated as of the Termination Date. SPP shall make reasonable efforts to mitigate the Member's Existing Obligations by commercially reasonable actions (such as prepayment of allocable debt, or investment of part or all of the Member's payment in an interest-bearing instrument) and, in its discretion, may further discount the Member's Existing Obligations to reflect any additional mitigation SPP determines it will achieve.
- (b) "Existing Obligations" are all of the following and other obligations as may be set forth in the Bylaws from time to time;
 - i. Member's unpaid annual membership fee,
 - ii. Member's unpaid dues, assessments, and other amounts charged under Section 3.8 of this Agreement, Section 8.4 of the Bylaws, or otherwise under the Bylaws, plus the Member's share of costs SPP

- customarily includes in such dues, assessments or other charges, but which as of the Termination Date SPP had not included in such dues assessments or other charges.
- iii. Member's share (computed in accordance with the Bylaws) of the entire principal amounts of all SPP Financial Obligations outstanding as of the Termination Date. "Financial Obligations" are all long-term (in excess of six (6) months) financial obligations of SPP, including but not limited to the following:
 - debts under all mortgages, loans, loan agreements, borrowings, promissory notes, bonds, and credit lines under which SPP is obligated, including principal and interest;
 - b. all payment obligations under equipment leases, financing leases, capital leases, real estate and office space leases, consulting contracts, and contracts for outsourced services;
 - c. any unfunded liabilities of any SPP employee pension funds, whether or not liquidated or demanded; and
 - d. the general and administrative overhead of SPP for a period of three (3) months.
- iv. Any costs, expenses or liabilities incurred by SPP directly due to the Termination, regardless of when incurred or payable, and including without limitation prepayment premiums or penalties arising under SPP Financial Obligations.
- v. Member's share (computed in accordance with the Bylaws) of all interest that will become due for payment with respect to all interest bearing Financial Obligations after the Termination Date and until the maturity of all Financial Oblations in accordance with their respective terms ("Future Interest"). In the event that a Financial Obligation carries a variable interest rate, the interest rate in effect at the Termination Date shall be used to calculate the applicable Future Interest. In determining the Member's share of

Future Interest, SPP shall take into account any reduction of Financial Obligations due to mitigation under this Section.

- (c) In the event of a Partial Termination, Existing Obligations shall first be calculated as though a Termination occurred, and the Member shall pay a percentage thereof as Existing Obligations due to the Partial Termination. Such percentage shall be the percentage reduction of the Net Energy for Load Ratio applicable to the Member resulting from the Partial Termination.
- In the event of a Termination or Partial Termination by a Member, the Member shall pay to SPP all costs SPP incurs to remove the Member's facilities and/or load from SPP markets and operations. Such costs will be determined by SPP and shall include but not be limited to costs associated with modifying systems and databases, staff time, legal costs, and all costs of completing other tasks necessary to process the Member's Termination. SPP will apply the Member's withdrawal deposit, as specified in Section 4.2.1(b), to such costs, and any costs exceeding the withdrawal deposit shall be included in the invoice to the Member as discussed in Section 4.3.2(e) of this Agreement.
- (e) SPP shall invoice Member for Existing Obligations within one month after the Termination Date, except that delay by SPP in issuing the invoice shall not diminish Member's obligation to make timely payment. The invoice shall be due and payable no later than five (5) business days after issuance. Any amounts owed by SPP to the Member shall, solely at SPP's election and in its discretion, be offset against the Member's Existing Obligations or paid to the Member concurrently with issuance of the invoice.
- (f) The Member acknowledges and agrees that Existing Obligations include amounts that SPP expects to accrue and that will become payable by SPP between the date of Member's Notice of Termination and the Member's Termination Date, and that no part of a payment of Existing Obligations shall be refundable to the Member under any circumstances, including

(except as provided in this Section with respect to mitigation or the execution of a new Membership—Agreement by the Member after the Member's Termination) any reduction of the Financial Obligations. Any disagreement as to the calculation of Existing Obligations shall be resolved in accordance with the dispute resolution procedures in the Bylaws. If, after Termination, the Member elects to re-join SPP and execute the Membership—Agreement then in effect, SPP, in its sole discretion, may elect to credit a portion or all of the Member's Existing Obligations paid to SPP upon the Member's earlier Termination against any future payments owed by the Member to SPP.

4.3.3 Construction of Transmission Facilities

Any obligations relating to the construction of new facilities pursuant to an approved plan of SPP shall be negotiated between SPP and the Transmission Owner prior to the Termination Date so as to continue the Transmission Owner's construction obligation for facilities for which SPP has issued a notification to construct to the Transmission Owner prior to the Termination Date. If such obligations cannot be resolved through negotiations, they shall be resolved in accordance with the dispute resolution procedures in the Bylaws and Membershp Agreement.

4.3.3A Financial Obligations for Transmission Facilities

This Section 4.3.3A applies to any Terminated Member that was a Transmission Owner at the time it submitted its notice of intent to withdraw. Such Terminated Member shall remain financially responsible for all financial obligations incurred and costs allocated to its load for transmission facilities approved prior to the Termination Date. Payments in fulfillment of any such obligations and allocated costs shall commence on the date that the costs of such transmission facilities are reflected in SPP's generally applicable rates, unless SPP and the Terminated Member agree to an alternate date. Rights, obligations, and payments applicable to time periods prior to the Termination Date shall be honored by SPP and the Terminated Member. Fulfillment and performance of such rights and obligations, and rights and obligations regarding the use of such

transmission facilities, shall be negotiated between SPP and the Terminated Member, and any disputes involving such rights and obligations shall be resolved in accordance with the dispute resolution procedures in the Bylaws and Membership-Agreement.

4.3.4 Regulatory and Other Approvals or Procedures

Any Termination with respect to a Transmission Owner shall be subject to applicable **F**ederal and state law and regulatory approvals or procedures.

5.1 Regulatory and Other Authorities

This Agreement and the participation of Member is subject to acceptance or approval by FERC, and may be subject to actions of respective Federal or state regulatory authorities to which Member may be subject, and to the actions of any other governmental body which may affect the ability of Member to participate in this Agreement. The following items describe Member's rights and obligations in the event regulatory and other approvals or acceptances are not obtained or changes are required:

- (a) In the event FERC disapproves or refuses to accept this Agreement or the changes to the OATT, transmission service agreements, and Bylaws developed together with this Agreement, then this Agreement shall cease to be effective except that the signatories shall be obligated to attempt expeditiously and in good faith to negotiate a substitute agreement and OATT, transmission service agreements, and Bylaws which address the reasons for such FERC action. If, despite such good faith negotiation, the signatories are unable to produce such a substitute agreement and OATT, transmission service agreements, and Bylaws, then the signatories shall have no further obligations under this Agreement or any filing associated herewith.
- (b) In the event of any order or decision by FERC or by a court modifying this Agreement or the OATT, transmission service agreements, and Bylaws submitted as part of the initial filing seeking FERC acceptance or approval, that in the judgment of Member adversely affects it, then Member, at its sole discretion, may withdraw from this Agreement by providing written notice to the President of SPP no later than thirty days after such order or decision without receiving any FERC authorization. In such event, Member will in good faith negotiate to determine whether changes should be made to the Agreement or the OATT, transmission service agreements, and Bylaws to address the reasons for Member's withdrawal.

5.2 Tax Authorities

If the Internal Revenue Service or any other federal, state, or local taxing authority issues, or fails to issue, any ruling, or imposes any requirement or obligation, in connection with this Agreement on Member adverse to Member (in its sole judgment), or if adherence to this Agreement jeopardizes the tax-exempt status of Member or its bonds, then Member may, within 30 days of the date of such final order, or a good faith belief of such adverse consequences, withdraw from this Agreement subject to receiving any necessary regulatory approvals. In such event, Member and SPP will, in good faith, negotiate to determine whether changes should be made to the Agreement to address the reasons for Member's withdrawal.

Nothing in this Agreement, nor Member's obligations and performance thereunder, shall affect, or require Member to take or refrain from taking any action that would affect the rights and obligations or enforceability of Member's present or future bond resolutions, tax-exempt debt covenants and financing agreements. Member shall determine in its sole discretion and judgment, in accordance with advice and opinions from its legal counsel, what actions, conduct and performance it is permitted to or must take under its bond resolutions, tax-exempt debt covenants and financing agreements. Member and SPP will meet and confer regarding the matter and, as necessary, negotiate in good faith to modify the Agreement to address the matter.

5.3 Effectiveness as to Certain Members

The effectiveness of this Agreement as to Member where it is a governmental entity or non-public utility and has outstanding tax-exempt bonds issued to finance, in whole or in part, generation, transmission, or Distribution Facilities is dependent upon satisfaction or Member's written waiver of the following conditions precedent:

- (a) Receipt of an unqualified opinion of a nationally recognized bond counsel to the effect that the provisions of this Agreement do not adversely affect the exclusion from gross income of interest on any such outstanding bonds issued to finance generation, transmission, and Distribution Facilities under the Internal Revenue code of 1986, as amended;
- (b) Receipt of an unqualified opinion of a nationally recognized bond counsel or general counsel to Member to the effect that the provisions of this Agreement do not constitute a breach or impairment of, or a default under, any agreement to which it is a party, including, but not limited to, its master bond resolution, as amended, and any power sales contracts with its municipal transmission users (if any), as amended, or other agreements;
- c) Receipt of a certificate of the trustee for any such outstanding bonds issued for generation, transmission and Distribution Facilities to the effect that Member's entry into this Agreement is permitted under the master bond resolution, as amended; and
- (d) Receipt of an opinion of nationally recognized bond counsel or general counsel to Member that it has full constitutional and statutory authority to enter into this Agreement.

In the event that any of the foregoing conditions are not satisfied or waived by Member, then it shall promptly give notice of its objections or conditions which have not been satisfied to SPP, and SPP shall expeditiously attempt in good faith to negotiate a substitute agreement.

8.1 Governing Law

This Agreement shall be interpreted, construed, and governed by the laws of the State of Arkansas, except to the extent preempted by the law and/or unless a court with jurisdiction rules otherwise, and in a case involving a Federal Power Marketing Agency, Federal law shall apply, provided, however, that (i) all matters relating to real property or any interest in realty shall be governed by the laws of the State wherein such real property or interest in realty is physically located, and (ii) any court or regulatory body applying Arkansas law shall give full effect to Section 3.12 of this Agreement regarding Member's obligations under state law.

8.2 Successors and Assigns

This Agreement shall inure to the benefit of, and be binding upon SPP and Member, their respective successors and assigns permitted hereunder, but shall not be assignable by SPP without prior written agreement from Member, with such written agreement not to be unreasonably withheld, or by Member, by operation of law or otherwise, without the approval of the Board of Directors which approval shall not be unreasonably withheld, except that no Board of Directors approval is required as to a successor in the operation of Transmission Owner's Tariff Facilities committed to administration by SPP by reason of a merger, consolidation, reorganization, sale, spin-off, or foreclosure, as a result of which substantially all such transmission facilities are acquired by such successor, and such successor becomes a Transmission Owner under this Agreement.

8.4 Severability

Except as may be stated otherwise in any Amendments to this Agreement, Eeach provision of this Agreement shall be considered severable, and if for any reason any provision of this Agreement, or the application thereof to any person, entity, or circumstance, is determined by a court or regulatory authority of competent jurisdiction to be invalid, void, or unenforceable, then the remaining provisions of this Agreement shall continue in full force and effect and shall in no way be affected, impaired, or invalidated, and such invalid, void, or unenforceable provision shall be replaced with a suitable and equitable provision in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid, void, or unenforceable provision. This Section 8.4 does not modify or change in any way the right of Member to withdraw as provided elsewhere in this Agreement.

8.6 Representations and Warranties

Member and SPP each represent and warrant to the other that as of the later of the date it executes this Agreement or the Effective Date of this Agreement:

- (a) It is duly organized, validly existing, and in good standing under the laws of the jurisdiction where organized;
- (b) Subject to any necessary approvals by **F**ederal or state regulatory authorities of SPP, the execution and delivery by Member and SPP of this Agreement, and the performance of its respective obligations hereunder have been duly and validly authorized by all requisite action on the part of the signatories and does not conflict with any applicable law or with any other agreement binding upon the signatories, other than third party joint agreements covered in this Agreement. This Agreement has been duly executed and delivered by Member and SPP, and, subject to the conditions set forth in this Agreement, constitutes the legal, valid, and binding obligation on the part of Member and SPP, enforceable against it in accordance with its terms except insofar as the enforceability thereof may limited by applicable bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium, or other similar laws affecting the enforcement of creditor's rights generally, and by general principles of equity regardless of whether such principles are considered in a proceeding at law or in equity; and
- (c) There are no actions at law, suits in equity, proceedings, or claims pending or, to the knowledge of Member or SPP, threatened against Member or SPP before or by any federal, state, foreign or local court, tribunal, or governmental agency or authority that might materially delay, prevent, or hinder the performance by such entity of its obligations hereunder.

8.10 Good Faith Efforts

Member and SPP agree that each shall in good faith take all reasonable actions necessary to fulfill its respective obligations under this Agreement. Where the consent, agreement, or approval of Member or SPP must be obtained hereunder, such consent, agreement, or approval shall not be unreasonably withheld, conditioned, or delayed. Where Member or SPP is required or permitted to act, or omit to act, based on its opinion or judgment, such opinion or judgment shall not be unreasonably exercised. To the extent that the jurisdiction of any #Federal or state regulatory authority applies to any part of this Agreement and/or the transactions or actions covered by this Agreement, Member and SPP secure any necessary or desirable approval or acceptance of such regulatory authorities of such part of this Agreement and/or such transactions or actions.

FERC rendition of the electronically filed tariff records in Docket No. ER14-02851-000

Filing Data: CID: C000771

Filing Title: Integrated System Bylaws and Membership Agreement Revisions

Company Filing Identifier: 860 Type of Filing Code: 10 Associated Filing Identifier:

Tariff Title: Governing Documents Tariff

Tariff ID: 15

Payment Confirmation: Suspension Motion:

Tariff Record Data:

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Sedtion 1 Definitions, Bylaws 1.0 Definitions, 3.0.0, A

Record Narative Name: Bylaws Section 1 Definitions

Tariff Record ID: 4

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE

Record Content Type: 1
Associated Filing Identifier:

1.0 Definitions

Affiliate Relationships

Affiliate Relationships are relationships between SPP Members that have one or more of the following attributes in common:

- (a) are subsidiaries of the same company;
- (b) one Member is a subsidiary of another Member;
- (c) have, through an agency agreement, turned over control of a majority of their generation facilities to another Member;
- (d) have, through an agency agreement, turned over control of a majority of their transmission system to another Member, except to the extent that the facilities are turned over to an independent transmission company recognized by FERC;
- (e) have an exclusive marketing alliance between Members; or
- (f) ownership by one Member of ten percent or greater of another Member.

Articles of Incorporation

SPP's articles of incorporation as filed with the state of Arkansas.

Board of Directors

The Board of Directors of SPP, which shall manage the general business of SPP pursuant to these Bylaws.

Bylaws

These bylaws.

Criteria

Planning and operating standards and procedures as approved by the Board of Directors.

Existing Obligations

Certain financial obligations as defined in Section 8.7.1 of these Bylaws.

ERO

The Electric Reliability Organization under FERC jurisdiction that regulates reliability of the electric power grid.

Federal Power Marketing Agency

This term shall include the term "Federal Power Marketing Administration" and have the same definition that is set forth in the Federal Power Act at 16 U.S.C. § 796(19), which defines "Federal power marketing agency" as "any agency or instrumentality of the United States (other than the Tennessee Valley Authority) which sells electric energy[.]"

Federal Power Marketing Agency Amendments

The amendments and revisions to the SPP Bylaws, the SPP Membership Agreement, and Section 39.3 of the OATT that are required by a Federal Power Marketing Agency for membership in SPP at the time of the Federal Power Marketing Agency's initial membership or as they may be revised in the future by mutual agreement between the Federal Power Marketing Agency and SPP.

Member

An entity that has met the requirements of Section 2.2 of these Bylaws.

Membership

The collective Members of SPP.

Membership Agreement

The contract, that specifies the rights and obligations of the parties, executed between SPP and an entity seeking to become an SPP member.

NERC

The North American Electric Reliability Corporation or successor organizations.

Net Energy for Load

The load served by transmission facilities under the SPP Open Access Transmission Tariff.

Officers

The officers of SPP as elected by the Board of Directors. The Officers consist of the President and the Corporate Secretary, at a minimum. Any Officer must be independent of any Member organization.

Organizational Group

A group, other than the Board of Directors, comprising a committee or working group that is charged with specific responsibilities toward accomplishing SPP's mission.

Regional Criteria

SPP planning and operating standards and procedures as approved by the Board of Directors.

Regional Entity Trustees

A governing body of SPP, independent of the Board of Directors, which specifically

oversees SPP's function as an ERO Regional Entity pursuant to the Delegation Agreement between SPP and the ERO.

Regional Reliability Standards

Electric reliability requirements submitted to the ERO by the Regional Entity Trustees; and once approved, implemented and enforced by SPP under authority as the Regional Entity.

Registered Entity(ies)

A bulk electric system owner, operator or user that is required to comply with ERO reliability standards pursuant to the Energy Policy Act of 2005.

SPP

Southwest Power Pool, Inc.

SPP Regional Entity

That part of SPP responsible for the delegated functions pursuant to the Delegation Agreement between SPP and the ERO.

SPP Compliance Monitoring and Enforcement Program

The program used by the North American Electric Reliability Corporation ("NERC") and the Regional Entities to monitor, assess, and enforce compliance with Reliability Standards within the United States.

Staff

The technical and administrative staff of SPP as hired by the Officers to accomplish SPP's mission.

Standards Development Team

An SPP Organizational Group assigned or choosing to develop an SPP Regional Reliability Standard for submission to the ERO for approval for enforcement.

Terminated Member

An entity that was a Signatory to the Membership Agreement but whose membership in

SPP has been terminated under Section 4 of the Membership Agreement.

Transmission Owning Member

A Member that has placed more than 500 miles of non-radial facilities operated at or

above 60 kV under the independent administration of SPP for the provision of regional

transmission service as set forth in the Membership Agreement.

Transmission Using Member

A Member that does not meet the definition of a Transmission Owning Member.

Western Area Power Administration-Upper Great Plains Region ("Western-UGP")

A division of the Western Area Power Administration that markets and transmits Federal

power from reservoir projects under the control of the Department of the Army or the

U.S. Bureau of Reclamation to Statutory Load Obligations, including preference power

customers in Iowa, Minnesota, Montana, Nebraska, North Dakota, and South Dakota

located in a defined marketing area. Western-UGP operates the WAUW Balancing

Authority Area in the Western Interconnection, where certain of its transmission facilities

are located.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Section 2.1, Bylaws 2.1 Qualifications, 1.0.0, A

Record Narative Name: Bylaws Section 2.1 Qualifications

Tariff Record ID: 6

Tariff Record Collation Value: 44060340 Tariff Record Parent Identifier: 5

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1

Associated Filing Identifier:

2.1 Qualifications

Membership in SPP is voluntary and is open to any electric utility, Federal Power

Marketing Agency, transmission service provider, any entity engaged in the business of

producing, selling and/or purchasing electric energy for resale, and any entity willing to

meet the membership requirements, including execution of the Membership Agreement. Membership also is open to any entity eligible to take service under the SPP Open Access Transmission Tariff (OATT). These entities desire the greater efficiency and service reliability gained through better coordination by voluntary association in SPP as constituted herein and in the SPP Articles of Incorporation. Members recognize that such association has a significant effect upon the availability and reliability of the bulk electric power supply of the region, and thereby affects the reliability of the nation's electric power supply.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 3.15, Bylaws 3.15 Liability, Insurance and Indemnification, 2.0.0, A Record Narative Name: Bylaws Section 3.15 Liability, Insurance and Indemnification

Tariff Record ID: 27

Tariff Record Collation Value: 216301700 Tariff Record Parent Identifier: 11

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

3.15 Liability, Insurance and Indemnification

For purposes of this section "SPP" refers to SPP and its officers, directors, Regional Entity Trustees, employees or agents, and "Member" refers to the Members of SPP as defined in these Bylaws. None of the provisions of this section, including the waiver of liability in Section 3.15.1 below, absolving SPP or its Members, directors, Regional Entity Trustees, officer, agents, employees or other representatives of liability or any provisions for insurance or indemnification apply to actions which are unlawful, undertaken in bad faith, or are the result of gross negligence or willful misconduct.

3.15.1 Waiver of Liability

- (a) SPP shall not be liable to any Member for damages arising out of or related to any directive, order, procedure, action, or requirement of SPP, under the then effective Bylaws and Criteria.
- (b) No Member shall be liable to any other Member or to SPP for damages arising out of or related to any action by the Member pursuant to any directive, order, procedure, action or requirement of SPP, under the then

- effective Bylaws and Criteria.
- (c) Each Member waives any future claim it might have against SPP or other Members arising out of or resulting from any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria.
- (d) SPP waives any future claim it might have against any Member arising out of or resulting from any actions taken by a Member pursuant to any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria.

3.15.2 Insurance

The President is authorized to procure insurance to protect SPP, its directors, Regional Entity Trustees, officers, agents, employees, or other representatives against damages arising out of or related to any directive, order, procedure, action or requirement of SPP, under the then effective Bylaws and Criteria or pursuant to the OATT.

3.15.3 Indemnification of Directors, Officers, Agents and Employees

Except for actions which are unlawful, undertaken in bad faith, or are the result of gross negligence or willful misconduct, SPP shall indemnify its directors, officers, agents, employees, or other representatives to the maximum extent allowed by law consistent with these Bylaws. Each director, Regional Entity Trustee, officer, agent, employee, or other representative of SPP shall be indemnified by SPP against all judgments, penalties, fines, settlements, and reasonable expenses, including legal fees, incurred as a result of, or in connection with, any threatened, pending or completed civil, criminal, administrative, or investigative proceedings to which the incumbent may be made a party by reason of acting or having acted in official capacity as a director, Regional Entity Trustee, officer, agent, employee, or representative of SPP, or in any other capacity which the incumbent may hold at the request of SPP, as its representative in any other organization, subject to the following conditions:

- (a) Such director, Regional Entity Trustee, officer, agent, employee, or other representative must have acted in good faith and, in the case of criminal proceedings, must have had no reasonable cause to believe that conduct was unlawful; provided, that SPP shall not provide indemnification of any conduct judged unlawful in criminal proceedings. When acting in official capacity, the incumbent must have reasonably believed that conduct was in the best interests of SPP, and, when acting in any other capacity, must have reasonably believed that conduct was at least not opposed to the best interests of SPP.
- (b) If the proceeding was brought by or on behalf of SPP, however, indemnification shall be made only with respect to reasonable expenses referenced above. No indemnification of any kind shall be made in any such proceeding in which the director, Regional Entity Trustee, officer, agent, employee, or other representative shall have been adjudged liable to SPP.
- (c) In no event, however, will indemnification be made with respect to

- any described proceeding which charges or alleges improper personal benefit to a director, Regional Entity Trustee, officer, agent, employee, or other representative and where liability is imposed on the basis of the receipt of such improper personal benefit.
- (d) In order for any director, Regional Entity Trustee, agent, employee, or other representative to receive indemnification under this provision, the person shall vigorously assert and pursue any and all defenses to those claims, charges, or proceedings covered herein which are reasonable and legally available and shall fully cooperate with SPP or any attorneys involved in the defense of any such claim, charges, or proceedings on behalf of SPP.
- (e) No indemnification shall be made in any specific instance until it has been determined by SPP that indemnification is permissible in that specific case, under the standards set forth herein and that any expenses claimed or to be incurred are reasonable. These two (2) determinations shall be made by a majority vote of at least a quorum of the Board of Directors consisting solely of directors who were not parties to the proceeding for which indemnification or reimbursement of expenses is claimed. If such a quorum cannot be obtained, a majority of at least a quorum of the full Board of Directors, including directors who are parties to said proceeding, shall designate a special legal counsel who shall make said determinations on behalf of SPP.
- (f) Any reasonable expenses, as shall be determined above, that have been incurred by a director, Regional Entity Trustee, officer, agent, employee, or other representative who has been made a party to a proceeding as defined herein, may be paid or reimbursed in advance upon a majority vote of a quorum of the full Board of Directors, including those who may be a party to the same proceeding. However, such director, Regional Entity Trustee,

officer, agent, employee, or other representative shall have provided SPP with (i) a written affirmation under oath that the incumbent, in good faith, believes the conditions of indemnification herein have been met; and (ii) a written undertaking that the incumbent shall repay any amounts advanced, with interest accumulated at a reasonable rate, if it is ultimately determined that such conditions are not met.

3.15.4 Limitations

The provisions of this section 3.15 are subject to applicable state and Federal laws, if any, which limit the ability of a Member to waive liability or enter into agreements of indemnity. Any benefits under this Section 3.15 shall not extend to any Member so limited by state or Federal law in complying with the provisions thereof.

3.15.5 Modification of Rights by Agreement

Any provision of this Section 3.15.1 may be waived or modified by express written agreement between SPP and Member. Such express written agreement shall apply solely to the subject matter of the agreement and is not intended to be a general waiver or modification of the rights provided in Section 3.15.1.

3.15.6 Procedural Rights Not Affected

The limitations of liability provided in Section 3.15.1 shall not affect any procedural rights or obligation a Member may have at law or equity.

 $Record\ Content\ Description,\ Tariff\ Record\ Title,\ Record\ Version\ Number,\ Option\ Code:$

Section 5.1, Bylaws 5.1 Members Committee, 1.0.0, A

Record Narative Name: Bylaws Section 5.1 Members Committee

Tariff Record ID: 38

Tariff Record Collation Value: 311034448 Tariff Record Parent Identifier: 37

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

5.1 Members Committee

The Members Committee shall work with the Board of Directors to manage and direct the general business of SPP. Its duties shall include, but are not limited to the following:

- (a) Provide individual and collective input to the Board of Directors, including but not limited to a straw vote from the Members Committee representatives as an indication of the level of consensus among Members, on all actions pending before the Board of Directors;
- (b) Serve on committees reporting to the Board of Directors as appointed by the Board of Directors; and
- (c) Provide input with the Board of Directors to the Regional Entity Trustees on SPP Regional Reliability Standards presented by the MOPC to the Trustees or otherwise developed under the auspices of the Trustees for submission to the ERO for its approval.

5.1.1 Composition and Qualifications

5.1.1.1 Composition

Provided that Membership is sufficient to accommodate these provisions, the Members Committee shall consist of up to 24 persons. Six representatives shall be investor owned utilities Members; five representatives shall be cooperatives Members; two representatives shall be municipals Members (including municipal joint action agencies); three representatives shall be independent power producers/marketers Members; two representatives shall be state power agencies Members; one representative shall be from a Federal Power Marketing Agency; two representatives shall be alternative power/public interest Members; one representative shall be from an independent transmission company Member, defined as having assets under the OATT and no Affiliate Relationships in other categories of Membership; one representative shall be a large retail customer Member, defined as non-residential end-use customers with individual or aggregated loads of 1-MW or more; and one representative shall be a small retail customer Member, defined as residential customers and other customers with individual or aggregated loads of less than 1-MW. Representatives will be

elected in accordance with Section 5.1.2 of these Bylaws.

5.1.1.2 Qualifications

A representative shall be an officer or employee of a Member with decision-making responsibility over SPP related activities, and must be the Member's representative to the Membership.

5.1.2 Term and Election

Representatives shall be nominated by the Corporate Governance Committee and elected each year at the meeting of Members to staggered three-year terms commencing upon election and continuing until their duly elected successors take office. The election process shall be as follows:

- (a) At least 90 calendar days prior to the meeting of Members at which election of new representatives is required, the Corporate Governance Committee shall nominate persons equal in number to the representatives to be elected;
- (b) At least 30 calendar days prior to the meeting of Members, the Corporate Governance Committee shall determine the persons it nominates for election as representatives, specifying the nominee for any vacancy to be filled. The Corporate Secretary shall prepare the ballot accordingly, leaving space for additional names, and shall deliver same to Members at least two weeks prior to the meeting of Members;
- (c) For purposes of electing and removing representatives only, each group of Members with Affiliate Relationships shall be considered a single vote;
- (d) At the meeting of Members, any additional nominee or nominees may be added to the ballot if a motion is made and seconded to add such nominee or nominees; and
- (e) The required number of representatives shall be elected by written ballot.

 A Member shall be entitled to cast a number of votes equal to the number of representatives to be elected. A Member may not cumulate votes.

 The candidates in each sector receiving the greatest number of votes will fill vacancies.

5.1.3 Resignation and Removal of Members Committee Representatives

Any representative may resign by written notice to the President noting the effective date of the resignation. A representative may be removed, with cause, by the affirmative vote of a majority of the Members at a meeting of Members. Removal proceedings may only be initiated by a petition signed by not less than twenty percent of the Members. The petition shall state the specific grounds for removal and shall specify whether the removal vote is to be taken at a special meeting of Members or at the next regular meeting of Members. A representative who is the subject of removal proceedings shall be given fifteen days to respond to the Member petition in writing to the President.

5.1.4 Vacancies

If a vacancy occurs the Corporate Governance Committee may elect an interim representative from the same sector to serve until a replacement representative from the same sector is elected and takes office. A special election shall be held at the next meeting of Members to fill the vacancy for the unexpired term. The replacement representative shall take office immediately following the election.

5.1.5 Meetings

The Members Committee shall meet only with the Board of Directors.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 6.6, Bylaws 6.6 Corporate Governance Committee, 3.0.0, A
Record Narative Name: Bylaws Section 6.6 Corporate Governance Committee
Tariff Record ID: 45
Tariff Record Reco

Tariff Record Collation Value: 371318924 Tariff Record Parent Identifier: 39

Proposed Date: 2014-11-10

Priority Order: 500
Record Change Type: CHANGE
Record Content Type: 1
Associated Filing Identifier:

6.6 Corporate Governance Committee

The Corporate Governance Committee is responsible for the overall governance structure, including nominations, for the company in accordance with its scope as approved by the Board of Directors.

To the extent that the membership allows, the CGC shall be comprised of ten members. One representative shall be the President of SPP who will serve as the Chair; one representative shall be the Chairman of the Board, unless his/her position is under consideration, in which case

the Vice Chairman of the Board; one representative shall be representative of and selected by investor owned utilities Members; one representative shall be representative of and selected by co-operatives Members; one representative shall be representative of and selected by municipals Members; one representative shall be representative of and selected by independent power producers/marketers Members; one representative shall be representative of and selected by state power agencies Members; one representative shall be representative of and selected by alternative power/public interest Members; one representative shall be representative of and selected by large/small retail Members; and one representative shall be representative of and selected by a Federal Power Marketing Agency Member(s).

Where a vacancy occurs with respect to a representative of a sector, the representatives from the appropriate sector will fill the vacancy. For purposes of selecting or removing representatives only, each group of Members with Affiliate Relationships shall be considered a single Member.

The CGC shall meet at least once per calendar year, and additionally as needed, provided that a quorum, as defined in these Bylaws, is present. The CGC shall report to the Board of Directors following each CGC meeting with respect to its activities and with such recommendations, as the CGC deems necessary.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 8.7, Bylaws 8.7 Financial Obligation of Withdrawing Members, 3.0.0, A Record Narative Name: Bylaws Section 8.7 Financial Obligation of Withdrawing Members Tariff Record ID: 57

Tariff Record ID: 57

Tariff Record Collation Value: 474663740 Tariff Record Parent Identifier: 50

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

8.7 Financial Obligation of Withdrawing Members

8.7.1 Existing Obligations

"Existing Obligations" are the following:

- a. Member's unpaid annual membership fee.
- b. Member's unpaid dues, assessments, and other amounts charged under Section 3.8 of the Membership Agreement, section 8.4 of the

Bylaws, or otherwise under the Bylaws, plus the Member's share of costs SPP customarily includes in such dues, assessments or other charges, but which as of the Termination Date SPP had not included in such dues, assessments or other charges.

- c. Member's share (computed in accordance with the Bylaws) of the entire principal amounts of all SPP Financial Obligations outstanding as of the Termination Date. "Financial Obligations" are all long-term (in excess of six (6) months) financial obligations of SPP, including but not limited to the following:
 - debts under all mortgages, loans, loan agreements, borrowings, promissory notes, bonds, and credit lines, under which SPP is obligated, including principal and interest:
 - ii. all payment obligations under equipment leases, financing leases, capital leases, real estate and office space leases, consulting contracts, and contracts for outsourced services;
 - iii. any unfunded liabilities of any SPP employee pension funds, whether or not liquidated or demanded; and
 - iv. the general and administrative overhead of SPP for a period of three (3) months.
- d. Any costs, expenses or liabilities incurred by SPP directly due to the Termination, regardless of when incurred or payable, and including without limitation prepayment premiums or penalties arising under SPP Financial Obligations.
- e. Member's share (computed in accordance with the Bylaws) of all interest that will become due for payment with respect to all interest bearing Financial Obligations after the Termination Date and until the maturity of all Financial Obligations in accordance with their respective terms ("Future Interest"). In the event that a Financial Obligation carries a variable interest rate, the interest rate in effect at the Termination Date shall be used to calculate the

applicable Future Interest. In determining the Member's share of Future Interest, SPP shall take into account any reduction of Financial Obligations due to mitigation under this Section.

8.7.2 Computation of a Member's Existing Obligations

For purposes of computing the Existing Obligations of any withdrawing or terminated Member in accordance with the Membership Agreement, such "Member's share" is a percentage calculated as follows:

A = 100 [0.25(1/N) + 0.75(B/C)]

Where: A = Member's share (expressed as a percentage)

N = Total number of Members

B = The Member's previous year Net Energy for Load

C = Total of factor B for all Members

The Finance Committee shall have the discretion to reduce the Existing Obligations of any withdrawing or Terminated Member, to reflect any SPP costs or expenses that may be mitigated in connection with such Member's withdrawal or termination. In the event of consolidation of affiliate memberships or the transfer of membership from one corporate entity to another, whereby one entity remains a member of SPP, the withdrawal obligation for the departing company(ies) may be waived at SPP's sole discretion.

8.7.3 Financial Obligations for Transmission Facilities

To the extent that Section 4.3.3A of the Membership Agreement is applicable, a Terminated Member shall remain financially responsible for all financial obligations incurred and costs allocated to its load for transmission facilities approved prior to the Termination Date.

8.7.4 Penalty Costs

A Terminated Member shall remain liable for its share of costs associated with penalties assessed against SPP by FERC, the FERC-approved Electric Reliability Organization, any Electric Reliability Organization-approved Regional Entity, or any other governmental or regulatory authority with jurisdiction over SPP that SPP incurs as a result of events that occurred prior to Member's Termination Date but that SPP is unable to recover under the SPP OATT.

8.7.5 Limitation on Financial and Penalty Obligations

- (a) Notwithstanding the delineation of Members' financial obligations in Section 8.7, a Federal Power Marketing Agency shall not be subject to the financial obligations listed in this Section 8.7 in the event FERC finds that SPP has not adhered to all of the Federal Power Marketing Agency Amendments as that term is defined in Section 1.0 of these Bylaws or if SPP files and FERC approves material changes to the Federal Power Marketing Agency Amendments.
- (b) Provided further, notwithstanding any language to the contrary in these Bylaws, a Federal Power Marketing Agency has not waived or conceded any defense it may have, including sovereign immunity, intergovernmental immunity, or lack of subject matter jurisdiction in any action against it by an Enforcement Authority, nor has it accepted any liability, responsibility, or obligation to pay any civil monetary penalties or fines imposed by an Enforcement Authority to which it would not have been subject in the absence of these Bylaws. SPP, in accepting Western-UGP as a member, does not thereby concede or accept responsibility for any portion of a penalty or fine attributable to the actions or omissions of Western-UGP. SPP will identify the amount of any penalty or fine that SPP allocates to Western-UGP or that SPP determines is attributable to Western-UGP and will identify that amount to FERC as uncollectable and not otherwise owed by SPP. Enforcement Authority in these Bylaws means the Federal Energy Regulatory Commission (FERC), Electric Reliability Organization (ERO), or Regional Entities with enforcement authority pursuant to a delegation from an ERO or FERC for the purpose of proposing and enforcing reliability standards.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Section 10.0, Bylaws 10.0 Amendments To These Bylaws, The Articles..., 1.0.0, A

Record Narative Name: Bylaws Section 10.0 Amendments to These Bylaws, The Articles of Incorporation and Membership Agreement

Tariff Record ID: 66

Tariff Record Collation Value: 552172352 Tariff Record Parent Identifier: 1

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE

Record Content Type: 1
Associated Filing Identifier:

10.0 Amendments To These Bylaws, The Articles Of Incorporation, And Membership

Agreement

Except for modifications to Section 4.0 BOARD OF DIRECTORS, Section 5.0

COMMITTEES ADVISING THE BOARD OF DIRECTORS, Section 8.7.5 LIMITATION ON

FINANCIAL AND PENALTY OBLIGATIONS, Section 9.0 REGIONAL ENTITY

FUNCTION, and Section 10.0 AMENDMENTS, these Bylaws may be amended, repealed, or

added to by the Board of Directors only upon 30 days written notice to the Membership of the

proposed modification(s). Approval of amendments to the Bylaws by the Board of Directors

must be by an affirmative vote of at least five directors. Sections 4.0, 5.0, 8.7.5, 9.0, and 10.0 of

these Bylaws and the Articles of Incorporation may be amended, repealed, or added to only by

approval of the Membership. Provided, that all changes to Federal Power Marketing

agency/administration representation in Section 5.1.1 and any change to Section 8.7.5 must be

mutually agreed to by the Federal Power Marketing Agency Member and SPP. All amendments

are subject to the requisite regulatory approval(s).

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

MA Table of Contents, MA Table of Contents, 2.0.0, A

Record Narative Name: Membership Agreement Table of Contents

Tariff Record ID: 72

Tariff Record Collation Value: 578008556 Tariff Record Parent Identifier: 69

Proposed Date: 2014-11-10

Priority Order: 600

Record Change Type: CHANGE Record Content Type: 1

Associated Filing Identifier:

MEMBERSHIP AGREEMENT

TABLE OF CONTENTS

1.0 Definitions

2.0 Rights, Powers and Obligations of SPP

2.1 Operation and Planning

2.1.1 General

2.1.2 Reliability

- 2.1.3 Transmission Maintenance
- 2.1.4 Generation Maintenance
- 2.1.5 Planning Activities
- 2.2 Non-Discriminatory Transmission Service
 - 2.2.1 Pricing
 - 2.2.2 Standards of Conduct
 - 2.2.3 OASIS
 - 2.2.4 Ancillary Services
 - 2.2.5 Transmission Service Scheduling
- 2.3 Fiduciary Responsibilities and Duties of SPP to Members
- 2.4 Additional Obligations of SPP
 - 2.4.1 Inspection and Auditing Procedures
 - 2.4.2 Stranded Cost Recovery Charges
- 3.0 Commitments, Rights, Powers, and Obligations of Member
 - 3.1 Redispatch and Curtailment
 - 3.2 Transmission and Generation Maintenance Practices
 - 3.3 Construction
 - 3.4 Use of Distribution Facilities
 - 3.5 Providing Information
 - 3.6 Facilities Access
 - 3.7 Inspection and Auditing Procedures
 - 3.8 Compliance and Bylaws and Other Policies and Procedures
 - 3.9 Planning and Participation
 - 3.10 Pricing
 - 3.11 No Waiver of Jurisdictional Immunity
 - 3.12 Compliance with Federal or State Law
- 4.0 Termination of Membership
 - 4.1 Events of Termination and Partial Termination
 - 4.2 Termination Procedures and Effective Dates
 - 4.2.1 Notice of Voluntary Withdrawal
 - 4.2.2 Effective Date of Termination
 - 4.3 Obligations Upon Termination
 - 4.3.1 Obligation to Hold Users Harmless
 - 4.3.2 Obligation to Pay Current and Existing Obligations
 - 4.3.3 Construction of Transmission Facilities
 - 4.3.3A Financial Obligations for Transmission Facilities
 - 4.3.4 Regulatory and Other Approvals or Procedure
- 5.0 Regulatory, Tax, And Other Authorities
 - 5.1 Regulatory and Other Authorities
 - 5.2 Tax Authorities
 - 5.3 Effectiveness as to Certain Members
- 6.0 Removal of Members

7.0 Effective Date, Duration, and Transition

8.0 Miscellaneous Provisions

- 8.1 Governing Law
- 8.2 Successors and Assigns
- 8.3 No Implied Waivers
- 8.4 Severability
- 8.5 Renegotiation
- 8.6 Representations and Warranties
- 8.7 Further Assurances
- 8.8 Delivery of Notices
- 8.9 Entire Agreement
- 8.10 Good Faith Efforts
- 8.11 Third Party Joint Agreements
- 8.12 Amendment
- 8.13 Counterparts

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Section 1.0, MA 1.0 Definitions, 3.0.0, A

Record Narative Name: Membership Agreement Section 1.0 Definitions

Tariff Record ID: 73

Tariff Record Collation Value: 586620624 Tariff Record Parent Identifier: 69

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

This Agreement is made between the Member and SPP, as defined herein.

1.0 Definitions

Agreement

This Membership Agreement.

Basin Electric Amendments

The amendments and revisions to the SPP OATT, to Basin Electric Power Cooperative's ("Basin Electric's") Network Integration Transmission Service Agreement, or to Basin Electric's Network Operating Agreement necessary for Basin Electric's initial SPP membership or as they may be revised in the future by agreement between Basin Electric and SPP.

Board of Directors

The Board of Directors elected pursuant to the Bylaws.

Bylaws

SPP's Bylaws or any successor document.

Distribution Facilities

Facilities that are the subject of a separate distribution charge pursuant to the Open Access Transmission Tariff.

Eastern Interconnection

One of the three major alternating-current electrical grids in North America. The Eastern Interconnection reaches from Central Canada eastward to the Atlantic coast (excluding Quebec), south to Florida, and back west to the foot of the Rockies (excluding most of Texas).

Effective Date

This Agreement is effective on January 1, 2000 or upon the date of execution by Member if after January 1, 2000.

Electric Transmission System

The transmission facilities subject to SPP's tariff administration, except for any Distribution Facilities.

Existing Obligations

Shall have the meaning given in Section 4.3.2(b).

Federal Power Marketing Agency

This term shall include the term "Federal Power Marketing Administration" and have the same definition that is set forth in the Federal Power Act at 16 U.S.C. § 796(19), which defines "Federal power marketing agency" as "any agency or instrumentality of the United States (other than the Tennessee Valley Authority) which sells electric energy[.]"

Federal Power Marketing Agency Amendments

The amendments and revisions to the SPP Bylaws, the SPP Membership Agreement, and Section 39.3 of the OATT that are required by a Federal Power Marketing Agency for initial SPP membership or as they may be revised in the future by mutual agreement between a Federal Power Marketing Agency and SPP.

Federal Power-Western-UGP

All capacity and energy generated at reservoir projects under the control of the Department of the Army or the Bureau of Reclamation in the marketing area of the Western-UGP for the purpose of fulfilling Western-UGP's Statutory Load Obligations for the sale of capacity and energy. This shall also include any capacity and energy delivered to or from Western-UGP under the pre-OATT bi-directional agreement with Southwestern Power Administration through Associated Electric Cooperative, Inc. for delivery and receipt at the Maryville Substation. Western-UGP's deliveries to Southwestern shall be considered part of Western's Statutory Load Obligations, and receipts from Southwestern to Western-UGP will be considered as coming from Federal resources. Federal Power-Western-UGP resources shall be eligible to be considered as Designated Resources.

FERC

The Federal Energy Regulatory Commission or successor organization.

Financial Obligations

Shall have the meaning given in Section 4.3.2(b).

Future Interest

Shall have the meaning given in Section 4.3.2(b).

Good Utility Practice

Any of the practices, methods, and acts engaged in or approved by a significant portion of the electric utility industry during the relevant time period, or any of the practices, methods, and acts which, in the exercise of reasonable judgment in light of the facts known at the time the decision was made, could have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety, and expedition. Good Utility Practice is not intended to be limited to the optimum practice, method, or act, to the exclusion of all others, but rather to be a range of acceptable practices, methods, or acts generally accepted in the region. SPP Criteria and NERC Policies and Standards are considered Good Utility Practice.

Heartland Amendments

The amendments and revisions to the SPP OATT, to Heartland Consumers Power District's ("Heartland's") Network Integration Transmission Service Agreement, or to Heartland's Network Operating Agreement necessary for Heartland's initial SPP membership or as they may be revised in the future by agreement between Heartland and SPP.

Member

Signatory to this Agreement that has completed the application requirements pursuant to the Bylaws.

NERC

North American Electric Reliability Corporation or successor organizations.

Non-Transmission Owner

A Member that is not a Transmission Owner. A Non-Transmission Owner that owns or controls Tariff Facilities may have its status changed to a Transmission Owner under this Agreement upon notice to SPP and execution of this Agreement as a Transmission Owner.

Open Access Transmission Tariff (OATT)

The SPP nondiscriminatory, Open-Access Transmission Tariff (OATT) on file with FERC pursuant to Section 205 of the Federal Power Act under which SPP will offer transmission service, or any such successor tariff.

Partial Termination

Shall have the meaning given in Section 4.1.

Regional Entity

An entity having enforcement authority delegated to it by NERC pursuant to a delegation

agreement accepted by FERC.

Reliability Coordinator

SPP, in performing its reliability coordinator function as recognized by NERC pursuant to its policies, and pursuant to SPP Criteria and this Agreement.

SPP

Southwest Power Pool, Inc., or successor organization.

SPP Criteria

SPP's approved operating and planning criteria.

SPP Region

The geographic area encompassing the transmission systems of Members that are Transmission Owners.

Standards of Conduct

SPP's Standards of Conduct that apply to the conduct of its directors, officers, employees, Regional Entity trustees, contractors, and agents.

Statutory Load Obligations

Western-UGP's power marketing function obligations under Federal law to deliver capacity and energy from the output of the Federal hydroelectric projects operated by the Department of the Army and the Bureau of Reclamation to loads which include project use loads, preference power customer loads defined pursuant to a power marketing plan, and other loads required to be served under Federal law.

Tariff Facilities

The Electric Transmission System and the Distribution Facilities subject to SPP's tariff administration.

Termination

Shall have the meaning given in Section 4.1.

Termination Date

Shall mean the date of Termination is effective in accordance with Section 4.2.2(b).

Transmission Customer

A customer under the Open Access Transmission Tariff

Terminated Member

An entity that was a Signatory to this Agreement but whose membership in SPP has been terminated under Section 4 of this Agreement.

Transmission Owner

A signatory to this Agreement which: (1) transfers functional control of Tariff Facilities related

to the rates, terms and conditions of the OATT to SPP by executing this Agreement; or (2) appoints SPP under another agreement to provide service under the Transmission Tariff over Tariff Facilities which it owns or controls; or (3) is assigned by SPP to construct and accepts the obligation to construct new Tariff Facilities; or (4) undertakes another Transmission Owner's obligation to construct Tariff Facilities in accordance with Section 3.3(c) of this Agreement and Attachment O of the SPP OATT.

Upper Missouri Zone ("UMZ")

The_Upper Missouri Zone ("UMZ" or "Zone 19") is the rate pricing zone initially consisting of the following facilities that meet the requirements of Attachment AI, upon the transfer of those facilities to the functional control of the Transmission Provider: (i) the facilities of Western-UGP within the Eastern and Western Interconnections; (ii) the facilities owned or leased by Basin Electric Power Cooperative or Heartland Consumers Power District within the Eastern Interconnection; (iii) a portion of the facilities owned or leased by Basin Electric Power Cooperative within the Western Interconnection; and (iv) other facilities of the Western Area Power Administration transferred to the functional control of the Transmission Provider.

Western Area Power Administration-Upper Great Plains Region ("Western-UGP")

A division of the Western Area Power Administration that markets and transmits Federal power from reservoir projects under the control of the Department of the Army or the U.S. Bureau of Reclamation to Statutory Load Obligations, including preference power customers in Iowa, Minnesota, Montana, Nebraska, North Dakota, and South Dakota located in a defined marketing area. Western-UGP operates the WAUW Balancing Authority Area in the Western Interconnection, where certain of its transmission facilities are located.

Western Interconnection

One of the three major alternating-current electrical grids in North America. The Western Interconnection stretches from Western Canada South to Baja California in Mexico, reaching eastward over the Rockies to the Great Plains.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 2.1, MA 2.1 Operation and Planning, 2.0.0, A
Record Narative Name: Membership Agreement Section 2.1 Operation and Planning
Tariff Record ID: 75
Tariff Record Collation Value: 603844760 Tariff Record Parent Identifier: 74
Proposed Date: 2014-11-10
Priority Order: 500
Record Change Type: CHANGE
Record Content Type: 1
Associated Filing Identifier:

2.1 Operation and Planning

2.1.1 General

(a) SPP shall schedule transactions and administer transmission service over Tariff Facilities as necessary to provide service in accordance with the SPP OATT.

- (b) SPP shall function in accordance with Good Utility Practice and shall conform to applicable reliability criteria, policies, standards, rules, regulations, guidelines and other requirements of SPP and NERC, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements specified in this paragraph), and all applicable requirements of Federal and state regulatory authorities.
- (c) SPP shall maintain a publicly available registry of all facilities that are not classified as critical energy infrastructure information that constitute the Electric Transmission System.
- (d) SPP shall review and approve, as appropriate, requests for service, schedule transmission transactions, and determine available transfer capability under the OATT, provided that SPP shall coordinate with the Transmission Owner when processing requests for service involving its Tariff Facilities.
- (e) SPP shall be responsible for coordinating with neighboring regional organizations and/or non-member transmission owners or providers as appropriate.
- (f) SPP shall not exercise its administration of transmission service over the Tariff Facilities in such a way as to interfere with contracts between Transmission Owner and any Transmission Customer that are in effect as of the Effective Date of this Agreement except as permitted by the OATT.
- (g) SPP shall be responsible for documenting all transmission service requests, the disposition of such requests, and any supporting data required to support the decision with respect to such requests. SPP shall negotiate as appropriate to develop reciprocal service, equitable tariff application, compensation principles, and any related arrangements.
- (h) SPP shall propose and file with FERC pursuant to Section 205 of the Federal Power Act modifications to the OATT and make any

- other necessary filings subject to approval by the Board of Directors.
- (i) SPP shall develop penalties and incentives, subject to FERC filings where appropriate.
- (j) SPP shall direct Transmission Owner pursuant to the provisions of Section 3.3 to construct transmission facilities in accordance with coordinated planning criteria, or if necessary under the OATT.
- (k) SPP shall have the authority to direct the day-to-day operations of the Tariff Facilities in order to carry out its responsibilities as a Transmission Provider and Reliability Coordinator; provided, however, nothing in this Agreement or the OATT shall be construed to require a change in the physical control of any Tariff Facilities using a Party's existing facilities or equipment.
- (l) SPP shall take any actions necessary for it to carry out its duties and responsibilities, subject to receiving any necessary regulatory approvals and any necessary approvals from the Board of Directors.

2.1.2 Reliability

SPP shall have responsibility for reliability of the Electric Transmission System in connection with its rights, powers, and obligations under this Agreement. SPP shall act as the Reliability Coordinator of the Electric Transmission System, and as such, shall have reliability monitoring and emergency response responsibilities pursuant to related SPP Criteria and the following requirements:

(a) SPP shall monitor real-time data to determine whether any control areas are experiencing generation capacity deficiencies. If a generation capacity deficiency event threatens the security of the Electric Transmission System, SPP is authorized to and shall direct the acquisition of generation capacity and, if that direction is not satisfied, is authorized to and shall direct the shedding of firm load in the deficient control area.

- (b) SPP shall work with other reliability coordinators and non-member transmission owners or providers to develop regional reliability plans and emergency operating procedures.
- (c) SPP shall maintain emergency response procedures for responding to specified critical contingencies and shall continuously analyze issues that may require the initiation of such actions.
- (d) SPP is authorized to and shall direct the response to any emergency and Members shall carry out the required emergency actions as directed by SPP (except in cases involving endangerment to the safety of employees or the public), including the shedding of firm load if required for regional reliability.
- (e) After the conclusion of an emergency condition, any affected entity that disagrees with SPP's handling of the emergency may resolve that disagreement pursuant to the dispute resolution procedures in the Bylaws.
- (f) SPP shall monitor and coordinate the maintenance of adequate Electric Transmission System voltage levels with control areas and Transmission Owner, where appropriate.
- (g) SPP shall direct redispatch of generation in accordance with the OATT and in its role as Reliability Coordinator, subject to the generator receiving appropriate compensation pursuant to an applicable rate schedule.

2.1.3 Transmission Maintenance

SPP is required to approve all planned maintenance of the Electric Transmission System consistent with the following requirements:

- (a) SPP shall review planned transmission maintenance schedules submitted by Transmission Owner for a minimum of a rolling one-year period. These planned maintenance schedules shall be updated daily. Planned transmission maintenance requests shall be submitted to SPP at least one week in advance of an outage.
- (b) SPP shall analyze a planned transmission maintenance request to

determine its effect on available transfer capability, ancillary services, the reliability of the Electric Transmission System, and any other relevant effects. Within two business days of receiving a planned maintenance request, SPP shall provide a response. If SPP's response indicates that such planned transmission maintenance will have an adverse impact, Transmission Owner shall work with SPP to minimize the impact of such planned maintenance, up to and including re-scheduling the maintenance.

- (c) SPP shall notify Transmission Owner of the need to change previously reviewed planned transmission maintenance outages if forced transmission outages or other circumstances compromise the integrity or reliability of the Electric Transmission System. If Transmission Owner is fully compensated for any additional costs resulting from any changes in maintenance schedules as provided in an applicable rate schedule, Transmission Owner shall revise maintenance outages to address such emergency circumstances.
- (d) As part of its review process, SPP shall identify planned transmission maintenance schedules that limit available transfer capability. If requested by a Transmission Customer, SPP shall identify opportunities and associated costs for rescheduling planned maintenance to enhance available transfer capability. Transmission Owner shall be compensated for the additional costs of rescheduled maintenance as provided in the SPP OATT.
- (e) SPP shall be responsible for documenting all planned transmission maintenance requests, the disposition of those requests, and all data supporting the disposition of each request.
- (f) SPP shall coordinate with Transmission Owner to the extent practicable to implement schedules for unplanned transmission maintenance when conditions endanger the safety of employees or the public, may result in damage to facilities, or may result in the unsatisfactory operation of its transmission system or any other

transmission system.

2.1.4 Generation Maintenance

SPP shall coordinate the maintenance of generating units as appropriate to the extent such generation maintenance directly affects the capacity or reliability of the Electric Transmission System and the generation is located in the SPP Region as follows:

- (a) SPP shall review planned generating unit maintenance schedules submitted by generation owners for a minimum of a rolling one-year period. The planned maintenance schedules shall be updated daily. SPP shall keep such information confidential.
- (b) SPP shall analyze a planned generating unit maintenance schedule to determine its effect on available transfer capability, ancillary services, the reliability of the Electric Transmission System, and any other relevant effects. SPP shall inform a generation owner if its maintenance schedule is expected to have an impact on the reliability of the Electric Transmission System.
- (c) As part of its review process, SPP shall identify generating unit maintenance schedules that limit available transfer capability and shall identify opportunities and associated costs for rescheduling planned maintenance to enhance available transfer capability.
- (d) A generation owner that changes planned maintenance at the request of SPP pursuant to this Section 2.1.4 shall be compensated in accordance with the SPP OATT.
- (e) SPP shall be responsible for documenting all planned generating unit maintenance schedules, all schedule changes, and all SPP studies and services performed with respect to planned generation maintenance.

2.1.5 Planning Activities

(a) SPP shall engage in such planning activities, in coordination with Member, as are necessary to fulfill its obligations under this Agreement, SPP Criteria and the OATT. Such planning shall

conform to applicable reliability requirements of SPP, NERC, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements), and all applicable requirements of Federal or state regulatory authorities. Such planning shall seek to minimize costs, consistent with the reliability and other requirements set forth in this Agreement. The division of responsibility for planning between Member and SPP is set forth in the SPP Criteria.

(b) As part of its planning activities, SPP shall be responsible for planning, and for directing or arranging, necessary transmission expansions, additions, and upgrades that will enable it to provide efficient, reliable and non-discriminatory transmission service and to coordinate such efforts with the appropriate state authorities.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 2.4, MA 2.4 Additional Obligations of SPP, 1.0.0, A Record Narative Name: Membership Agreement Section 2.4 Additional Obligations of SPP Tariff Record ID: 78

Tariff Record Collation Value: 629680964 Tariff Record Parent Identifier: 74

Proposed Date: 2014-11-10 Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

2.4 Additional Obligations of SPP

2.4.1 Inspection and Auditing Procedures

SPP shall grant Member, its employees, agents, or external auditors, and Federal and state regulatory authorities having jurisdiction over SPP or Member, such access to SPP's books, records, business practices, control procedures and required audit test results, and related financial transactions and settlement activities as is necessary to verify compliance by SPP with this Agreement, to audit and verify transactions under this Agreement, and to assist Member in complying with its statutory and regulatory requirements. Such access shall be at reasonable times and under reasonable conditions. SPP shall also comply with the reporting requirements of Federal and state regulatory

authorities having jurisdiction over SPP with respect to the business aspects of its operations. Contact between officers, employees, and agents of Member and those of SPP shall comply with the Standards of Conduct.

2.4.2 Stranded Cost Recovery Charges

SPP shall collect and distribute, as appropriate, any stranded cost recovery charges pursuant to applicable schedules accepted by appropriate regulatory entities.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Section 3.0, MA 3.0 Commitments, Rights, Powers, and Obligations..., 1.0.0, A

Record Narative Name: Membership Agreement Section 3.0 Commitments, Rights, Powers and Obligations of Member

Tariff Record ID: 79

Tariff Record Collation Value: 638293032 Tariff Record Parent Identifier: 69

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

3.0 Commitments, Rights, Powers, and Obligations of Member

Member has made the following commitments, and shall have the following rights and shall be responsible for the following functions, some of which apply only to a Transmission Owner, some only to a Non-Transmission Owner.

- (a) Transmission Owner shall transfer functional control related to the rates, terms and conditions of the OATT of its Transmission Facilities, subject to receiving all necessary regulatory authorizations, thereby allowing SPP to (i) direct the operation of the Transmission Facilities in accordance with the terms of this Agreement; (ii) administer transmission service under the Transmission Tariff over that Transmission Owner's Tariff Facilities; and (iii) receive funds from Transmission Customers relating to transmission service over Tariff Facilities and distribute funds to the Transmission Owner. Where Member, owns generators within the SPP Region which directly affect the capacity or reliability of the Electric Transmission System, it shall offer to provide the ancillary services required under the OATT at rates approved by regulatory authorities, where appropriate, to the extent such generators are able to provide such ancillary services.
- (b) Transmission Owner shall operate and maintain its Tariff Facilities subject to the

- requirements of this Agreement.
- (c) Where Transmission Owner is a balancing area operator, it shall continue to operate its balancing areas for local generation control and economic dispatch, and shall be responsible for identifying and addressing local problems in a reliable manner.
- (d) Transmission Owner shall provide transmission service over its Tariff Facilities at the direction of SPP pursuant to the terms of the OATT.
- (e) Member agrees to comply with the instructions of SPP in its role as Reliability Coordinator.
- (f) Transmission Owner shall retain all rights of ownership, including legal and equitable title in its Tariff Facilities, subject to the provisions of this Agreement. Transmission Owner, or one acting under its authority, shall retain all rights to access to its Tariff Facilities so long as such access is consistent with the provisions of this Agreement.
- (g) Notwithstanding any other provision in this Agreement, Transmission Owner shall not be obligated or be considered to be allowing transmission over its facilities if such transmission would cause the loss of the tax-exempt status of Transmission Owner or any bonds or other debt of Transmission Owner.
- (h) Notwithstanding any other provisions of this Agreement, Member reserves the right to exercise operational authority of Member's Tariff Facilities (1) to protect public safety and the safety of its workers, to prevent damage to equipment, and to preserve reliability in compliance with NERC standards, and (2) as necessary to preserve Member's rights, duties and obligations regarding electric service to its retail and wholesale native load customers pursuant to any applicable Federal or state law and consistent with NERC standards, if SPP's exercise of operational authority over the Tariff Facilities would endanger said electric service or is contrary to or would curtail, surrender or delegate such Federal or state law rights, duties and obligations. Member will, as soon as reasonably practicable thereafter, notify SPP of such actions taken by Member. Member and SPP will meet and confer regarding the matter and, as necessary, negotiate in good faith to modify the Agreement to address the matter.

3.3 Construction

- (a) As part of its planning activities, SPP shall be responsible for planning, and for directing or arranging, necessary transmission expansions, additions, and upgrades that will enable it to provide efficient, reliable and non-discriminatory transmission service and to coordinate such efforts with the appropriate Federal or state authorities, including the Member's governing board where it serves as that authority, or in the case of a Federal Power Marketing Agency, the Administrator. Transmission Owner shall use due diligence to construct transmission facilities as directed by SPP in accordance with the OATT and this Agreement, subject to such siting, permitting, and environmental constraints as may be imposed by Federal, state, and local laws and regulations, and subject to the receipt of any necessary Federal or state regulatory approvals, including, as necessary, the Member's governing board where it serves as that authority. Such construction shall be performed in accordance with Good Utility Practice, applicable SPP Criteria, industry standards, Transmission Owner's specific reliability requirements and operating guidelines (to the extent these are not inconsistent with other requirements), and in accordance with all applicable requirements of Federal or state regulatory authorities. Transmission Owner shall be fully compensated to the greatest extent permitted by FERC, or other legislative or regulatory authority for the costs of construction undertaken in accordance with the OATT.
- (b) After a new transmission project has received the required approvals and been approved by SPP, SPP will direct the appropriate Transmission Owner(s) to begin implementation of the project in accordance with

Attachment O of the OATT.

3.10 Pricing

Transmission Owner shall possess the unilateral right to file with FERC, and, if the Transmission Owner is non-jurisdictional, the Transmission Owner shall have the option to file with FERC, or submit to SPP for filing with FERC, pursuant to Section 205 of the Federal Power Act and Federal statutory and regulatory requirements, including Delegation Order No. 00-37.00A, 10 CFR Part 903 and 18 CFR Part 300, as amended or superseded, modifications to change the rates or rate structure for transmission service over its Tariff Facilities, including filing a fixed revenue requirement and supporting data or a rate formula template for its cost of service revenue requirements, and to submit proposals or filings governing new construction with FERC; provided, however, Transmission Owner may not submit a proposal which results in a Transmission Customer paying two or more transmission charges for transmission for one transaction under the OATT (excluding Distribution Facilities for which an additional charge may be imposed, and Grandfathered Agreements as defined in the OATT). Transmission Owner shall notify SPP in advance of its intention to submit a filing to FERC and provide SPP with a copy of the filing. No approval from SPP is required for such filings.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 3.11, MA 3.11 No Waiver of Jurisdictional Immunity, 1.0.0, A

Record Narative Name: Membership Agreement Section 3.11 No Waiver of Jurisdictional Immunity

Tariff Record ID: 90

Tariff Record Collation Value: 733025780 Tariff Record Parent Identifier: 79

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

3.11 No Waiver of Jurisdictional Immunity

If Member is not subject to the jurisdiction of FERC as a public utility under the Federal Power Act, Member shall not be required to take any action or participate in any filing or appeal that would confer FERC jurisdiction over Member that does not otherwise exist. Any order, decision, rule or regulation issued by FERC to SPP or any other Members or Member of SPP relating to matters exempt from FERC jurisdiction under Section 201(f) of the Federal Power Act shall not apply directly or separately to a non-jurisdictional Member. Without limiting the generality of the foregoing, except as otherwise provided in the Federal Power Act, a non-jurisdictional Member shall not be bound or obligated by any FERC order, decision, rule or regulation requiring a change in the rates, terms or conditions for transmission service or compensation for utilizing the transmission facilities of a non-jurisdictional Member, which conflicts with applicable Federal or state law, including any order requiring the suspension of the use of such rates, terms or conditions or the payment of refunds of rates or compensation previously collected or received. A non-jurisdictional Member and SPP acknowledge that FERC, in the context of its jurisdiction over SPP's rates, may review a non-jurisdictional Member's revenue requirement and rates to the extent they comprise or affect the rates charged by SPP or other Members. In the case of a Federal Power Marketing Agency, this review shall be consistent with the Delegation Order No. 00-037.00A, as superseded or amended, from the Secretary of Energy to the Power Marketing Administrations and the FERC, including the regulations implementing this review authority. If FERC does not accept a non-jurisdictional Member's revenue requirement or rates, the non-jurisdictional Member may terminate this Agreement pursuant to the withdrawal provisions of the Agreement. In such event, the non-jurisdictional Member and SPP agree to meet and confer prior to any termination of this Agreement. Nothing in this Agreement, or the participation of a non-jurisdictional Member in SPP and its operations waives any objection to or otherwise constitutes a consent to, the jurisdiction by FERC that does not otherwise exist over the non-jurisdictional Member or its transmission service, facilities and rates.

Section 3.12, MA 3.12 Compliance with Federal or State Law, 1.0.0, A

Record Narative Name: Membership Agreement Section 3.12 Compliance with State Law

Tariff Record ID: 91

Tariff Record Collation Value: 741637848 Tariff Record Parent Identifier: 79

Proposed Date: 2014-11-10 Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

3.12 Compliance with Federal or State Law

Notwithstanding any other provision of this Agreement, a non-jurisdictional Member shall not be required to take any action or do any other thing with respect to rates, charges, terms or conditions of service, the resolution of disputes under this Agreement or any other matter regarding its obligations and performance under this Agreement, that (i) the non-jurisdictional Member is not permitted by Federal or state law to undertake or that is prohibited in whole or in part by any Federal or state law or regulation applicable to the non-jurisdictional Member; or (ii) would require the non-jurisdictional Member to violate a provision of such state or Federal law or regulation in order to comply with this Agreement. Determination of compliance with and permissible action, conduct or obligations under this Section 3.12 by a non-jurisdictional Member shall be within the sole jurisdiction of the non-jurisdictional Member's governing board, or in the case of a Federal Power Marketing Agency, its Administrator, subject to applicable Federal or state court review. A non-jurisdictional Member shall not object to SPP's participation in any Federal or state proceedings that impact the non-jurisdictional Member's ability to perform under this Agreement or determinations regarding such impact. To the extent possible without violating Federal or state law, a non-jurisdictional Member shall notify SPP in advance of any action that the non-jurisdictional Member is required to take that the non-jurisdictional Member believes would constitute a violation of Federal or state law, and the non-jurisdictional Member and SPP promptly shall meet and confer regarding the matter. As necessary, the non-jurisdictional Member and SPP agree to negotiate in good faith to modify the Agreement as consistent as possible with the original intent to allow SPP to exercise operational authority over the non-jurisdictional Member's Tariff Facilities as otherwise provided in the Agreement. If the non-jurisdictional Member and SPP are unable to resolve the matter, the non-jurisdictional Member may terminate this Agreement pursuant

to the withdrawal provisions of the Agreement.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:
Section 4.1, MA 4.1 Events of Termination and Partial Termination, 1.0.0, A
Record Narative Name: Membership Agreement Section 4.1 Events of Termination and Partial Termination
Tariff Record ID: 93
Tariff Record Collation Value: 758861984

Tariff Record Parent Identifier: 92

Proposed Date: 2014-11-10 Priority Order: 500 Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

4.1 Events of Termination and Partial Termination.

A "Termination" shall mean any cessation of Membership, voluntary or involuntary, or a termination of this Agreement for any reason including the following:

- a. Member voluntarily withdraws from membership under Sections 4.0 or
 5.0 of this Agreement;
- b. An involuntary termination of membership occurs pursuant to Section 6.0 of this Agreement;
- c. Member withdraws from membership or terminates this Agreement to comply with the terms of any applicable law or regulation;
- d. A withdrawal from membership or termination of this Agreement is ordered by any court or administrative agency of competent jurisdiction; SPP reserves the right, but is not obligated, to maintain before such court or administrative agency, or on any appeal, that FERC has preemptive jurisdiction;
- e. A material breach or repudiation of this Agreement, in the discretion of the non-breaching or non-repudiating party;
- f. The liquidation or dissolution of SPP, unless a third party has assumed the rights and obligations of SPP under this Agreement consistent with Section 8.2, and has reasonably demonstrated capability to perform SPP's obligations under this Agreement;
- g. An agreement between SPP and the Member to terminate this Agreement.A "Partial Termination" occurs upon a Member's voluntary removal of a portion

of its transmission facilities or customers from the SPP Region, including, by way of example and not limitation, sale of a part of the Member's distribution or transmission network or transfer to another service provider of a portion of its retail load.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:
Section 4.2, MA 4.2 Termination Procedures and Effective Dates, 2.0.0, A
Record Narative Name: Membership Agreement Section 4.2 Termination Procedures and Effective Dates
Tariff Record ID: 94
Tariff Record Collation Value: 767474052 Tariff Record Parent Identifier: 92
Proposed Date: 2014-11-10
Priority Order: 500
Record Change Type: CHANGE
Record Content Type: 1
Associated Filing Identifier:

4.2 Termination Procedures and Effective Dates

4.2.1 Notice of Voluntary Withdrawal

- (a) **Notice.** Subject to Section 4.3, a Member may withdraw voluntarily from this Agreement, provided that it has given written notice to the President of its intent to withdraw. Notice of intent to withdraw must state a proposed date for the withdrawal and be delivered to the President no less than twenty-four (24) months prior to such date. The President will advise the Members and the Board of Directors of any withdrawal notices received. In order to assure that there is no more than one proposed termination date with respect to a Member, a withdrawal notice shall be deemed to supersede any prior withdrawal notice given by the Member, except that a Member may not submit a withdrawal notice less than twenty-four (24) months prior to the termination date proposed in the Member's previous notice of intent to withdraw. Voluntary withdrawal is a Termination and creates the same obligations as a Termination for any other reason. Upon receiving a notice of intent to withdraw, SPP shall account for such notice of intent to withdraw in the SPP planning process, unless the Member plans to continue to take transmission service from SPP after the termination date.
- (b) Withdrawal Deposit. A Member submitting a written notice of its intent to withdraw from this Agreement must simultaneously submit a cash withdrawal deposit to SPP, as set forth in the table below. SPP will not accept a notice of intent to withdraw without a withdrawal deposit. SPP will treat the withdrawal deposit as a pre-payment of a portion of the costs SPP incurs to process the Member's withdrawal from SPP, as set forth in Section 4.3.2(d) of this Agreement, or the costs associated with reintegrating the Member into SPP if the Member subsequently rescinds its notice of intent to withdraw and SPP incurs costs to reintegrate the

Member. Withdrawal deposits are as follows:

Member Category	Withdrawal Deposit
Load Serving Entity	\$ 150,000
Non-Load Serving Entity	\$ 50,000

If the cost of processing Member's withdrawal as calculated by SPP pursuant to Section 4.3.2(d) of this Agreement exceeds the withdrawal deposit, the additional amount shall be included in the invoice SPP provides to the Member under 4.3.2(e) of this Agreement. If the Member rescinds its notice of intent to withdraw and the cost of processing the Member's withdrawal and subsequent reintegration into SPP exceeds the withdrawal deposit, SPP shall invoice the Member for the amount of the cost that exceeds the deposit, and the Member shall provide payment to SPP within thirty (30) days of receipt of the invoice. If the withdrawal deposit exceeds the costs of processing the Member's withdrawal and/or reintegration, SPP shall refund the difference to the Member. Notwithstanding the foregoing, a Federal Power Marketing Agency shall not be required to make a withdrawal deposit and shall only be responsible for paying SPP costs after they are incurred and appropriately invoiced pursuant to 4.3.2(e).

4.2.2 Effective Date of Termination

- Owner subject to FERC jurisdiction, the Termination Date shall be the date proposed in the withdrawal notice under Section 4.2.1 or otherwise agreed by SPP. If the withdrawing Member is a Transmission Owner subject to FERC jurisdiction, the Termination Date shall be the later of (i) the proposed date specified in the withdrawal notice or otherwise agreed by SPP, (ii) the effective date, if any, set by the FERC order approving the withdrawal; or (iii) the date that such FERC order is no longer subject to review by a court of competent jurisdiction.
- (b) **Termination other than Voluntary Withdrawal**. If the Termination occurs for any reason other than the Member's voluntary withdrawal under Section 4.2.1 or by agreement with SPP, the Termination Date shall be as follows:

- (i) If the Member is not a Transmission Owner subject to FERC jurisdiction, the Termination Date shall be the date of the event by which the Termination occurs, for example, the date a party gives notice that it will treat a breach or repudiation as a Termination or the date a Member withdraws in order to comply with the terms of a law or regulation. The foregoing notwithstanding, if the Termination occurs due to the order of a court or administrative agency, the Termination Date shall be the date the order is no longer subject to review by a court of competent jurisdiction.
- (ii) If the Member is a Transmission Owner requiring regulatory agency approval prior to effectively withdrawing from SPP, then the Termination Date shall be the later of (i) the effective date, if any, set by the regulatory agency order approving the Termination; or (ii) the date that such regulatory order is no longer subject to review by a court of competent jurisdiction.
- (iii) In conjunction with the requirements and limitations imposed in Section 4.2.2(b)(ii) above; Transmission Owning Members of SPP wishing to withdraw from SPP and requiring regulatory agency approval prior to such withdrawal being effective are required to initiate the required regulatory filings seeking approval to withdraw prior to the end of the 24-month notice period. Additionally, these members must provide SPP copies of the regulatory filings in a timely manner after filing with the appropriate regulatory agency. Failure to comply with the terms of this paragraph will effectively rescind the notice of the withdrawing member.
- (iv) Member may terminate this Agreement with less than the required twenty-four (24) months' notice, in the event that the Federal or state law governing Member changes, or any provisions of this Agreement, the provisions of SPP's OATT, or SPP's Bylaws are changed or modified in a manner that causes a conflict with the

Member's Federal or state law, regulations, or rate schedules, and the internal dispute resolution process described in Section 12 of the OATT is unable to resolve such conflict. In such event, Member and SPP shall meet and confer to facilitate the withdrawal as soon as practicable as necessary to ensure compliance with Federal or state law.

Any Member with Transmission Facilities located in the Upper (v) Missouri Zone may terminate this Agreement with less than the required twenty-four (24) month notice in the event that Western-UGP or Basin Electric Power Cooperative withdraws from SPP in accordance with its respective withdrawal rights or if FERC finds that SPP has not adhered to all of the Federal Power Marketing Agency Amendments, the Basin Electric Amendments, or the Heartland Amendments. In such event, Member and SPP shall meet and confer to facilitate the withdrawal as soon as practicable or as necessary to ensure compliance with state or Federal law. In the event of a withdrawal by Western-UGP or Basin Electric Power Cooperative, Member's withdrawal will become effective on the same date as that of Western-UGP or Basin Electric Power Cooperative. Such Member also may terminate this Agreement in the event that SPP files and FERC approves changes to the Federal Power Marketing Agency Amendments, the Basin Electric Amendments, or the Heartland Amendments that have a material adverse effect on such Member. If such Member exercises its withdrawal rights under this provision, the financial obligations will be calculated under § 4.3 of this Agreement.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 4.3, MA 4.3 Obligations Upon Termination, 2.0.0, A

Record Narative Name: Membership Agreement Section 4.3 Obligations Upon Termination

Tariff Record ID: 95

Tariff Record Collation Value: 776086120 Tariff Record Parent Identifier: 92

Proposed Date: 2014-11-10

Priority Order: 600

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

4.3 Obligations Upon Termination

4.3.1 Obligation to Hold Users Harmless

Transmission Customers taking service which involves facilities being withdrawn by a Transmission Owner from the functional control of SPP and where such service is under transmission contracts executed before the Termination Date shall continue to receive the same service for the remaining term of each such contract at the same rates, terms, and conditions that would have been applicable if the Termination or Partial Termination had not occurred. Transmission Owner agrees to continue providing service to such Transmission Customers in accordance with the preceding sentence, and shall receive revenues calculated in accordance with the OATT but no more in revenues for that service that if there had been no Termination or Partial Termination.

4.3.2 Obligation to Pay Current and Existing Obligations

- (a) In the event of a Termination or Partial Termination, Member shall pay all obligations incurred under this Agreement at any time prior to the Termination Date. In addition, in order for SPP to recover a portion of certain debts and cost payable by SPP after the Termination Date as further specified in this Agreement, the Member shall pay all Existing Obligations (as defined herein) calculated as of the Termination Date. SPP shall make reasonable efforts to mitigate the Member's Existing Obligations by commercially reasonable actions (such as prepayment of allocable debt, or investment of part or all of the Member's payment in an interest-bearing instrument) and, in its discretion, may further discount the Member's Existing Obligations to reflect any additional mitigation SPP determines it will achieve.
- (b) "Existing Obligations" are all of the following and other obligations as may be set forth in the Bylaws from time to time;
 - i. Member's unpaid annual membership fee,
 - ii. Member's unpaid dues, assessments, and other amounts charged

- under Section 3.8 of this Agreement, Section 8.4 of the Bylaws, or otherwise under the Bylaws, plus the Member's share of costs SPP customarily includes in such dues, assessments or other charges, but which as of the Termination Date SPP had not included in such dues assessments or other charges.
- iii. Member's share (computed in accordance with the Bylaws) of the entire principal amounts of all SPP Financial Obligations outstanding as of the Termination Date. "Financial Obligations" are all long-term (in excess of six (6) months) financial obligations of SPP, including but not limited to the following:
 - a. debts under all mortgages, loans, loan agreements, borrowings, promissory notes, bonds, and credit lines under which SPP is obligated, including principal and interest;
 - all payment obligations under equipment leases, financing leases, capital leases, real estate and office space leases, consulting contracts, and contracts for outsourced services;
 - c. any unfunded liabilities of any SPP employee pension funds, whether or not liquidated or demanded; and
 - d. the general and administrative overhead of SPP for a period of three (3) months.
- iv. Any costs, expenses or liabilities incurred by SPP directly due to the Termination, regardless of when incurred or payable, and including without limitation prepayment premiums or penalties arising under SPP Financial Obligations.
- v. Member's share (computed in accordance with the Bylaws) of all interest that will become due for payment with respect to all interest bearing Financial Obligations after the Termination Date and until the maturity of all Financial Oblations in accordance with their respective terms ("Future Interest"). In the event that a Financial Obligation carries a variable interest rate, the interest rate in effect at the Termination Date shall be used to calculate the

applicable Future Interest. In determining the Member's share of Future Interest, SPP shall take into account any reduction of Financial Obligations due to mitigation under this Section.

- (c) In the event of a Partial Termination, Existing Obligations shall first be calculated as though a Termination occurred, and the Member shall pay a percentage thereof as Existing Obligations due to the Partial Termination. Such percentage shall be the percentage reduction of the Net Energy for Load Ratio applicable to the Member resulting from the Partial Termination.
- In the event of a Termination or Partial Termination by a Member, the Member shall pay to SPP all costs SPP incurs to remove the Member's facilities and/or load from SPP markets and operations. Such costs will be determined by SPP and shall include but not be limited to costs associated with modifying systems and databases, staff time, legal costs, and all costs of completing other tasks necessary to process the Member's Termination. SPP will apply the Member's withdrawal deposit, as specified in Section 4.2.1(b), to such costs, and any costs exceeding the withdrawal deposit shall be included in the invoice to the Member as discussed in Section 4.3.2(e) of this Agreement.
- (e) SPP shall invoice Member for Existing Obligations within one month after the Termination Date, except that delay by SPP in issuing the invoice shall not diminish Member's obligation to make timely payment. The invoice shall be due and payable no later than five (5) business days after issuance. Any amounts owed by SPP to the Member shall, solely at SPP's election and in its discretion, be offset against the Member's Existing Obligations or paid to the Member concurrently with issuance of the invoice.
- (f) The Member acknowledges and agrees that Existing Obligations include amounts that SPP expects to accrue and that will become payable by SPP between the date of Member's Notice of Termination and the Member's Termination Date, and that no part of a payment of Existing Obligations

shall be refundable to the Member under any circumstances, including (except as provided in this Section with respect to mitigation or the execution of a new Agreement by the Member after the Member's Termination) any reduction of the Financial Obligations. Any disagreement as to the calculation of Existing Obligations shall be resolved in accordance with the dispute resolution procedures in the Bylaws. If, after Termination, the Member elects to re-join SPP and execute the Agreement then in effect, SPP, in its sole discretion, may elect to credit a portion or all of the Member's Existing Obligations paid to SPP upon the Member's earlier Termination against any future payments owed by the Member to SPP.

4.3.3 Construction of Transmission Facilities

Any obligations relating to the construction of new facilities pursuant to an approved plan of SPP shall be negotiated between SPP and the Transmission Owner prior to the Termination Date so as to continue the Transmission Owner's construction obligation for facilities for which SPP has issued a notification to construct to the Transmission Owner prior to the Termination Date. If such obligations cannot be resolved through negotiations, they shall be resolved in accordance with the dispute resolution procedures in the Bylaws and Agreement.

4.3.3A Financial Obligations for Transmission Facilities

This Section 4.3.3A applies to any Terminated Member that was a Transmission Owner at the time it submitted its notice of intent to withdraw. Such Terminated Member shall remain financially responsible for all financial obligations incurred and costs allocated to its load for transmission facilities approved prior to the Termination Date. Payments in fulfillment of any such obligations and allocated costs shall commence on the date that the costs of such transmission facilities are reflected in SPP's generally applicable rates, unless SPP and the Terminated Member agree to an alternate date. Rights, obligations, and payments applicable to time periods prior to the Termination Date shall be honored by SPP and the Terminated Member. Fulfillment and performance of such rights and obligations, and rights and obligations regarding the use of such

transmission facilities, shall be negotiated between SPP and the Terminated Member, and any disputes involving such rights and obligations shall be resolved in accordance with the dispute resolution procedures in the Bylaws and Agreement.

4.3.4 Regulatory and Other Approvals or Procedures

Any Termination with respect to a Transmission Owner shall be subject to applicable Federal and state law and regulatory approvals or procedures.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 5.1, MA 5.1 Regulatory and Other Authorities, 1.0.0, A Record Narative Name: Membership Agreement Section 5.1 Regulatory and Other Authorities Tariff Record ID: 97
Tariff Record Collation Value: 793310256 Tariff Record Parent Identifier: 96

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

5.1 Regulatory and Other Authorities

This Agreement and the participation of Member is subject to acceptance or approval by FERC, and may be subject to actions of respective Federal or state regulatory authorities to which Member may be subject, and to the actions of any other governmental body which may affect the ability of Member to participate in this Agreement. The following items describe Member's rights and obligations in the event regulatory and other approvals or acceptances are not obtained or changes are required:

(a) In the event FERC disapproves or refuses to accept this Agreement or the changes to the OATT, transmission service agreements, and Bylaws developed together with this Agreement, then this Agreement shall cease to be effective except that the signatories shall be obligated to attempt expeditiously and in good faith to negotiate a substitute agreement and OATT, transmission service agreements, and Bylaws which address the reasons for such FERC action. If, despite such good faith negotiation, the signatories are unable to produce such a substitute agreement and OATT, transmission service agreements, and Bylaws, then the signatories shall have no further obligations under this Agreement or any filing associated herewith.

(b) In the event of any order or decision by FERC or by a court modifying this Agreement or the OATT, transmission service agreements, and Bylaws submitted as part of the initial filing seeking FERC acceptance or approval, that in the judgment of Member adversely affects it, then Member, at its sole discretion, may withdraw from this Agreement by providing written notice to the President of SPP no later than thirty days after such order or decision without receiving any FERC authorization. In such event, Member will in good faith negotiate to determine whether changes should be made to the Agreement or the OATT, transmission service agreements, and Bylaws to address the reasons for Member's withdrawal.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Section 5.2, MA 5.2 Tax Authorities, 1.0.0, A

Record Narative Name: Membership Agreement Section 5.2 Tax Authorities

Tariff Record ID: 99

Tariff Record Collation Value: 801922324 Tariff Record Parent Identifier: 96

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

5.2 Tax Authorities

If the Internal Revenue Service or any other Federal, state, or local taxing authority issues, or fails to issue, any ruling, or imposes any requirement or obligation, in connection with this Agreement on Member adverse to Member (in its sole judgment), or if adherence to this Agreement jeopardizes the tax-exempt status of Member or its bonds, then Member may, within 30 days of the date of such final order, or a good faith belief of such adverse consequences, withdraw from this Agreement subject to receiving any necessary regulatory approvals. In such event, Member and SPP will, in good faith, negotiate to determine whether changes should be made to the Agreement to address the reasons for Member's withdrawal.

Nothing in this Agreement, nor Member's obligations and performance thereunder, shall affect, or require Member to take or refrain from taking any action that would affect the rights and obligations or enforceability of Member's present or future

bond resolutions, tax-exempt debt covenants and financing agreements. Member shall determine in its sole discretion and judgment, in accordance with advice and opinions from its legal counsel, what actions, conduct and performance it is permitted to or must take under its bond resolutions, tax-exempt debt covenants and financing agreements. Member and SPP will meet and confer regarding the matter and, as necessary, negotiate in good faith to modify the Agreement to address the matter.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 5.3, MA 5.3 Effectiveness as to Certain Members, 1.0.0, A Record Narative Name: Membership Agreement Section 5.3 Effectiveness as to Certain Members Tariff Record ID: 100

Tariff Record Collation Value: 810534392 Tariff Record Parent Identifier: 96

Proposed Date: 2014-11-10

Priority Order: 500 Record Change Type: CHANGE

Record Content Type: 1
Associated Filing Identifier:

5.3 Effectiveness as to Certain Members

The effectiveness of this Agreement as to Member where it is a governmental entity or non-public utility and has outstanding tax-exempt bonds issued to finance, in whole or in part, generation, transmission, or Distribution Facilities is dependent upon satisfaction or Member's written waiver of the following conditions precedent:

- (a) Receipt of an unqualified opinion of a nationally recognized bond counsel to the effect that the provisions of this Agreement do not adversely affect the exclusion from gross income of interest on any such outstanding bonds issued to finance generation, transmission, and Distribution Facilities under the Internal Revenue code of 1986, as amended;
- (b) Receipt of an unqualified opinion of a nationally recognized bond counsel or general counsel to Member to the effect that the provisions of this Agreement do not constitute a breach or impairment of, or a default under, any agreement to which it is a party, including, but not limited to, its master bond resolution, as amended, and any power sales contracts with its municipal transmission users (if any), as amended, or other agreements;
- c) Receipt of a certificate of the trustee for any such outstanding bonds issued for generation, transmission and Distribution Facilities to the effect that Member's entry into this Agreement is permitted under the master

bond resolution, as amended; and

(d) Receipt of an opinion of nationally recognized bond counsel or general counsel to Member that it has full constitutional and statutory authority to enter into this Agreement.

In the event that any of the foregoing conditions are not satisfied or waived by Member, then it shall promptly give notice of its objections or conditions which have not been satisfied to SPP, and SPP shall expeditiously attempt in good faith to negotiate a substitute agreement.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Section 8.1, MA 8.1 Governing Law, 1.0.0, A

Record Narative Name: Membership Agreement Section 8.1 Governing Law

Tariff Record ID: 104

Tariff Record Collation Value: 844982664 Tariff Record Parent Identifier: 103

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

8.1 Governing Law

This Agreement shall be interpreted, construed, and governed by the laws of the State of Arkansas, except to the extent preempted by the law and/or unless a court with jurisdiction rules otherwise, and in a case involving a Federal Power Marketing Agency, Federal law shall apply, provided, however, that (i) all matters relating to real property or any interest in realty shall be governed by the laws of the State wherein such real property or interest in realty is physically located, and (ii) any court or regulatory body applying Arkansas law shall give full effect to Section 3.12 of this Agreement regarding Member's obligations under state law.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Section 8.2, MA 8.2 Successors and Assigns, 1.0.0, A

Record Narative Name: Membership Agreement Section 8.2 Successors and Assigns

Tariff Record ID: 105

Tariff Record Collation Value: 853594732 Tariff Record Parent Identifier: 103

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

8.2 Successors and Assigns

This Agreement shall inure to the benefit of, and be binding upon SPP and Member, their respective successors and assigns permitted hereunder, but shall not be assignable by SPP without prior written agreement from Member, with such written agreement not to be unreasonably withheld, or by Member, by operation of law or otherwise, without the approval of the Board of Directors which approval shall not be unreasonably withheld, except that no Board of Directors approval is required as to a successor in the operation of Transmission Owner's Tariff Facilities committed to administration by SPP by reason of a merger, consolidation, reorganization, sale, spin-off, or foreclosure, as a result of which substantially all such transmission facilities are acquired by such successor, and such successor becomes a Transmission Owner under this Agreement.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:

Section 8.4, MA 8.4 Severability, 1.0.0, A

Record Narative Name: Membership Agreement Section 8.4 Severability

Tariff Record ID: 107

Tariff Record Collation Value: 870818868 Tariff Record Parent Identifier: 103

Proposed Date: 2014-11-10

Priority Order: 500

Record Change Type: CHANGE Record Content Type: 1 Associated Filing Identifier:

8.4 Severability

Except as may be stated otherwise in any Amendments to this Agreement, each provision of this Agreement shall be considered severable, and if for any reason any provision of this Agreement, or the application thereof to any person, entity, or circumstance, is determined by a court or regulatory authority of competent jurisdiction to be invalid, void, or unenforceable, then the remaining provisions of this Agreement shall continue in full force and effect and shall in no way be affected, impaired, or invalidated, and such invalid, void, or unenforceable provision shall be replaced with a suitable and equitable provision in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid, void, or unenforceable provision. This Section 8.4 does not modify or change in any way the right of Member to withdraw

as provided elsewhere in this Agreement.

Record Content Description, Tariff Record Title, Record Version Number, Option Code:
Section 8.6, MA 8.6 Representations and Warranties, 1.0.0, A
Record Narative Name: Membership Agreement Section 8.6 Representations and Warranties
Tariff Record ID: 109
Tariff Record Collation Value: 888043004 Tariff Record Parent Identifier: 103
Proposed Date: 2014-11-10
Priority Order: 500
Record Change Type: CHANGE
Record Content Type: 1

8.6 Representations and Warranties

Associated Filing Identifier:

Member and SPP each represent and warrant to the other that as of the later of the date it executes this Agreement or the Effective Date of this Agreement:

- (a) It is duly organized, validly existing, and in good standing under the laws of the jurisdiction where organized;
- (b) Subject to any necessary approvals by Federal or state regulatory authorities of SPP, the execution and delivery by Member and SPP of this Agreement, and the performance of its respective obligations hereunder have been duly and validly authorized by all requisite action on the part of the signatories and does not conflict with any applicable law or with any other agreement binding upon the signatories, other than third party joint agreements covered in this Agreement. This Agreement has been duly executed and delivered by Member and SPP, and, subject to the conditions set forth in this Agreement, constitutes the legal, valid, and binding obligation on the part of Member and SPP, enforceable against it in accordance with its terms except insofar as the enforceability thereof may limited by applicable bankruptcy, insolvency, reorganization, fraudulent conveyance, moratorium, or other similar laws affecting the enforcement of creditor's rights generally, and by general principles of equity regardless of whether such principles are considered in a proceeding at law or in equity; and
- (c) There are no actions at law, suits in equity, proceedings, or claims pending or, to the knowledge of Member or SPP, threatened against Member or SPP before or by any Federal, state, foreign or local court, tribunal, or

governmental agency or authority that might materially delay, prevent, or hinder the performance by such entity of its obligations hereunder.

Record Content Description, Tariff Record Title, Record Version Number, Option Code: Section 8.10, MA 8.10 Good Faith Efforts, 1.0.0, A
Record Narative Name: Membership Agreement Section 8.10 Good Faith Efforts
Tariff Record ID: 113
Tariff Record Collation Value: 922491276
Tariff Record Parent Identifier: 103
Proposed Date: 2014-11-10
Priority Order: 500
Record Change Type: CHANGE
Record Content Type: 1
Associated Filing Identifier:

8.10 Good Faith Efforts

Member and SPP agree that each shall in good faith take all reasonable actions necessary to fulfill its respective obligations under this Agreement. Where the consent, agreement, or approval of Member or SPP must be obtained hereunder, such consent, agreement, or approval shall not be unreasonably withheld, conditioned, or delayed. Where Member or SPP is required or permitted to act, or omit to act, based on its opinion or judgment, such opinion or judgment shall not be unreasonably exercised. To the extent that the jurisdiction of any Federal or state regulatory authority applies to any part of this Agreement and/or the transactions or actions covered by this Agreement, Member and SPP secure any necessary or desirable approval or acceptance of such regulatory authorities of such part of this Agreement and/or such transactions or actions.

20140911-5115 FERC PDF (Unofficial) 9/11/2014 12:25:11 PM	
Document Content(s)	
860-2f3a2177-1c91-4d6b-a73d-aaee20f450a5.PDF	.1-2
Clean860-56ad2b2f-cfd0-4046-8cd0-700c64d0da3e.PDF	.3-59
Marked860-e1e28810-daa4-47ab-9703-1c611b62c070.PDF	.60-116
FERC GENERATED TARIFF FILING RTF	.117-168